

# Ca2+ currents in cardiac myocytes: Old story, new insights.

Fabien Brette, Jérôme Leroy, Jean-Yves Le Guennec, Laurent Sallé

# ► To cite this version:

Fabien Brette, Jérôme Leroy, Jean-Yves Le Guennec, Laurent Sallé. Ca2+ currents in cardiac myocytes: Old story, new insights.. Progress in Biophysics and Molecular Biology, 2006, 91 (1-2), pp.1-82. 10.1016/j.pbiomolbio.2005.01.001 . inserm-00141882

# HAL Id: inserm-00141882 https://inserm.hal.science/inserm-00141882

Submitted on 1 Aug 2007  $\,$ 

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



# Ca<sup>2+</sup> currents in cardiac myocytes: old story, new insights

Prog Biophys Mol Biol 91, 1-2 (2006) 1-82

Fabien Brette<sup>1</sup>, Jérôme Leroy<sup>2</sup>, Jean-Yves Le Guennec<sup>3</sup> and Laurent Sallé<sup>4</sup>

<sup>1</sup> School of Biomedical Sciences, University of Leeds, Leeds, LS2 9NQ, UK
<sup>2</sup> Department of Pharmacology, University College London, London WC1E 6BT, UK
<sup>3</sup> Emi-U 0211 Nutrition, Croissance, Cancer, Université de Tours, 37000 Tours, France
<sup>4</sup> EA3212 Laboratoire de Physiologie Cellulaire, Université de Caen, 14032 Caen, France

Correspondence to Dr Fabien Brette at the above address<sup>1</sup> e-mail: f.brette@leeds.ac.uk Tel: +44-(0)113-343-4276 Short title: Ca<sup>2+</sup> currents and cardiac myocytes

118 pages, 2 tables, 6 figures.

# ABSTRACT

Calcium is a ubiquitous second messenger which plays key roles in numerous physiological functions. In cardiac myocytes,  $Ca^{2+}$  crosses the plasma membrane via specialized voltage gated  $Ca^{2+}$  channels which have two main functions: (i) carrying depolarizing current by allowing positively charged  $Ca^{2+}$  ions to move into the cell; (ii) triggering  $Ca^{2+}$  release from the sarcoplasmic reticulum. Recently, it has been suggested than  $Ca^{2+}$  channels also participate in excitation-transcription coupling. The purpose of this review is to discuss the physiological roles of  $Ca^{2+}$  currents in cardiac myocytes. Next, we describe local regulation of  $Ca^{2+}$  channels by cyclic nucleotides. We also provide an overview of recent studies investigating the structure-function relationship of  $Ca^{2+}$  channels in cardiac myocytes using heterologous system expression and transgenic mice, with descriptions of the recently discovered  $Ca^{2+}$  channels  $\alpha_{1D}$  and  $\alpha_{1E}$ . We finally discuss the potential involvement of  $Ca^{2+}$  currents in cardiac pathologies, such as diseases with autoimmune components, and cardiac remodelling.

Key words: Cardiac, myocytes, calcium current, L type calcium channel, T type calcium channel, modulation, auto-immune, remodelling.

# **1 INTRODUCTION**

The importance of extracellular Ca<sup>2+</sup> in cardiac contraction has been known since the classical experiments of Ringer at the end of the XIX<sup>th</sup> century, which demonstrated that frog cardiac muscle cannot contract in  $Ca^{2+}$  free solutions (Ringer, 1883). More than half a century was needed to elucidate the mechanisms behind this observation. In the 1950's, cellular electrophysiology was dominated by neuronal preparations and the discovery of a surprisingly long action potential in cardiac preparations was unexpected (Draper and Weidmann, 1951). These action potentials showed a "plateau" phase (~200 ms duration) clearly contrasting with action potentials recorded from the squid axon (~5 ms) (Hodgkin and Huxley, 1952a). Characterization of the current(s) involved in this plateau phase was possible by the development of voltage clamp techniques. By recording inward current using two microelectrode voltage clamp in Purkinje fibers, Reuter (1967) was able to demonstrate that depolarization to -40 mV inactivated the Na<sup>+</sup> current (I<sub>Na</sub>) and further depolarization resulted in a small "slow inward" (si) current. This denomination arose by comparing the electrical characteristics of the si current with the major inward current described at this time, I<sub>Na</sub> (Hodgkin and Huxley, 1952b), which is fast and large. Reuter also showed that si current was unaffected by tetrodotoxin but abolished by  $Ca^{2+}$  free solutions and concluded that *si* current was carried by Ca<sup>2+</sup> (Reuter, 1967). This finding was confirmed in other cardiac preparations using other voltage clamp techniques (e.g. sucrose gap in frog trabeculae, Rougier et al., 1969).

The development of the isolated cell preparation led to improvement of the voltage clamp conditions, by increasing the quality of voltage clamp and reducing extracellular ionic accumulation-depletion in inter-cellular clefts in multi-cellular preparations. Insenberg and Klockner (1982) were the first to describe the *si* current as "large and fast" hence proposing that *si* was an inaccurate name for this current. A considerable further technical advance came with the development of the patch clamp technique (Hamill et al., 1981) which allows control of intracellular solutions (permitting the use of intracellular and extracellular channel blockers to separate the current of interest from other ionic currents). Tsien's laboratory proposed renaming *si* current as L-type Ca<sup>2+</sup> current, for "Long Lasting" (Nowycky et al., 1985); indeed, L type Ca<sup>2+</sup> current contrasted with another Ca<sup>2+</sup> current present in some cardiac cell types (*e.g.* sinoatrial node, atrial and Purkinje cells) which is "Tiny and Transient" (T-type Ca<sup>2+</sup> current, Nowycky et al., 1985). From the mid 1980's, investigation of Ca<sup>2+</sup> current (I<sub>Ca</sub>) in cardiac myocytes led to a profusion of data. The role of the Ca<sup>2+</sup> current (I<sub>Ca</sub>, L and T) as a depolarizing current was characterized in all cardiac cells types. Meanwhile, the development of fluorescent dyes to monitor intracellular ion concentrations helped to elucidate the role of

HAL author manuscript inserm-00141882, version 1

the L type  $Ca^{2+}$  current in excitation-contraction coupling (EC coupling) as the major trigger for  $Ca^{2+}$  release from the sarcoplasmic reticulum (via the  $Ca^{2+}$ -induced  $Ca^{2+}$ -release mechanism, CICR). More recently, the development of molecular biology techniques has allowed the identification of the molecular components of cardiac  $Ca^{2+}$  channels identified in electrophyiological studies. Molecular biology now provides tools to alter gene expression in transgenic mice, allowing knockout or overexpression of specific targets and the discovery of new  $Ca^{2+}$  channels in cardiac preparations.

This review is subdivided into four parts. In part I, a description is given of the physiological functions of cardiac  $Ca^{2+}$  channels *i.e.* their role in membrane depolarization, EC coupling and excitation-transcription coupling (ET coupling). Part II deals with regulation of cardiac  $Ca^{2+}$  channels, with emphasis on recent information about local signaling and interplay between different regulatory pathways. Part III provides recent information on the structure-function relationship of cardiac  $Ca^{2+}$  channels investigated using molecular biology and expression systems. Part IV discusses the involvement of  $I_{Ca}$  in cardiac pathologies, such as diseases with autoimmune components, and remodeling.

# HAL author manuscript inserm-00141882, version 1

# 2 PHYSIOLOGICAL FUNCTION OF CARDIAC Ca<sup>2+</sup> CHANNELS

The view that cardiac  $I_{Ca}$  can be classified into two main types, L-type and T-type, has been challenged recently by work on transgenic mice. The major L type  $Ca^{2+}$  channel expressed in the heart is generally thought to contain  $\alpha_{1C}$  (Cav1.2 in the new nomenclature, see section 4 for details and also table 1) as the pore-forming subunit. However, another high voltage activated channel (HVA),  $Ca_{V1,3}$  ( $\alpha_{1D}$ ), is also present in cardiac myocytes. This channel has properties similar to Cav1.2, although it activates at slightly more negative potentials (see section 3.1.1 and see Figure 1). Mice that lack  $Ca_V 1.2$  expression ( $\alpha_{1C}$ knockout,  $Ca_V 1.2 \alpha 1^{-/-}$ ) have been generated; this knockout mutation is embryonic lethal (Seisenberger et al., 2000), emphasising the crucial importance of Ca<sub>v</sub>1.2 in cardiac physiology. However before the embryo dies Ca<sub>V</sub>1.3 gene expression is upregulated, suggesting a compensatory regulation (Xu et al., 2003), although this does not lead to a viable mouse. The reverse also occurs: mice that lack  $Ca_V 1.3\alpha 1$  expression ( $\alpha 1D$  knockout,  $Ca_V 1.3$  $\alpha 1^{-/-}$ ) show compensatory upregulation of Ca<sub>V</sub>1.2 channel expression (Zhang et al., 2002b). In this mouse  $Ca^{2+}$  channel function is unaffected by the loss of one  $Ca_V 1.3$  allele (heterozygous mice), suggesting that there is upregulation to the same level of the output from the remaining functional allele (Zhang et al., 2002b). The same upregulation has been observed in  $Ca_V 1.2$ heterozygous mice (Seisenberger et al., 2000). The mechanism underlying this upregulation is not known. Recently, a third HVA channel has been described in rat and mouse cardiac myocytes (R type,  $\alpha_{1E}$  or Ca<sub>V</sub>2.3) (Weiergraber et al., 2000; Mitchell et al., 2002; Lu et al., 2004).

The low voltage activated (LVA, see Figure 1) family (the T type  $Ca^{2+}$  channel) has not expanded in a similar way, although the cloning of 3 genes coding for T-type  $Ca^{2+}$  channels has led to a better understanding of these channels in cardiac myocytes (Perez-Reyes, 2003).

In this section, we will review the potential physiological role of these  $Ca^{2+}$  channels in different types of cardiac myocytes. As described above,  $Ca^{2+}$  entering the cardiac cell has three main roles: carrying depolarizing current, activating contraction and regulating gene expression.

# 2.1 Depolarizing current

As  $Ca^{2+}$  enters the cell, it carries positive charges, which contributes to the electrical current responsible for depolarization. This function differs between cell types in the heart.

# 2.1.1 Working cells (ventricular, atrial)

Working cells in the heart are strongly contractile *i.e.* atrial and ventricular myocytes.  $I_{CaL}$  is present in all atrial and ventricular cells studied to date.  $I_{CaT}$  is not detectable in most ventricular myocytes *i.e.* frog, calf, cat, rabbit, ferret, mouse, rat, human (Argibay et al., 1988; Bean, 1989; Nuss and Houser, 1993; Yuan and Bers, 1994; Yuan et al., 1996; Nascimento, 1996; Li et al., 1999; Chen et al., 2002) but has been described in modest amounts in guineapig (Mitra and Morad, 1986; Sipido et al., 1998a).  $I_{CaT}$  is present in atrial myocytes from frog, dog, cat, rat and guinea pig (Wu and Lipsius, 1990; Romanin et al., 1992; Xu and Best, 1992; Alvarez et al., 1996; Yue et al., 1999), but not human (Ouadid et al., 1991), heart.

Although I<sub>CaL</sub> is present on the sarcolemma of cardiac myocytes, it has become increasingly clear that  $I_{CaL}$  is not uniformly distributed on the cardiac cell membrane. The sarcolemma of mammalian ventricular myocytes contains invaginations called transverse (t)tubules, whereas most mammalian atrial myocytes lack t-tubules (reviewed in Brette and Orchard, 2003). An early study used the properties of dihydropyridines (DHPs) to bind L-type Ca<sup>2+</sup> channels (also named DHP receptors, DHPr) and showed that in the rabbit heart, the membrane fraction from the t-tubules had a higher density of DHPr than did membrane from the surface sarcolemma (Brandt, 1985). The same technique was used to show that in rat, DHPr are 3 times more concentrated in the t-tubules than on the surface sarcolemma (Wibo et al., 1991). Subsequently, the development of immunohistochemical techniques, coupling confocal microscopy and specific antibodies, has been widely used to investigate the location of proteins within cardiac myocytes. The first immunohistochemical study of Ca<sup>2+</sup> channel distribution, in rabbit myocytes, showed that in ventricular cells immunostaining occurred primarily at the t-tubules (Carl et al., 1995) whereas in rabbit atrial myocytes, L type Ca<sup>2+</sup> channel staining was observed in discrete spots along the sarcolemma but was absent from the interior of the fibers (Carl et al., 1995). Therefore, it appears that the L type Ca<sup>2+</sup> channel is concentrated at the t-tubules; this has been confirmed in guinea pig (Gathercole et al., 2000) and rat (Scriven et al., 2000). Comparative studies suggest that the t-tubular concentration of the L-type  $Ca^{2+}$  channel is greater in rat ventricular myocytes than in those from the rabbit (Takagishi et al., 2000).

Thus a high density of L-type  $Ca^{2+}$  channel at the t-tubules has been reported in all immunohistochemical studies; the channel has, however, also been reported to be present in appreciable amounts on the surface membrane (Gathercole et al., 2000; Musa et al., 2002). Contrary to binding studies, where a value of  $Ca^{2+}$  channel density is achieved (t-tubules *vs* surface membrane), to date no study has quantified the  $Ca^{2+}$  channel density using

immunohistochemical technique. However, images from confocal microscopy are clear enough to confirm the concentration of  $Ca^{2+}$  channel at the t-tubules compare to cell surface.

Although binding studies and immunohistochemical data show preferential localization of the L type Ca<sup>2+</sup> channel within the t-tubules of ventricular myocytes, this may not reflect the distribution of its function, which will also depend on local environment. This is particularly true for  $I_{Ca}$  since the number of channels activated when gating charge is being recorded is ~10 times higher than the number of channels recorded when monitoring membrane current (Bean and Rios, 1989; Hadley and Lederer, 1989; Hadley and Lederer, 1991b). This can be explained if ~ 10x more channels respond to the change in membrane potential with intra-molecular movement, leading to gating charge, than switch to the open state, resulting in ionic movement and hence current (Hadley and Lederer, 1991b).

Alternative approaches have therefore been used to study the functional localization of  $I_{Ca}$  on cardiac sarcolemma. The first study of localization of function, rather than protein, was made using a double barreled microperfusion system coupled to two patch-clamp pipettes (Jurevicius and Fischmeister, 1997), which showed that  $I_{Ca}$  is uniformly distributed on the sarcolemma membrane of frog ventricular myocytes. However these myocytes lack t-tubules and this technique do not have access to the t-tubules. An alternative approach has been to use the diffusion delay between changing the bulk extracellular solution and the change in the t-tubules. In guinea pig ventricular myocytes, a rapid change of extracellular [Ca<sup>2+</sup>] leads to a biphasic change in  $I_{Ca}$ , 36% of  $I_{Ca}$  changing rapidly (~20 ms time constant) and 64% slowly (~200 ms) (Shepherd and McDonough, 1998). This indicates ~64% of functioning Ca<sup>2+</sup> channels are in the t-tubules. In atrial myocytes, which lack t-tubules, changing the bathing Ca<sup>2+</sup> produced a monophasic rapid changes of  $I_{Ca}$  (Shepherd and McDonough, 1998).

Recently a new technique has been developed, derived from scanning ion-conductance microscopy (Hansma et al., 1989), in which a patch pipette is used to scan the cell surface and monitor membrane currents (Gorelik et al., 2002). It is unclear whether this technique can monitor current within the t-tubules, although currents at the t-tubule opening can be recorded. Using this technique it has been shown that  $I_{Ca}$  is not present on the cardiac cell surface membrane, but occurs only at the entrance of the t-tubules (Gu et al., 2002). This is surprising since cell-attached patches from cardiac cells routinely contain Ca<sup>2+</sup> channels, suggesting that functional Ca<sup>2+</sup> channels are present in the surface sarcolemma (*e.g.* the visualization of sparklets by Wang et al., 2001, described in section 2.2.1).

Another approach developed recently has been to adapt the "osmotic shock" technique used previously to detubulate skeletal muscle to disrupt the t-tubules of rat ventricular myocytes (Kawai et al., 1999). T-tubules seal off within the cell and hence are physically and

electrically uncoupled from cell surface membrane. By comparing currents from detubulated and control myocytes it is possible to estimate the proportion of current within the t-tubules. Detubulation decreases cell capacitance by ~30% in good agreement with the proportion of cell membrane within the t-tubules (Page, 1978), while I<sub>Ca</sub> decreases by 75% (Kawai et al., 1999). After correction for the presence of non-detubulated myocytes, this suggests that ~90% of I<sub>Ca</sub> is concentrated within the t-tubules. A further study using the same technique, but using solutions designed to better isolate I<sub>CaL</sub>, found ~80% of I<sub>CaL</sub> within the t-tubules *i.e.* L type Ca<sup>2+</sup> channel 6 times more concentrated at the t-tubules than at the cell surface (Brette et al., 2004b).

Thus there is both structural and functional evidence that  $I_{CaL}$  is concentrated at the ttubules in mammalian ventricular myocytes. Study of the localization of T-type channels has been hampered by the lack of specific toxins or drugs, precluding binding studies, and the lack of specific antibodies. Furthermore, no functional study has been performed to date, hence it is unknown whether  $I_{CaT}$  is concentrated at the t-tubules or uniformly distributed.

 $I_{CaT}$  inactivates quickly (see section 3.1.1), consequently  $I_{CaL}$  is responsible for the long duration (plateau phase) of cardiac action potential from ventricular and atrial myocytes. In a given species, the plateau phase is longer in ventricular myocytes than in atrial cells, although it is likely that this is due to differences in currents other than  $I_{Ca}$  (e.g.  $I_{K1}$  guinea pig, (Hume and Uehara, 1985)).

The development of the patch clamp technique has led to a better understanding of the role of I<sub>CaL</sub> during the plateau phase, by using action potential voltage clamp command waveforms. I<sub>Ca</sub> during the action potential differs significantly from that elicited using square pulses (Yuan et al., 1996; Linz and Meyer, 1998a; Li et al., 1999). Figure 2 shows the time course of I<sub>CaL</sub> in three different species widely used in cardiac physiology: I<sub>Ca</sub> is more sustained during an action potential than during a square pulse, because the peak of the action potential occurs at positive potential (30-50 mV), hence Ca channels activate rapidly but the driving force for  $\mathrm{Ca}^{2+}$  is low. As the membrane potential falls,  $I_{\mathrm{Ca}}$  remains large since the membrane potential changes quicker than voltage dependent inactivation. This sustained I<sub>CaL</sub>, explaining the plateau phase, has been recorded in rat (Yuan et al., 1996), rabbit (Yuan et al., 1996; Puglisi et al., 1999), guinea pig (Grantham and Cannell, 1996; Linz and Meyer, 1998a) and human (Li et al., 1999). Yuan et al., (1996) compared Ca2+ entry during step and action potential depolarizations in rat and rabbit myocytes. They showed that during square pulses the amount of  $Ca^{2+}$  entry is higher in rat than in rabbit. This reflects differences in  $I_{Ca}$  activation and inactivation. However when an action potential waveform was used Ca<sup>2+</sup> entry was much higher in rabbit than in rat ventricular myocytes (21 vs 14 µmol/L cytosol respectively), because of the short action potential in rat myocytes. The profile of  $I_{Ca}$  during an action potential is highly modulated by Ca dependent inactivation and this will be discussed in section 3.2.1

### 2.1.2 Nodal cells

Cardiac pacemaker cells exhibit automaticity as a result of gradual depolarization of the membrane potential during electrical diastole. Work from various laboratories has shown that multiple mechanisms contribute to pacemaker activity, although the relative contribution of each current to pacemaker function is controversial (reviewed in Lipsius et al., 2001).

In atrioventricular nodal cells, both T and L type  $Ca^{2+}$  currents have been found (Liu et al., 1993; Munk et al., 1996), which play a crucial role in conduction through the node (Noma et al., 1980) (see also section 5.1.1). Since there is a paucity of data regarding the atrioventricular node, the following discussion will focus on sinoatrial node cells.

A role for I<sub>CaT</sub> in the generation of cardiac automaticity has been proposed because of the electrophysiological properties of this channel (low threshold of activation) and its high expression in this cell type. Current-clamp studies of spontaneous action potentials have shown that  $40\mu$ M Ni<sup>2+</sup>, which selectively blocks I<sub>CaT</sub>, slows the late phase of depolarization, hence slowing the firing of rabbit sinoatrial node cells (Hagiwara et al., 1988; Doerr et al., 1989; Zhou and Lipsius, 1994; Satoh, 1995). The contribution of I<sub>CaT</sub> is thought to result from the depolarizing effect of inward  $Ca^{2+}$  current. In addition, a window current (see section 3.1.2) from I<sub>CaT</sub> might participate in this process during the late phase of the pacemaker potential. Recently, it has been proposed that ICaT might also be involved in the generation of spontaneous activity by another mechanism: Huser et al., (2000) showed that I<sub>CaT</sub> could trigger local  $Ca^{2+}$  release events ( $Ca^{2+}$  sparks, see section 2.2.1). Local increases in Ca will increase inward Na/Ca exchange current which will further depolarize the cell (Huser et al., 2000). This might explain early work showing that ryanodine significantly reduced the rate of spontaneous beating of guinea pig sinoatrial node preparation (Rigg and Terrar, 1996). In contrast, there is evidence that I<sub>CaT</sub> might not be involved in sinus rhythm: a recent report showed that Ca sparks in sinoatrial pacemaker cells do not require membrane depolarization (Vinogradova et al., 2004). This is consistent with a recent report showing that mice deficient in  $\alpha_{1H}$  T type Ca<sup>2+</sup> channel (Ca<sub>v</sub>3.2 see section 4.2) do not have a significantly different heart rate (Chen et al., 2003). Since the  $Ni^{2+}$  sensitivity of  $Ca_V 3.2$  is similar to that of native  $I_{CaT}$  it was thought that Ca<sub>v</sub>3.2 was the major isoform in cardiac cells. However the mRNA of Ca<sub>v</sub>3.1, another T type Ca<sup>2+</sup> channel isoform, is present in abundance (Bohn et al., 2000) (see also section 4.2). This isoform recovers faster from inactivation (120 ms vs 400 ms for Ca<sub>V</sub>3.2, Klockner et al., 1999)

Hagiwara et al., 1988). However, to the best of our knowledge no cardiac anomalies have been reported in transgenic mice lacking Ca<sub>V</sub>3.1 (-/-) (Kim et al., 2001). Thus the role and isoform of I<sub>CaT</sub> in pacemaking cells remains unclear. I<sub>CaL</sub> may also participate in pacemaker activity. However, given the relatively positive membrane potential for activation (-40 mV) I<sub>CaL</sub> is probably only relevant in the latter phase of pace making. I<sub>CaL</sub> is also essential for the rapid phase of depolarization (upstroke, depending on the presence or absence of I<sub>Na</sub> in the preparation) and may explain the much slower action

 $I_{CaL}$  may also participate in pacemaker activity. However, given the relatively positive membrane potential for activation (-40 mV)  $I_{CaL}$  is probably only relevant in the latter phase of pace making.  $I_{CaL}$  is also essential for the rapid phase of depolarization (upstroke, depending on the presence or absence of  $I_{Na}$  in the preparation) and may explain the much slower action potential upstroke in nodal cells compared to ventricular myocytes (~2 V/sec, sinoatrial node and ~20 V/sec atrioventricular node *vs* 200 to 300 V/sec in ventricular myocytes, (Schram et al., 2002)). In sinoatrial node cells,  $I_{CaL}$  appears to be modulated by  $Ca^{2+}$ -calmodulindependent protein kinase II (CaMKII) under basal condition. Vinogradova *et al.* (2000) showed that KN93, a CaMKII specific blocker, slows spontaneous excitation in the sinoatrial node of rabbit myocytes. This effect appeared to be due to a decrease in  $I_{CaL}$  amplitude and a hyperpolarizing shift of the steady state inactivation curve, hence decreasing the  $I_{CaL}$  window current (Vinogradova *et al.*, 2000).

which is more consistent with the kinetics of recovery of the native I<sub>CaT</sub> (e.g. 140 ms in rabbit,

Transgenic mice have provided new insights into the role of HVA in cardiac pacemaking. An early study of electrocardiogram recordings from  $\alpha_{1D}$  deficient mice revealed sinoatrial node dysfunction with bradycardia and arrhythmia (Platzer et al., 2000) indicating an essential role for pacemaking in mouse heart. At this time, cell isolation from mouse sinoatrial node cells was not established. However it has since been shown by two groups that sinoatrial node cells isolated from  $\alpha_{1D}$  deficient mice show decreases in spontaneous rate of firing (Zhang et al., 2002b; Mangoni et al., 2003) due to a significant depolarizing shift in the activation threshold and a decrease in the amplitude of I<sub>CaL</sub> (Zhang et al., 2002b; Mangoni et al., 2003). Indeed,  $\alpha_{1D}$  is characterized by a more negative activation threshold than  $\alpha_{1C}$  (see Figure 1), allowing  $\alpha_{1D}$  to mediate a substantial Ca influx in sinoatrial node cells (Zhang et al., 2002b).

These studies provide functional evidence that  $I_{CaL}$  in sinoatrial node cells is due, in part, to  $\alpha_{1D}$  subunit activity and that this channel plays a role in the generation of pacemaking in mouse. To date no pharmacological tools that are selective for  $Ca_V 1.3$  ( $\alpha_{1D}$ ) L-type  $Ca^{2+}$  channels have been described, hence it is not possible to extend this finding to other species. Interestingly, it has been shown that in sinoatrial node from rabbit  $I_{CaL}$  can be activated at more negative potentials (-60 mV) than the classic potential for  $\alpha_{1C}$  (-40 mV) (Verheijck et al., 1999).

# 2.1.3 Conducting tissue

Conducting tissues in the heart are Purkinje and His cells. Both L and T type  $Ca^{2+}$  current are present in conducting tissue, with T type Ca current being fairly large compared with other cardiac preparations (Hirano et al., 1989; Tseng and Boyden, 1989; Zhou and January, 1998). For example, in dog Purkinje cells,  $I_{CaT}$  density is similar to  $I_{CaL}$  (Zhou and January, 1998).  $I_{CaL}$  density is smaller than in ventricular myocytes, consistent with the less positive plateau phase (Schram et al., 2002).  $I_{CaT}$  inhibition does not affect Purkinje fibre automaticity suggesting that  $I_{CaT}$  may not be important for conducting tissue pacemaking (Pinto et al., 1999), but, L and T type  $Ca^{2+}$  channels are important in the conduction of the action potential.

# 2.2 Excitation-contraction coupling

Contraction of striated muscle cells is due to  $Ca^{2+}$  binding to troponin C (see Bers 2001 for review). As described above, significant  $Ca^{2+}$  entry occurs during an action potential in cardiac myocytes and this  $Ca^{2+}$  might activate troponin C directly. However, in mammalian cardiac myocytes, the main mechanism is  $Ca^{2+}$  entry triggering  $Ca^{2+}$  release (CICR) from the sarcoplasmic reticulum.

### 2.2.1 Triggering sarcoplasmic reticulum Ca release

The phenomenon of CICR was initially demonstrated in skinned skeletal myocytes (Endo et al., 1970; Ford and Podolsky, 1970) and subsequently in skinned cardiac Purkinje cells (Fabiato and Fabiato, 1973; Fabiato and Fabiato, 1975), by abruptly increasing the bathing Ca<sup>2+</sup> concentration. Fabiato extensively described CICR in cardiac preparations, demonstrating that CICR is graded both by the magnitude and the rate of change of the trigger Ca<sup>2+</sup>, and provided compelling evidence for the existence of this mechanism in cardiac myocytes (Fabiato and Fabiato, 1978; Fabiato, 1982; Fabiato, 1983; Fabiato, 1985). The next step was to determine whether CICR constituted the normal Ca2+ release process in intact cardiac myocytes, and if so, which source of  $Ca^{2+}$  provided the trigger for release. During the 1980's it became clear that I<sub>Ca</sub> provided the normal trigger for Ca<sup>2+</sup> release via CICR in cardiac myocytes: several laboratories showed the similar bell shaped voltage dependence of Ca<sup>2+</sup> release and I<sub>Ca</sub> (Cannell et al., 1987; Beuckelmann and Wier, 1988; Callewaert et al., 1988). Further support came from the observation of tail currents and transients, due to I<sub>Ca</sub> inducing Ca<sup>2+</sup> release (Cannell et al., 1987; Beuckelmann and Wier, 1988): when voltage steps to +100 mV are used to induce Ca channel opening, but no Ca influx because the membrane potential is near the  $Ca^{2+}$  equilibrium potential, no Ca transient is observed (Cannell et al., 1987). However when the membrane potential is stepped back to negative voltages,  $Ca^{2+}$  enters the cell before the  $Ca^{2+}$  channels close and a Ca transient is observed. Further evidence for  $I_{Ca}$  as

the main trigger for CICR is voltage-dependent reduction of the "gain" of EC coupling (Wier et al., 1994), which was first described as maximum sarcoplasmic reticulum  $Ca^{2+}$  flux/  $I_{Ca}$  peak (Wier et al., 1994), although most investigators now use a simplification of this definition ( $Ca^{2+}$  transient amplitude/  $I_{Ca}$  peak) (see Bers, 2001 for more details). Gain decreases with voltage; Wier et al., (1994) suggested that this reflected the link between CICR and the unitary current flowing through a single  $Ca^{2+}$  channel ( $i_{Ca}$ ).

By the end of the 1980's, there was compelling evidence that CICR was the major mechanism of  $Ca^{2+}$  release in the heart and the L type Ca channel was the major trigger. However, this led to a paradox since it was unclear how such a positive feedback mechanism could be graded (Fabiato, 1985). Stern (1992) showed mathematically that global entry of  $Ca^{2+}$  leading to global  $Ca^{2+}$  release from the sarcoplasmic reticulum (common pool theory) is inherently unstable and should result in all or nothing behavior. Instead, Stern proposed a local control theory to explain  $Ca^{2+}$  release from the sarcoplasmic reticulum, in which RyRs are under the control of a single  $Ca^{2+}$  channel. He envisaged that this one  $Ca^{2+}$  channel could either regulate a single RyR ("Ca<sup>2+</sup> synapse" model) or a group of RyRs ("cluster bomb" model) in a single  $Ca^{2+}$  release unit (Stern, 1992). Each  $Ca^{2+}$  release unit operates independently and the whole-cell  $Ca^{2+}$  transient is the temporal and spatial summation of these individual localized release events.

Functional evidence supporting Stern's theory was provided with the development of confocal microscopy and new fast  $Ca^{2+}$  indicators with high contrast (*i.e.* fluo-3). Cheng et al (1993) described small, local and brief increases of intracellular Ca<sup>2+</sup>. These discrete sarcoplasmic reticulum  $Ca^{2+}$  release events by RyRs were named  $Ca^{2+}$  sparks (Cheng et al., 1993). It appears now that they are due a cluster (4-20) of RyRs operating in concert (Wang et al., 2004b). Ca<sup>2+</sup> sparks occur in a stochastic and discrete manner and are predominantly generated at the t-tubules in ventricular myocytes (Shacklock et al., 1995; Parker et al., 1996). In atrial myocytes, which lack t-tubules,  $Ca^{2+}$  sparks occur predominately at the cell periphery (Kockskamper et al., 2001; Woo et al., 2003a). Structural studies also support the local control theory. Electron microscopy and immunocytochemistry show co-localization of the L type Ca<sup>2+</sup> channel and RyRs at the junctional sarcoplasmic reticulum in several species (guinea-pig, rabbit, rat, mouse; Carl et al., 1995; Franzini-Armstrong et al., 1999; Gathercole et al., 2000; Scriven et al., 2000). In ventricular myocytes this co-localization occurs at the t-tubules and at the surface sarcolemma, with no obvious difference at the two sites (Carl et al., 1995; Franzini-Armstrong et al., 1999). The junctional cleft between the sarcolemma and the sarcoplasmic reticulum membrane results in a restricted diffusion space, termed "fuzzy space" by Lederer et al., (1990). Interestingly, this gap is the same at the cell surface and the t-tubules (Page, 1978),

although junctional sarcoplasmic reticulum is more prominent at the t-tubules (Bers, 2001). In chick and atrial myocytes, which lack t-tubules, L type  $Ca^{2+}$  channel and RyR co-localization is seen only at cell surface membrane (Carl et al., 1995; Sun et al., 1995).

The functional relationship between single Ca<sup>2+</sup> channels and RyRs was initially investigated indirectly. Early work by Lopez-Lopez et al., (1995) provided indirect evidence that a single channel activates  $Ca^{2+}$  sparks. Using verapamil to decrease the open probability. but not the amplitude, of single L type Ca currents, they showed that the probability of evoking a local release followed the voltage dependence of i<sub>Ca</sub> (Lopez-Lopez et al., 1995). Direct demonstration that a single  $Ca^{2+}$  channel can trigger a  $Ca^{2+}$  spark was a technical challenge. Gigaseal formation during patch clamp experiments leads to the formation of  $\Omega$ -shaped membrane in the pipette (Sakmann and Neher, 1995), which will disrupt the delicate coupling of Ca<sup>2+</sup> channel with RyR. Wang et al., (2001) overcame this drawback by using the loose-seal patch clamp technique, which allows formation of sufficient resistance between the cell membrane and the pipette without destroying the coupling of the two channels. Although the success rate was low, they succeeded in recording Ca<sup>2+</sup> sparks directly activated by the opening of a single  $Ca^{2+}$  channel. They recorded low amplitude  $Ca^{2+}$  events that were insensitive to ryanodine (opening of  $i_{Ca}$ ), which they named Ca<sup>2+</sup> sparklet (Wang et al., 2001), and high amplitude Ca<sup>2+</sup> events, representing Ca<sup>2+</sup> sparks, rising from the shoulder of an ongoing  $Ca^{2+}$  sparklet (see Figure 3). It is noteworthy that the demonstration of communication between the L type Ca<sup>2+</sup> channel and RyRs was demonstrated at the cell surface. Indeed, to date no technique allows recording of single  $Ca^{2+}$  channel activity within the t-tubules.

Recently, a transgenic mouse overexpressing  $\alpha_{1C}$  has been generated (Muth et al., 1999). I<sub>Ca</sub> density was increased compared with control mice but interestingly the gain function of sarcoplasmic reticulum Ca<sup>2+</sup> release remained unchanged (Song et al., 2002). This suggests that the overexpressed L type Ca<sup>2+</sup> channels are able to form functional couplings with the RyRs which are as effective as the native L type Ca<sup>2+</sup> channels in triggering sarcoplasmic reticulum Ca release.

Although any increase in  $Ca^{2+}$  in the fuzzy space is able to induce CICR from the sarcoplasmic reticulum, as demonstrated by the use of flash photolysis and caged  $Ca^{2+}$  (Valdeolmillos et al., 1989; Niggli and Lederer, 1990; Nabauer and Morad, 1990), it is now generally accepted that  $I_{Ca}$  is the major trigger for sarcoplasmic reticulum  $Ca^{2+}$  release. Alternative pathways, such as Na/Ca exchange, have been shown to be weak trigger under physiological condition (reviewed in Wier and Balke 1999).

The possibility that T-type Ca current might participate in CICR remains uncertain due to its absence in most ventricular myocytes (see section 2.1.1). In ventricular guinea-pig myocytes, in which  $I_{CaT}$  is present, it was found that sarcoplasmic reticulum  $Ca^{2+}$  release produced by  $I_{CaT}$  is delayed and slower than  $I_{CaL}$  (Sipido et al., 1998a). Similar results have been observed in Purkinje cells, where  $I_{CaT}$  is more substantial (Zhou and January, 1998). It appears therefore that a given  $Ca^{2+}$  influx through  $I_{CaT}$  is much less effective and slower than  $I_{CaL}$  in triggering sarcoplasmic reticulum  $Ca^{2+}$  release. This suggests that  $I_{CaT}$  does not belong to the junctional domain near the RyR. Specific antibodies against T-type  $Ca^{2+}$  channels are not available yet, but when available might confirm these functional studies. To the best of our knowledge, no functional investigation of  $I_{CaT}$  in CICR in atrial myocytes has been performed.

There is no information about the presence of  $\alpha_{1D}$  in working cells, and hence whether this subunit participates in EC coupling. Indirect support of non-involvement of  $\alpha_{1D}$  in EC coupling comes from Ca<sub>v</sub>1.3 -/- transgenic mice, which show normal cardiac function except bradycardia (Seisenberger et al., 2000) (see also section 2.1.2). Similarly, no information is available about a potential role for  $\alpha_{1E}$  in EC coupling (see also section 4.3).

Voltage activated Ca<sup>2+</sup> released (VACR), as in skeletal muscle, has been proposed in the heart. Although in skeletal muscle, it is clear that the L type Ca<sup>2+</sup> channel ( $\alpha_{1S}$  subunit, see section 4.1.1) is the voltage sensor for VACR, in the heart, it remain unclear if any Ca<sup>2+</sup> channel acts as the voltage sensor (see for review Ferrier and Howlett 2001). Evidence for VACR in the heart has been provided mainly by one laboratory and many technical issues have been raised against this mechanism (Wier and Balke, 1999). Most of the studies showing VACR have used a double step protocol to separately activate VACR and CICR: a first step to -40 mV to activate VACR and a second step to 0 mV to activate CICR (Ferrier and Howlett, 1995; Hobai et al., 1997; Ferrier et al., 1998; Howlett et al., 1998). However VACR requires internal cAMP, which will shifts the voltage dependence of activation of I<sub>CaL</sub> to below –40 mV (see section 3.3.1); VACR has therefore been explained by CICR due to I<sub>CaL</sub> in several studies (Piacentino, III et al., 2000; DelPrincipe et al., 2000; Griffiths and MacLeod, 2003; Trafford and Eisner, 2003; Brette et al., 2003b).

# 2.2.2 Direct activation of contraction

The role of  $Ca^{2+}$  influx in activating troponin C directly is difficult to investigate in cardiac myocytes since  $I_{Ca}$  also triggers  $Ca^{2+}$  release (above). Fabiato (1983) calculated that the role of  $Ca^{2+}$  influx in directly activating contraction is minor. Ryanodine, which blocks sarcoplasmic reticulum  $Ca^{2+}$  release, has frequently been used to investigate this issue. However, in the presence of ryanodine,  $Ca^{2+}$  dependent inactivation of  $I_{Ca}$  is smaller than

HAL author manuscript inserm-00141882, version 1

under normal conditions (see section 3.2.1), thus enhancing the possible role of  $I_{Ca}$  in directly activating contraction. Furthermore, most studies are performed at low stimulation frequency and room temperature, hence the value presented below might slightly differ from a more physiological context (higher stimulation rate and 37°C). Ryanodine has no effect on frog ventricular myocytes, but reduces contraction of rabbit (by 30%) and rat (by 80%) ventricular myocytes (Bers, 2001). These results do not show that under physiological conditions 70% of contraction is activated by I<sub>Ca</sub> in rabbit ventricular myocytes (20% in the rat), but suggest that when the sarcoplasmic reticulum is not functional  $I_{Ca}$  can provide a significant amount of  $Ca^{2+}$ for contraction (Bers, 2001). Closer analysis suggests that Ca from  $I_{Ca}$  can account for ~23% of contraction in rabbit and ~8% in rat ventricular myocytes (Delbridge et al., 1996; Delbridge et al., 1997). An approximate sequencing of cardiac muscle preparations, from most to least reliant on sarcoplasmic reticulum  $Ca^{2+}$  release: adult mouse ventricular > adult rat ventricular > dog ventricular ~ ferret ventricular > cat ventricular > neonate rat ventricular ~ rabbit atrial > human ventricular > rabbit ventricular > guinea pig ventricular > neonate rabbit ventricular > fetal ventricular (human, cat and rabbit) > frog ventricular (see Bers 2001 for details). This sarcoplasmic reticulum dependence of contraction follows structural data. In frog ventricular myocytes, the sarcoplasmic reticulum is sparse, more developed in rabbit ventricular myocytes and even more so in rat myocytes. Such a change in sarcoplasmic reticulum structure also occurs during development. Ventricular myocytes from newborn animals shows little sarcoplasmic reticulum and enhanced  $I_{Ca}$ , hence contraction in these cells relies mainly on  $Ca^{2+}$ influx directly activating contraction (Fabiato and Fabiato, 1978; Cohen and Lederer, 1988; Haddock et al., 1999). Interestingly, in a given species, atrial muscle is more CICR dependent than ventricular muscle (Fabiato, 1982).

### 2.2.3 Loading of the sarcoplasmic reticulum

Fabiato suggested that the early part of  $I_{Ca}$  triggers sarcoplasmic reticulum  $Ca^{2+}$  release and that the late part of  $I_{Ca}$  loads the sarcoplasmic reticulum with  $Ca^{2+}$  for subsequent release (Fabiato, 1985). He proposed that the sarcoplasmic reticulum has two compartments, one for uptake, one for release with a lag for diffusion of  $Ca^{2+}$  between the two compartments (Fabiato, 1985). Indeed, the pioneering work of Beeler and Reuter (1970) showed that while  $I_{Ca}$  is fully activated and does not change during repetitive depolarization, the first contraction is very small and the force increases subsequently to reach a steady state (staircase phenomena). They proposed that  $I_{Ca}$  have to flow to fill intracellular stores from which  $Ca^{2+}$ can be released during subsequent depolarizations (Beeler and Reuter, 1970). More recently, the contribution of the late phase of  $I_{Ca}$  to sarcoplasmic reticulum  $Ca^{2+}$  loading has been

demonstrated in action potential clamp experiments in guinea-pig ventricular myocytes (Linz and Meyer, 1998b), and a series of papers from Eisner's laboratory has shown that I<sub>Ca</sub> participates substantially in sarcoplasmic reticulum Ca<sup>2+</sup> loading (see for review Trafford et al., 2002). Trafford et al., (1997) demonstrated that during stimulation following sarcoplasmic reticulum Ca<sup>2+</sup> depletion, integrated I<sub>Ca</sub> was initially large but progressively become smaller (from ~14  $\mu$ mol/L cell to 8  $\mu$ mol/L cell). In parallel, Ca<sup>2+</sup> efflux from the Na/Ca exchanger increased from ~3 µmol/L cell to 8 µmol/L cell (Trafford et al., 1997). This results to a net  $Ca^{2+}$  influx during the first few contractions until a steady state is reached where  $Ca^{2+}$  influx matches  $Ca^{2+}$  efflux (Eisner et al., 2000).

# 2.3 Excitation-transcription coupling

An emerging field in cellular physiology is the link between membrane ion channels and regulation of gene expression, namely excitation-transcription coupling (ET coupling). Genome-wide screens have identified over 300 different genes and approximately 30 transcription factors that are regulated by intracellular  $Ca^{2+}$  (Feske et al., 2001). Studies over the past decade suggest that most  $Ca^{2+}$  activated transcription factors require a rise in nuclear Ca<sup>2+</sup> for activation. Recent evidence, however, mainly in neuronal studies, indicates that a rise of Ca<sup>2+</sup> in the vicinity of sarcolemmal Ca<sup>2+</sup> channels also plays a key role in regulating transcription (see for review Dolmetsch 2003).

One of the best studied transcription factors is the cAMP-responsive element binding protein (CREB). CREB binds to the cAMP response element (CRE) and to the Ca<sup>2+</sup> response element (CARE), and is thus activated by both cAMP and Ca2+ via the calmodulin and CaMKII cascade. CREB is activated by phosphorylation of several serines, including Ser133, which recruits the coactivator CREB binding protein (CBP) (Shaywitz and Greenberg, 1999). In neurons, it has been shown that CREB phosphorylation on Ser133 can occur even in the presence of EGTA, a slow Ca<sup>2+</sup> buffer that prevents nuclear Ca<sup>2+</sup> elevation but allows intracellular  $Ca^{2+}$  elevation close to the plasma membrane (Deisseroth et al., 1996; Hardingham et al., 2001) (see also section 3.2.1). Calmodulin near the L-type Ca<sup>2+</sup> channel appears to be the link between local Ca<sup>2+</sup> influx and CREB activation, by translocation of calmodulin to the nucleus (Deisseroth et al., 1998). Whether Ca<sup>2+</sup> or Calmodulin or CaMKII is translocated to the nucleus to activate CREB in cardiac cells has not yet been determined (see Figure 4). Another well characterized  $Ca^{2+}$ -dependent transcription factor is the nuclear factor of activated T cells (NFAT). NFAT plays a critical role in the activation of cells in the immune system and also in the development of the cardiovascular system (Crabtree, 2001). For example, NFAT3 translocates from the cytoplasm into the nucleus in response to dephosphorylation of several of its serines by the  $Ca^{2+}$ -calmodulin activated phosphatase calcineurin (Clipstone and Crabtree, 1992). NFAT3 appears to be important also in the development of heart failure (see section 5.2.2).  $Ca^{2+}$  can also regulate transcription without the intermediate of kinases and phosphatases, as in the case of EF hand protein downstream regulatory element antagonistic modulator (DREAM) (Carrion et al., 1999). At resting nuclear  $Ca^{2+}$  levels, DREAM is bound to DNA and suppresses transcription, possibly by preventing the interaction of transcription factors and coactivators (Ledo et al., 2002). When a DREAM tetramer binds  $Ca^{2+}$  it disrupts the interaction between DREAM and other proteins relieving the transcriptional block. These events occur in the nucleus, suggesting that inactivation of DREAM requires a nuclear  $Ca^{2+}$  elevation. Interestingly, DREAM may also have a role outside the nucleus, because its sequence is nearly identical to that of a neuronal  $Ca^{2+}$  sensor acting on K<sup>+</sup> channels (Carafoli, 2002).

A central question in excitation-transcription coupling is how cardiac cells can distinguish changes of Ca<sup>2+</sup> distinct from the cyclic variation of Ca<sup>2+</sup> at each heartbeat and convert them into specific transcriptional response. One possible answer is that transcription factors have distinct response characteristics that may refine the message of intracellular Ca<sup>2+</sup> oscillations; the time course, amplitude and spatial localization of a rise of  $Ca^{2+}$  conveys important information about the type and intensity of the stimulus. For example, in B lymphocytes, the transcriptional factor NFAT is differentially activated by brief  $Ca^{2+}$  signals of high magnitude compared with prolonged  $Ca^{2+}$  signals of lower amplitude (Dolmetsch et al., 1997). Location is also important since regulatory proteins may be confined to distinct intracellular domains (e.g. nucleus) or anchored by specific binding proteins (Pawson and Scott, 1997). For example, the nucleus experiences very different  $Ca^{2+}$  signals than the cytoplasm: a preliminary study has shown prolonged discrete  $Ca^{2+}$  release events (~2 sec duration) in the nucleus of rat ventricular myocytes, clearly contrasting with  $Ca^{2+}$  spark characteristics in the cytosol (Yang and Steele, 2004). In addition, specific isoforms of CaMKII-8 might have different locations within the cell (i.e. cytosol vs nucleus) and can be linked to the expression of gene during the hypertrophic response (Ramirez et al., 1997; Hoch et al., 1999). Privileged avenues are also important. In cultured neurons, an early study showed that  $Ca^{2+}$  influx through L type  $Ca^{2+}$  channels activates early gene expression (Murphy et al., 1991). In hippocampal slices,  $Ca^{2+}$  influx through L type  $Ca^{2+}$  channels leads to phosphorylation of CERB at Ser133 and activation of CERB-dependent transcription. It has been shown that the critical element in signaling from the L type  $Ca^{2+}$  channel is calmodulin since mutation of the calmodulin site of L type Ca<sup>2+</sup> channels reduces their ability to activate CREB and mitogen-activated protein kinases (MAPK) pathway (Dolmetsch et al., 2001). A

17

very recent study estimated that the concentration of free calmodulin in the vicinity of L-type  $Ca^{2+}$  channel is 2.5 mM, *i.e.* ~5 times the calmodulin concentration in the bulk cytoplasm (Mori et al., 2004). This has clear implication in the transduction of  $Ca^{2+}$ /calmodulin signal (*e.g.* activation of CREB, above), but also raises the question of how calmodulin is concentrated in the region of the L type  $Ca^{2+}$  channel.

In the heart, there is an emerging picture about the importance of ET coupling however  $Ca^{2+}$ -dependent transcription factors are only now beginning to be explored.

# 3 MODULATION OF CARDIAC Ca<sup>2+</sup> CHANNELS

 $Ca^{2+}$  channels play a major role in cardiac cell function; it is therefore not surprising that they can be modulated by a variety of physiological factors. In this section we will briefly describe how  $Ca^{2+}$  channels are modulated by voltage (for extensive review see McDonald et al., 1994). We will then focus on modulation by  $Ca^{2+}$ : interestingly,  $Ca^{2+}$  itself modulates  $Ca^{2+}$ channel activity, causing important negative and positive feedback. We will also describe modulation by cyclic nucleotides which occur physiologically when hormones or neurotransmitters bind to seven span transmembrane receptors. In the majority of cases, phosphorylation pathways are used to affect  $Ca^{2+}$  channel activity, however interplay between cyclic nucleotides also modifies the effects on  $Ca^{2+}$  channels.

# 3.1 Regulation by Voltage

# 3.1.1 Activation

 $Ca^{2+}$  channels are voltage-dependent with the open probability of the channel increases with depolarization. In cardiac myocytes, I<sub>Ca</sub> reaches a peak in 2~7 ms depending on the temperature and holding potential (McDonald et al., 1994). Recently, new insights into Ca<sup>2+</sup> channel activation have been provided using molecular biology techniques (see section 4). The threshold potential for activation of T type Ca<sup>2+</sup> channels is more negative than for L-type Ca<sup>2+</sup> channels. Thus L and T type I<sub>Ca</sub> can be separated by using different holding potentials: positive to – 40 mV I<sub>CaT</sub> will inactivate, I<sub>CaL</sub> will remain mainly unchanged, although the recent discovery of another L type Ca<sup>2+</sup> channel in cardiac myocytes ( $\alpha_{1D}$ ), slightly alters this notion. Figure 1 shows the current voltage relations (I-V) of the 3 major Ca<sup>2+</sup> channels in cardiac cells. The I-V curves are all bell-shaped, and T type Ca<sup>2+</sup> channels show a more negative threshold and peak than  $\alpha_{1C}$ , which explains their potential physiological role in nodal cells (see section 2.1.2).

 $Ca^{2+}$  channels can also deactivate (closure of the channel before inactivation). This occurs mainly during short square voltage clamp pulse experiments and has been used by some investigators to investigate tail currents (see section 2.2.1). T type deactivation is slower than L type, but the physiological role of  $I_{Ca}$  deactivation during the action potential remains uncertain.

## 3.1.2 Inactivation

 $I_{Ca}$  inactivation was described with the first recording, since inactivation by definition is a decrease of the current during a maintained depolarization. Both T-type and L-type Ca<sup>2+</sup> current show voltage dependent inactivation. Using a double pulse protocol, it has been shown that steady state inactivation of  $I_{CaT}$  occurs at more negative potential than  $I_{CaL}$  (see for review McDonald et al., 1994).  $I_{CaT}$  voltage dependent inactivation occurs quickly, which suggested the name "Transient" proposed by Tsien and collaborators (Nowycky et al., 1985).  $I_{CaL}$  inactivation is less marked, which accounts for the role of  $I_{Ca}$  during the plateau phase of the cardiac action potential. Species differences exist in voltage dependent inactivation. Yuan et al., (1996) showed that steady state inactivation of rat  $I_{CaL}$  occurs at more negative potentials than in rabbit. This has implications for window currents, when steady state activation and inactivation curves overlap. This defines a voltage window in which some Ca<sup>2+</sup> channels will conduct (at ~ -25 mV). Since this window occurs during the plateau phase of the cardiac action potential it has been proposed as a mechanism for arrhythmias (January and Riddle, 1989; Ming et al., 1994). A rough sequence of  $I_{CaL}$  window currents from larger to smaller is: guinea pig > rat > rabbit (Josephson et al., 1984; Yuan et al., 1996).

An "ultra slow" inactivation has also been described for  $I_{CaL}$  because it occurs on a second time scale (Boyett et al., 1994; McMorn et al., 1996). The physiological role of this inactivation remains unclear.

# 3.1.3 Facilitation

Facilitation of a current is defined by an increase in the amplitude of the current. In cardiac myocytes,  $I_{CaL}$  facilitation can occur when increasing simulation frequency or when first depolarizing the cell to a high positive potential. Rate dependent facilitation is linked to Ca and will therefore be discussed in the next paragraph. The voltage dependent facilitation described in cardiac myocytes (Pietrobon and Hess, 1990; Sculptoreanu et al., 1993a; Xiao et al., 1994a; Hirano et al., 1999; Kamp et al., 2000) is similar to that described in chromaffin cells (Artalejo et al., 1990; Artalejo et al., 1992) and skeletal myocytes (Sculptoreanu et al., 1993b). This facilitation is observed when Ba<sup>2+</sup> is used as the charge carrier (contrary to the frequency dependent facilitation, below). It appears to be mediated by cAMP dependent protein kinase and at the single channel level produces mode 2 gating behavior (long opening, Hirano et al., 1999) as does modulation by the cAMP/PKA pathway (see section 3.3.1). However, conflicting results exist: Dai et al (1999) reported that the  $\beta_{2a}$  subunit supports this facilitation, whereas the  $\alpha_2$ - $\delta$  subunit prevents it (Dai et al., 1999). The importance of the  $\beta$ 

subunit in voltage-dependent facilitation has been confirmed by Kamp et al., (2000). However, this facilitation occurs only when  $\beta_1$ ,  $\beta_3$ ,  $\beta_4$ , but not  $\beta_2$  are co expressed with  $\alpha_{1C}$  (Cens et al., 1998). A similar voltage-dependent facilitation has been observed for  $I_{CaT}$  in frog atrial cells (Alvarez et al., 1996). A G protein mechanism has been proposed to explain such facilitation in neuronal Ca channels (see for review Dolphin 2003). However another study suggests that voltage directly affects channel activity without the involvement of second messenger signals (Publicover et al., 1995).

Clearly, the mechanism(s) behind voltage-dependent facilitation of both L and T type  $Ca^{2+}$  currents are not fully elucidated, and need further investigation. Furthermore, the unphysiological test pulses (to +100 mV) required to elicit such facilitation render a physiological role unlikely.

Recently,  $I_{Ca}$  facilitation induced by a moderate depolarization of the diastolic membrane potential (in the range of -80 to - 40 mV) has also been described in guinea pig, rat, and human myocytes (Barrere-Lemaire et al., 2000; Brette et al., 2003a). Modest depolarization induces an increase in  $I_{Ca}$  amplitude and a slowing of  $I_{Ca}$  inactivation, which was blocked by ryanodine, suggesting a role for sarcoplasmic reticulum Ca release (Barrere-Lemaire et al., 2000; Brette et al., 2003a). The mechanism and physiological role of this facilitation is not elucidated yet, although it has been suggested that it can regulate cardiac automaticity in sinoatrial node cells (Mangoni et al., 2000).

# 3.2 Regulation by Calcium

 $Ca^{2+}$ -dependent inactivation of  $Ca^{2+}$  channels was first described in *Paramecium* (Brehm and Eckert, 1978) and shortly after in cardiac cells (Mentrard et al., 1984). Subsequent work showed that cardiac L type  $Ca^{2+}$  channels exhibit both voltage and Ca dependent inactivation (Kass and Sanguinetti, 1984; Lee et al., 1985; Hadley and Hume, 1987). Although an increase in intracellular  $Ca^{2+}$  can decrease  $I_{Ca}$  amplitude, a second  $Ca^{2+}$ -dependent process has been shown to potentiate  $I_{Ca}$  in the heart: ouabain can stimulate  $I_{Ca}$  in ferret ventricular muscle (Marban and Tsien, 1982), and this effect is linked to an elevation of intracellular  $Ca^{2+}$ . Subsequently, direct demonstration of  $Ca^{2+}$ -dependent facilitation of L type Ca channels has come from studies using flash photolysis of caged  $Ca^{2+}$  (Gurney et al., 1989; Hadley and Lederer, 1991a). Molecular biology has provided evidence that the same molecular determinants are involved in this facilitation and  $Ca^{2+}$ -dependent inactivation (see below).

In contrast, T type Ca channels do not show modulation by  $Ca^{2+}$  (see Perez-Reyes 2003 and section 4.2). The following discussion will focus on  $I_{CaL}$ .

# 3.2.1 Inactivation

Several studies have shown that  $Ca^{2+}$ -dependent inactivation of  $I_{CaL}$  is mediated by  $Ca^{2+}$ binding to calmodulin, which is tethered to the L type Ca channel (see 4.1.1 for details). A recent study showed that adenoviral expression of mutant calmodulin, deficient in Ca<sup>2+</sup> binding, in cultured rat myocytes results in slowing of inactivation (by  $\sim 2x$ ) of I<sub>CaL</sub> (Yang et al., 2003). Ca<sup>2+</sup> entering the cell via the channel can inactivate the channel, as demonstrated in frog myocytes, which have sparse sarcoplasmic reticulum (Argibay et al., 1988). However in mammalian heart cells the sarcoplasmic reticulum is well developed and releases a considerable amount of  $Ca^{2+}$  (see section 2.2.1). An early study suggested that sarcoplasmic reticulum Ca<sup>2+</sup> release modulates I<sub>CaL</sub> in rat ventricular myocytes (Cohen and Lederer, 1988). The participation of sarcoplasmic reticulum  $Ca^{2+}$  release in  $Ca^{2+}$ -dependent inactivation of  $I_{Cal}$ . was demonstrated by two groups in 1995. Monitoring  $I_{CaL}$  and intracellular Ca<sup>2+</sup>, Sipido et al., (1995) showed that global sarcoplasmic reticulum  $Ca^{2+}$  release inactivates  $I_{CaI}$  in a reversible fashion in guinea pig ventricular myocytes. Using very long voltage pulses (2.4 sec), I<sub>CaL</sub> inactivated (due to  $Ca^{2+}$ -dependent inactivation) and then recovered by the end of the pulse. Although the sarcoplasmic reticulum was extensively loaded with Ca<sup>2+</sup> to record such reversible inactivation, it was the first demonstration that sarcoplasmic reticulum Ca<sup>2+</sup> release modulates  $I_{CaL}$  (Sipido et al., 1995). These authors also suggested the existence of  $Ca^{2+}$ gradients in the microdomain near the  $Ca^{2+}$  channel since early inhibition of  $I_{CaL}$  exceeds intracellular Ca<sup>2+</sup> measured using Fura-2. This was confirmed by Morad's laboratory which provided striking evidence for discrete  $Ca^{2+}$  dependent inactivation of L type  $Ca^{2+}$  channel independent of bulk cytosolic Ca<sup>2+</sup> concentration (Sham et al., 1995). They proposed the term "functional coupling" between the L-type Ca2+ channel and RyR and suggested that Ca2+ dependent inactivation of  $I_{CaL}$  is due to local  $Ca^{2+}$  signaling rather than global  $Ca^{2+}$  signaling. This functional coupling was confirmed by structural data showing co-localization of the Ltype  $Ca^{2+}$  channel and RyR (see section 2.2.1). The "fuzzy space" between the L-type  $Ca^{2+}$ channel and RyR form a restricted ionic diffusion space. Sham et al (1995) showed that ryanodine significantly slowed I<sub>CaL</sub> inactivation and that this occured even in the presence of 10 mM EGTA in the pipette solution. Sham (1997) extended this study showing that I<sub>Cal.</sub> decay was not affected by the presence of 10 mM EGTA, despite abolition of the Ca<sup>2+</sup> transient and cell contraction. However, 10 mM BAPTA significantly slowed I<sub>Cal.</sub> decay (Sham, 1997). This is consistent with the existence of local  $Ca^{2+}$  signaling, occurring in a restricted space and exceeding global  $Ca^{2+}$ . The inability of intracellular EGTA to prevent sarcoplasmic reticulum Ca<sup>2+</sup> release in the fuzzy space is not surprising. Naraghi and Neher (1997) calculated that the length constant of 2 mM EGTA to chelate  $Ca^{2+}$  from the mouth of a C at 2 b HAL author manuscript inserm-00141882, version 1 si

 $Ca^{2+}$  channel is 419 nm. This is clearly above the ~12 nm separating the  $Ca^{2+}$  channel and RyR at the dyadic junction (see section 2.2.1). Interestingly, the length constant of 2 mM BAPTA is 28 nm (Naraghi and Neher, 1997). Such a difference is explained by the on rate of  $Ca^{2+}$ binding  $(k_{on} \text{ (mol}^{-1} \text{ sec}^{-1}) \text{ EGTA} < \text{BAPTA}$  by two order of magnitude). Hence, EGTA is not fast enough to chelate  $Ca^{2+}$  released from the sarcoplasmic reticulum (even at 10 mM) whereas BAPTA is. This observation was subsequently confirmed by several laboratories showing that sarcoplasmic reticulum  $Ca^{2+}$  release occurs and modulates  $I_{CaL}$  in the presence of EGTA (Masaki et al., 1997; Delgado et al., 1999; Barrere-Lemaire et al., 2000; Wu et al., 2001; Guo and Duff, 2003; Brette et al., 2003a; Brette et al., 2003b) whereas BAPTA prevents it (Sham et al., 1998; Brette et al., 2003a). The slow binding properties of EGTA have been used in conjunction with a fast, low-affinity indicator, (Oregon Green 488 BAPTA-5N) to directly measure local sarcoplasmic reticulum  $Ca^{2+}$  release flux named " $Ca^{2+}$  spikes" (Song et al., 1998). These  $Ca^{2+}$  spikes peaked in ~15 ms at 0 mV and decayed in ~50 ms. Interestingly, the difference in I<sub>Ca</sub> in the absence and presence of ryanodine showed similar kinetics (Brette et al., 2003a), emphasizing that  $I_{Ca}$  is an excellent sensor of local  $Ca^{2+}$  in the cell. When sarcoplasmic reticulum  $Ca^{2+}$  release is enhanced (*e.g.* in the presence of PKA stimulation),  $I_{Ca}$ decay can show a notch during this inactivation phase (Brette et al., 2003a; Brette et al., 2003b), indicating a recovery of  $Ca^{2+}$ -dependent inactivation, as described by Sipido et al., (1995), but on a more physiological time scales (tens of msec). This also suggests that  $Ca^{2+}$ dependent inactivation is not an absorptive state (Brette et al., 2003b; Findlay, 2004).

Local  $Ca^{2+}$  modulation of  $I_{CaL}$  has also been demonstrated in atrial myocytes in which the fuzzy space is restricted to the surface membrane (see section 2.2.1) (Sun et al., 1997; Hatem et al., 1997). In ventricular myocytes, dyadic junctions are found both at the t-tubule and the cell surface and a recent report showed that modulation of  $I_{CaL}$  by sarcoplasmic reticulum  $Ca^{2+}$  release is more pronounced at the t-tubules (Brette et al., 2004b).

 $Ca^{2+}$  dependent inactivation provides an important negative feedback to limit the amount of  $Ca^{2+}$  entry via  $I_{CaL}$ , limiting the depolarizing current and further  $Ca^{2+}$  release from the sarcoplasmic reticulum. Using action potential waveform voltage clamp to record  $I_{Ca}$ , it has been shown that  $Ca^{2+}$  dependent inactivation resulting from sarcoplasmic reticulum Ca release reduced the influx of Ca during an action potential by 30% in guinea pig ventricular myocytes (Grantham and Cannell, 1996). Of course, this effect is species-dependent, however similar results have been found in rabbit (50%, Puglisi et al., 1999) and rat (~ 40%, Takamatsu et al., 2003) ventricular myocytes. Thus if sarcoplasmic reticulum  $Ca^{2+}$  is reduced, the increase in  $I_{Ca}$ will help reload the sarcoplasmic reticulum  $Ca^{2+}$  store (as described in section 2.2.3 and reviewed in Eisner et al., 2000). Beyond considerations of EC coupling,  $Ca^{2+}$ -dependent inactivation is important in controlling cardiac excitation. Early studies showed that BAPTA, but not EGTA, inside a patch pipette, induced a significant prolongation of action potential duration, probably due to reduced  $Ca^{2+}$ -dependent inactivation which suggested a negative feed back role played by this phenomena (White and Terrar, 1992; Le Guennec and Noble, 1994). More recently, elimination of  $Ca^{2+}$ -dependent inactivation of native Ca channels by adenoviral transfection of guinea-pig ventricular myocytes with engineered calmodulins, rendered  $Ca^{2+}$  insensitive by mutation, resulted in a 4-5 fold increases in action potential duration (Alseikhan et al., 2002) consistent with this process regulating action potential duration.

### 3.2.2 Facilitation

At the same time that direct  $Ca^{2+}$ -dependent facilitation was being characterized (above) an intriguing phenomena was observed: an increase in stimulation frequency could increase  $I_{CaL}$  amplitude and slow inactivation. This was described first in frog atrium (Noble and Shimoni, 1981), and then in rat (Richard et al., 1990; Richard et al., 1993), guinea-pig (Mitra and Morad, 1986; Lee, 1987; Fedida et al., 1988a; Fedida et al., 1988b; Zygmunt and Maylie, 1990), rabbit (Hryshko and Bers, 1990), ferret (Yuan and Bers, 1994) dog (Tseng, 1988) and human (Piot et al., 1996) cardiomyocytes, although a pronounced decrease of  $I_{Ca}$  has been observed in mouse (Sipido et al., 1998b) in which the decrease might reflect insufficient time to recover from voltage dependent inactivation.

Depolarization from near physiological potential (~ -80 mV) to near the peak of the I<sub>Ca</sub>-V curve (~0 mV) produces a gradual increase and slowing of decay of I<sub>Ca</sub> (Lee, 1987), whereas when the same protocol is applied from –40 mV a progressive decline is observed (Tseng, 1988; Hryshko and Bers, 1990). Tseng (1988) characterized a holding potential threshold (-60 mV) above which increasing the frequency of stimulation leads to a decrease of I<sub>Ca</sub> amplitude.  $Ca^{2+}$  influx through the  $Ca^{2+}$  channel is the basis for this facilitation since facilitation is eliminated when Ba, Sr or Na are used as the charge carrier (Lee, 1987; Fedida et al., 1988a; Zygmunt and Maylie, 1990; Tiaho et al., 1994; Xiao et al., 1994a; Dzhura et al., 2000).

It is striking that  $Ca^{2+}$  underlies frequency-dependent facilitation of  $I_{Ca}$ , as well as causing inactivation (above). However, the mechanism underlying this facilitation is still controversial although three hypotheses have been proposed. The first hypothesis is that this facilitation is due to calmodulin tethered to the  $Ca^{2+}$  channel at the IQ motif (calmodulin binding region, see section 4 for more details). Point mutations within the IQ motif results in marked effects on Ca-dependent facilitation. (Zuhlke et al., 1999; Zuhlke et al., 2000). The second hypothesis is phosphorylation of the L-type Ca channel by CaMKII (Yuan and Bers, 1994; Anderson et al.,

HAL author manuscript inserm-00141882, version 1

1994; Xiao et al., 1994a). Evidence supporting this mechanism includes the observation that CaMKII inhibitors such as KN-62 (Yuan and Bers, 1994) inhibit facilitation. Facilitation is also eliminated by CaMKII inhibitory peptides including CamKII290-309, CaMKII273-302 (Yuan and Bers, 1994) and ICK (Xiao et al., 1994a). Interestingly, a CaMKII inhibitory peptide (AC3-I) nearly abolished I<sub>Ca</sub> facilitation but the facilitation was rescued by a peptide mimetic IQ domain, reinforcing the importance of IQ motif in Ca<sup>2+</sup> channel modulation (Wu et al., 2001). At the single channel level, constitutively active CaMKII induced longer channel opening of the L type  $Ca^{2+}$  channel (Dzhura et al., 2000). The third hypothesis is that facilitation is related to reduced Ca<sup>2+</sup> dependent inactivation due to sarcoplasmic reticulum Ca<sup>2+</sup> release at high rates. This suggestion is based on the observation that sarcoplasmic reticulum Ca<sup>2+</sup> release inhibitors blunt facilitation, highlighting the fact that facilitation and inactivation Ca<sup>2+</sup> dependent share common properties. Facilitation occurs when EGTA is used as an intracellular Ca chelator but reduced by BAPTA (Tseng, 1988; Tiaho et al., 1994; Xiao et al., 1994a; Bates and Gurney, 1999) and sarcoplasmic reticulum Ca<sup>2+</sup> release inhibitors (Ryanodine, thapsigargin, caffeine) (Tseng, 1988; Delgado et al., 1999; Wu et al., 2001; Guo and Duff, 2003). These observations emphasize the role of local  $Ca^{2+}$  signaling in  $Ca^{2+}$ dependent modulation of I<sub>CaL</sub>, but do not provide a clue which of these mechanisms, which are not mutually exclusive, underlies I<sub>CaL</sub> facilitation.

Finally, the physiological role of this facilitation remains unclear. The increase in  $Ca^{2+}$  entry might overcome the decrease in  $I_{CaL}$  due to voltage dependent inactivation following an increase in stimulation frequency and offset  $Ca^{2+}$  dependent inactivation. In rat ventricular myocytes, it has been proposed that this facilitation may participate in the increase of action potential duration following an increase in pacing rate (Fauconnier et al., 2003).

# **3.3 Regulation by cyclic nucleotides**

The most common cyclic nucleotides are cyclic adenosine monophosphate (cAMP) and cyclic guanosine monophosphate (cGMP). Although there is a consensus on the modulation of  $I_{Ca}$  by cAMP (via cAMP-dependent protein kinase A (PKA),) cGMP modulation of  $I_{CaL}$  is more contraversial.

# 3.3.1 cAMP

It is well established that in cardiac myocytes, an elevation of cAMP activates cAMP protein kinase dependent (PKA), thus phosphorylating the L type  $Ca^{2+}$  channel (Bers 2001 and see also section 4.1). In contrast, the cAMP/PKA signal pathway has no effect on T-type  $Ca^{2+}$  currents recorded from canine atrial cells (Bean, 1985), rabbit sinoatrial node cells (Hagiwara

et al., 1988), canine Purkinje cells (Hirano et al., 1989; Tseng and Boyden, 1989) or guinea pig ventricular myocytes (Tytgat et al., 1988).

In the heart, the major pathway to increase cAMP is catecholamine binding to a seven span transmembrane ( $\beta$ -adrenergic) receptor, which couples the stimulatory guanosine triphosphate (GTP) regulatory protein (Gs), stimulating adenylyl cyclase (AC) to produce (cAMP) (Bers, 2001). Other hormones can use the same pathway: for example, histamine, acting on the H2 receptor, can increase  $I_{CaL}$  via the cAMP/PKA signal pathway (Hescheler et al., 1987), and serotonin, acting on the 5-HT<sub>4</sub> receptor, which is present in human atrial myocytes (Ouadid et al., 1992) (see also section 5.1.2).

PKA phosphorylation of the L-type  $Ca^{2+}$  channel causes a dramatic increase in global  $I_{CaL}$  (~2-7 fold) and shifts activation and inactivation to more negative potentials (~10 mV) (Tsien et al., 1986; Hartzell et al., 1991). At the single channel level, PKA phosphorylation increases open probability (by switching to gating mode 2) and increases the number of channels available, although the unitary current remains the same (Yue et al., 1990). There is extensive evidence that the  $\beta$ -adrenergic/cAMP/PKA pathway upregulates  $I_{CaL}$  via phosphorylation of the channel; this is reviewed elsewhere (Tsien et al., 1986; McDonald et al., 1994).

Although an increase of intracellular cAMP leads to activation of PKA, evidence is accumulating that compartmentalization of cAMP is essential for intracellular signaling. Early work showed that PKA activation by prostaglandin E and isoprenaline exerted differential effects on glycogen phosphorylase phosphorylation in rat heart (Keely, 1979). Similarly, cAMP accumulation stimulated via the glucagon-like peptide receptor was recently found to be completely uncoupled from inotropic effects in cardiomyocytes (Vila Petroff et al., 2001), whereas glucagon can increase I<sub>CaL</sub> via the cAMP/PKA cascade in cardiac myocytes (Mery et al., 1990). The first direct evidence that cAMP signaling is spatially regulated in cardiac myocytes was obtained using the double barrelled microperfusion system described in section 2 on frog myocytes. Localized β-adrenergic receptor activation by isoproterenol caused only local activation of I<sub>Ca</sub>, suggesting localized elevation of intracellular cAMP (Jurevicius and Fischmeister, 1996). However, when AC was directly activated with forskolin, distant L-type Ca channels were also activated by PKA phosphorylation. Similar results were obtained if isoproterenol and IBMX (IsoButyMethylXanthine, a phosphodiesterase (PDE) inhibitor) were applied simultaneously. Hence the authors concluded that PDE activity around the L-type Ca channels restricted cAMP diffusion. In a subsequent study, they characterized the PDE isoforms involved by using specific inhibitors. They concluded that PDE3 and PDE4 contribute equally to the functional coupling of  $\beta$ -adrenergic receptors to nearby Ca channels via local elevations of cAMP. Such local elevations of cAMP near the cell membrane and

restriction by PDE has also been demonstrated in HEK cells expressing cyclic nucleotidegated channels to monitor cAMP concentration (Rich et al., 2001). However cAMP compartmentalization in mammalian cardiac myocytes has only recently been demonstrated directly, using FRET. Imaging of cAMP in neonatal rat cardiomyocytes showed that the noradrenaline-induced cAMP signal diffused only ~1 µm, but phosphodiesterase inhibition led to generalized cAMP elevation (Zaccolo and Pozzan, 2002). Thus, in mammalian myocytes cAMP degradation by phosphodiesterase may also spatially limit cAMP signals.

Other mechanisms have also been proposed to explain L-type Ca channel modulation by spatially compartmentalized cAMP signaling in mammalian cardiac myocytes. In mammalian heart, three different β-adrenoceptor subtypes have been cloned and identified pharmacologically to date:  $\beta_1$ ,  $\beta_2$ ,  $\beta_3$ . However, only two types of  $\beta$ -adrenergic receptors have been reported to modulate  $I_{CaL}$ ;  $\beta_1$  and  $\beta_2$ .  $\beta_1$  is the major receptor in mammalian myocytes (e.g. ~ 70% in human see for review Brodde and Michel 1999). Both  $\beta_1$ - and  $\beta_2$ - adrenergic receptors couple to G<sub>s</sub> to activate adenylyl cyclase, and stimulation of both receptor subtypes increases the intracellular level of cAMP, hence  $I_{CaL}$ . The activation of  $\beta_1$ -adrenergic receptors produces a robust increase of I<sub>CaL</sub> and also leads to phosphorylation of phospholamban present on the sarcoplasmic reticulum and of troponin I (Bers, 2001). By contrast,  $\beta_2$ -adrenergic receptor stimulation increases I<sub>CaL</sub>, but does not increase global cAMP levels or phosphorylate any non-sarcolemmal proteins. To explain this difference, Lakatta and colleagues proposed that in adult rat and dog ventricular cardiomyocytes,  $\beta_2$ - adrenergic receptors also couple to G<sub>i</sub> and that this will spatially restrict the increase in cAMP (Xiao and Lakatta, 1993; Xiao et al., 1994b; Xiao et al., 1995; Kuschel et al., 1999; Xiao et al., 1999; Chen-Izu et al., 2000). Evidence that  $\beta_2$ -adrenergic receptor stimulation leads to local signaling and phosphorylation of  $I_{CaL}$  was obtained using the cell attached configuration of the patch clamp technique.  $\beta_2$ adrenergic receptor agonists only activated I<sub>Ca</sub> if it was included in the patch pipette (local signaling) whereas  $\beta_1$ -adrenergic receptors agonists outside the patch can phosphorylate I<sub>CaL</sub> (diffusive signaling). This local signaling is achieved by the dual coupling of  $\beta_2$ -adrenergic receptors to Gs and Gi since treatment with pertussis toxin (PTX, disrupting Gi regulation) allowed  $\beta_2$ -adrenergic receptors to activate I<sub>CaL</sub> via a diffusing signal (reviewed in Xiao 2001). Comparable data were recently obtained in murine ventricular cardiomyocytes in which the  $\beta_2$ adrenergic receptor agonist zinterol failed to increase force of contraction or  $[Ca^{2+}]_i$  transient amplitude. However, after treatment of the cardiomyocytes with PTX, zinterol, via  $\beta_2$ adrenergic receptor stimulation, increased both parameters significantly (Xiao et al., 1999). Despite compelling evidences, the dual activation of Gs and Gi via  $\beta_2$ -adrenergic receptors stimulation remains controversial (Kuznetsov et al., 1995; Laflamme and Becker, 1998) and whether human cardiac  $\beta_2$ -adrenoceptors also couple to G<sub>i</sub> is unknown. In addition, recent studies identify compartmentalization of  $\beta_2$ -adrenergic receptors and AC to caveolae in neonatal and adult rat cardiomyocytes (Rybin et al., 2000). Caveolae are small (70-90 nm in diameter) invaginated foldings of the sarcolemma that are enriched in cholesterol and glycosphingolipids (see for review Razani et al., 2002). On the contrary,  $\beta_1$ -adrenergic receptors are more evenly distributed (Ostrom et al., 2001). Interestingly, M<sub>2</sub> muscarinic receptors, which can inhibit cAMP production by the Gi modulation of AC, are exluded from caveolae (Feron et al., 1997). This might contribute to the differential effect of M<sub>2</sub> muscarinic receptors activation on cAMP production from  $\beta_1$ -adrenergic receptors (decrease) and  $\beta_2$ adrenergic receptors (no change) (Aprigliano et al., 1997). Collectively, these studies indicate that components of the  $\beta_2$ -adrenergic receptor signaling pathway lead to spatial compartmentalization of cAMP signaling by restriction of receptors to membrane microdomains and/or by dual stimulation of Gs/Gi.

The idea that the receptors might not be evenly distributed on the cell surface membrane is supported by results obtained in neurons in which it has been shown that  $\beta_2$ -adrenergic receptors co-assemble with the L type Ca channel, Gs, adenylyl cyclase, PKA and phosphatase 2A (Davare et al., 2001). Whether the same co-assembly exists in cardiac myocytes is currently unknown, however some pieces of information support this idea. L type Ca<sup>2+</sup> channels are concentrated at the t-tubules in mammalians cardiac myocytes (see section 2.1.1) and immunohistochemical data shows that many of the proteins in the cAMP/PKA pathway are also located predominantly at the t-tubules (Gs protein, (Laflamme and Becker, 1999); AC (Gao et al., 1997a), PKA, (Santana et al., 2002)). Interestingly, the protein phosphatase calcineurin (enzyme counter-balancing the cAMP/PKA phosphorylation) is also concentrated at the t-tubules and co-localized with the L type Ca<sup>2+</sup> channel (Santana et al., 2002).

Finally, PKA is spatially localized via binding to A-kinase anchoring proteins (AKAP). This might explain why most studies in mammalian expression systems have failed to reproduce the observation that  $I_{Ca}$  increases upon activation of the PKA pathway (see section 3.3.1). An early study showed that mAKAP is concentrated at the t-tubules (Yang et al., 1998), however this now appears to belong to the RyR macromolecular complex (Marks, 2001). Recently, it has been shown that AKAP15 is anchored to the cardiac L-type Ca<sup>2+</sup> channel (Hulme et al., 2003). Both AKAP15 and L type Ca<sup>2+</sup> channel co-localize at t-tubules (Hulme et al., 2003). In mammalian cardiac myocytes, the cAMP microdomain might be different at the cell surface and the t-tubules. Using the detubulation technique, a recent study showed that L-type Ca<sup>2+</sup> channels are better coupled to the  $\beta$ -adrenergic pathway at the t-tubules than at the

surface membrane (Brette et al., 2004a). This indicates that  $\beta$ -adrenergic receptors are present on the surface membrane, however, to the best of our knowledge, no immunohistocemistry data are available concerning a possible  $\beta$ -adrenergic receptors sub-localization (t-tubules *vs* surface membrane) in cardiac myocytes.

In summary, modulation of  $I_{CaL}$  via cAMP appears to be highly regulated by compartmentalization, due to the existence of microdomains formed by PDE and/or receptor localization.

### 3.3.2 cGMP

cGMP is an important intracellular second messenger of various extracellular stimuli regulating L-type Ca current. cGMP is the product of guanylyl cyclase activity which is stimulated by nitric oxide (NO). In the heart, NO formation via NO synthase (NOS) can be activated via various factors (*e.g.* atrial natriuretic factor, cathecolamines acting on  $\beta_3$ adrenergic receptors, stretch; see for review Hare and Stamler 1999). It has also been proposed that acetylcholine acting via M<sub>2</sub>-receptors can induce NO synthesis, although this is controversial, in addition to activation of the Gi pathway (reviewed in Harvey and Belevych 2003, see also section 5.1.2). Classically, the regulation of L type Ca<sup>2+</sup> channel activity in the heart is due to the activity of the NOS3 enzyme subtype (see for review Hare 2003), but a recent study using knockout mice also showed that NOS1 regulates L type Ca<sup>2+</sup> channels by the signaling pathway downstream of cGMP (below) it has also been suggested that NO may have a direct effect (cGMP independent) via either nitrosylation of the L type Ca<sup>2+</sup> channel (Campbell et al., 1996; Hu et al., 1997) or by modulating the activity of G proteins (Abi-Gerges et al., 2002).

In cardiomyocytes, most studies agree that L-type  $Ca^{2+}$  current is modulated by cGMP and it is generally believed that cGMP plays a role antagonistic to that of cAMP: the "yinyang" hypothesis (Goldberg et al., 1975). However, both inhibition and stimulation of I<sub>CaL</sub> by cGMP mediated mechanisms have been reported (see table 1). The variety of results may be explained by: (i) the balance between synthesis and degradation of cGMP and cAMP; (ii) modulation of I<sub>CaL</sub> via PKG dependent phosphorylation or cross-talk with cAMP signaling via cGMP-sensitive PDE (PDE2 and PDE3); (iii) whether cGMP is able to modulate the three main target enzymes, namely PKG, PDE2 and PDE3 in a given species. The effects of each of these enzymes is discussed below and illustrated in figure 5.

**PKG** dependent modulation of I<sub>CaL</sub>

The activation of cGMP dependent protein kinase (PKG) is a function of intracellular concentration of cGMP. Direct intracellular dialysis of PKG from a patch pipette decreases  $I_{Ca,L}$  via phosphorylation of the L-type  $Ca^{2+}$  channel by PKG or an associated regulatory protein (Mery et al., 1991; Sumii and Sperelakis, 1995; Haddad et al., 1995) (see also table 1). The modulation of  $I_{Ca,L}$  by PKG has been associated with phosphorylation at serine 533 of the  $\alpha_{1C}$  subunit in *Xenopus* oocytes since replacement of serine by alanine at a consensus site (S533A) for PKG-dependent phosphorylation prevents inhibition by the PKG activator 8-Br-cGMP (Jiang et al., 2000). Phosphorylation at serine 533 of the  $\alpha_{1C}$  subunit seems to be a sufficient condition since inhibition still occurs in the absence of either  $\beta$ - or  $\alpha_2\delta$  subunits (Jiang et al., 2000, see also 4.1.1 and see Figure 6)

However, since  $Ca^{2+}$  channel phosphorylation is a balance between kinase and phosphatase activity, an alternative mechanism has been proposed in which PKG stimulates phosphatase activity, hence decreasing  $I_{CaL}$ . This hypothesis was proposed after it was shown that inhibition of  $I_{CaL}$  induced by 8-Br-cGMP in guinea pig ventricular cells was antagonized by dialysis with okadaic acid (inhibitor of phosphatases 1 and 2A) (Sakai et al., 1999). In a subsequent study, Shen and Pappano (2002) failed to observe an 8-Br-cGMP induced inhibition of  $I_{CaL}$  prestimulated by IBMX when ATP $\gamma$ S was included in the pipette. Since ATP $\gamma$ S is a kinase but not a phosphatase substrate, they concluded that, as the effect of thiophosphorylation by PKA on  $I_{CaL}$  cannot be reversed by activation of PKG, the effect occurs by dephosphorylation of the site phosphorylated by PKA (Shen and Pappano, 2002). However, using a similar approach in rat ventricular myocytes, Méry et al., (1991) observed an irreversible effect of cGMP in cells perfused with ATP $\gamma$ S and suggested that PKG phosphorylation was directed to the Ca<sup>2+</sup> channel or an auxiliary protein than to phosphatase.

There is a third hypothesis to explain the effects of cGMP on  $I_{CaL}$ . The NO donor DEANO is able to antagonize the stimulation of  $I_{CaL}$  induced by low isoproterenol concentration, via PKG activation (Abi-Gerges et al., 2001). While these results can be interpreted as a PKG-dependent phosphorylation of the channel (above), surprisingly, the inhibitory effect of DEANO on  $I_{CaL}$  was prevented by PTX, which inactivates  $G_i/G_o$  proteins (Abi-Gerges et al., 2001). This suggests a PKG mediated phosphorylation of Gi protein, as reported in expression system (Pfeifer et al., 1995).

Although PKG activation normally inhibits  $I_{Ca}$ , a PKG mediated increase in basal  $I_{Ca}$  caused by NO donors and 8-Br-cGMP has been described in rabbit newborn and juvenile ventricular and adult atrial myocytes but not in rabbit adult ventricular cells (Kumar et al., 1997; Han et al., 1998; Wang et al., 2000). This PKG mediated increase of  $I_{CaL}$  has been correlated with higher PKG expression, which could be responsible for differences in the

30

significance of cGMP in this species (Kumar et al., 1999; Wang et al., 2000). To answer this (and other) question, an alternative approach was used to investigate the specific effect of PKG. In transgenic mice in which PKG I is overexpressed in myocardium (Schroder et al., 2003), basal Ca channel activity is decreased (table 1) by both the nitric oxide donor DEANO and 8-Br-cGMP in myocytes from transgenic mice but not from wild type mice. This supports the concept that the level of PKG expression is important for basal I<sub>CaL</sub> modulation and further demonstrates the inhibitory effect on Ca<sup>2+</sup> channel activity induced by PKG (Schroder et al., 2003).

# cGMP sensitive PDEs modulation of I<sub>CaL</sub>

I<sub>CaL</sub> is stimulated by cAMP-dependent protein kinase (see above) and the cAMP level is regulated in part by the rate of degradation by PDEs. At least four PDE subtypes are present in cardiomyocytes (for review Maurice et al., 2003). Among them, PDE2 and PDE3 are stimulated and inhibited respectively by cGMP. This dual regulation of PDE by cGMP may lead to the different cGMP-dependent responses of I<sub>Ca</sub> observed in different species (Ono and Trautwein, 1991; Mery et al., 1993; Kirstein et al., 1995; Shirayama and Pappano, 1996; Vandecasteele et al., 1998) (see also table 1). For example, in human atrial cells, nanomolar concentrations of NO donors (SIN-1 and SNAP) stimulate basal I<sub>CaL</sub> (Kirstein et al., 1995; Vandecasteele et al., 1998), whereas at micromolar concentrations, the stimulatory effect of SIN-1 on I<sub>Ca</sub> was strongly attenuated (Kirstein et al., 1995). Similar effects have been found with direct intracellular application of low and high cGMP concentrations (Rivet-Bastide et al., 1997; Vandecasteele et al., 2001). These results suggest the development of a secondary inhibitory effect at higher concentrations of NO donors and cGMP and thus with a lower sensitivity to cGMP. The stimulatory effect of low cGMP concentration by activation of PKA results from inhibition of PDE3 (Kirstein et al., 1995; Vandecasteele et al., 2001). The inhibition at higher cGMP concentration occurs via stimulation of PDE2, thus decreasing PKA activity (Vandecasteele et al., 2001). The interplay between mechanisms is species dependent: in frog and human, PDE2 and 3 activation/inhibition is the major mechanism for cGMP modulation of I<sub>CaL</sub> (Mery et al., 1993; Vandecasteele et al., 2001). However in guinea-pig, PDE3 (but not PDE2) and PKG appear important. This explains the biphasic response of I<sub>CaL</sub> to cGMP, first an increase due to an inhibition of PDE3, then a decrease due to PKG phosphorylation (Ono and Trautwein, 1991; Shirayama and Pappano, 1996).

In summary, cGMP appears to produce different effects upon  $I_{CaL}$  depending on: (i) the presence and relative activities of PKG, PDE2, PDE3 in a given species; (ii) whether adenylyl and/or guanylyl cyclases are constitutively active; (iii) experimental conditions (*i.e.* the level to which the NO/cGMP/PKG cascade is stimulated, the nature and concentration of the agent

31

used to elicit a response, and whether the effect of cGMP on  $I_{Ca}$  is examined under basal conditions or after prestimulation by the cAMP cascade). This modulation can vary with development, and under pathophysiological conditions.

Compared with cAMP, little is known about compartmentalization of cGMP signaling. NOS3 localizes to caveolae (Feron et al., 1998), allowing spatial confinement of NO effects (Barouch et al., 2002). It is also tempting to speculate that if PDEs are localized with Ca<sup>2+</sup> channels, cGMP might selectively activate PDE2 and thereby decrease cAMP modulation of  $I_{CaL}$ . Some evidence to support this hypothesis has been reported, such as significant local depletion of cAMP near the L-type Ca<sup>2+</sup> channels due to activation of PDE2 by stimulation of guanylyl cyclase in frog ventricular myocytes (Dittrich et al., 2001).

# 4 BIOPHYSICAL STRUCTURE-FUNCTION OF CARDIAC Ca<sup>2+</sup> CHANNELS

Pharmacological tools have led to the classification of L-type  $Ca^{2+}$  channels, which are highly sensitive to DHPs, expressed in different tissues and heart in particular (Nowycky et al., 1985). Their high affinity for DHP agonists and antagonists made possible their biochemical characterization first in skeletal muscle (Tanabe et al., 1987) and subsequently in cardiac cells (Mikami et al., 1989). Identification of the  $\alpha_1$  subunit which forms the pore of the DHP receptor and confers its sensitivity to voltage and pharmacological components on this  $Ca^{2+}$ channel was then possible (see for review Catterall 2000). Development of molecular biology over the last decades helped to clone these subunits and to start to elucidate how  $Ca^{2+}$  channels work and are regulated at a molecular level. Ten different  $\alpha_1$  subunits for  $Ca^{2+}$  channels have been cloned with specialized biophysical and pharmacological properties and different patterns of expression. A new nomenclature, proposed in 2000, divided the  $Ca^{2+}$  channels into three sub-families (Ertel et al., 2000). The first subfamily of high voltage activated channels (HVA),  $Ca_V1$ , consists of the DHP sensitive L-type  $Ca^{2+}$  channels. The second one, of HVA, DHP insensitive,  $Ca_V2$  channels mainly expressed in brain and the last one,  $Ca_V3$ , corresponds to the T-type channels (low voltage activated channel, LVA).

HVA  $Ca^{2+}$  channels are multimers of proteins and can be composed of five subunits unlike LVA where the  $\alpha_1$  subunit would be enough to form a channel. HVA  $Ca^{2+}$  channels contain a principal transmembrane subunit  $\alpha_1$  of about 190 kDa which forms the pore of the channels and auxiliary subunits such as an intracellular  $\beta$  of 50-70 kDa depending on  $\beta$ subtype, a disulfide-linked  $\alpha_2$ – $\delta$  subunit complex of 170 kDa (Catterall 2000; Peres-Reyes 2003). Although,  $\gamma_1$ , a protein of 33 kDa is well known to co-purify with the L-type  $Ca^{2+}$ channels of skeletal muscle (Takahashi et al., 1987), it is still controversial whether recently cloned  $\gamma$ -like subunits ( $\gamma_{2-8}$ ) are real auxiliary subunits for HVA  $Ca^{2+}$  channels (Black, III, 2003; Qiao and Meng, 2003).

# 4.1 Molecular Structure of the predominant L-type Ca<sup>2+</sup> channel

# 4.1.1 Structure-function of $\alpha_{1C}$

Using homology with the skeletal muscle  $Ca^{2+}$  channel, the cardiac L type  $Ca^{2+}$  channel has been identified and cloned (Mikami et al., 1989). It is encoded by *CACNA1C* (Soldatov, 1994) and has been mapped to the distal region of chromosome 12p13 (Schultz et al., 1993).  $\alpha_{1C}$ , like other  $\alpha_1$  subunits of HVA and LVA  $Ca^{2+}$  channels, is a protein of about 190 kDa with

24 putative transmembrane segments according to its hydropathy profile. It is organized into HAL author manuscript inserm-00141882, version 1

four repeated domains (I to IV) of six transmembrane segments (S1 to S6) with intracellular Nand C- termini (Tanabe et al., 1987; Catterall, 2000) (see Figure 6). L type Ca<sup>2+</sup> channels have complex permeation properties and exhibit various selectivities for divalent and monovalent cations (see Tsien et al., 1987 for review). Indeed,  $Ca^{2+}$  is both an effective permeator and a potent blocker of L type  $Ca^{2+}$  channel. A model for the mechanism of  $Ca^{2+}$  channel permeation has been proposed by Hess and Tsien (1984) and Almers and McCleskey (1984) which proposed that  $Ca^{2+}$  ions pass though two  $Ca^{2+}$  selective sites in single file pore. Divalent cations with low binding affinity can move quickly through the channel. Those with higher affinity will also move quickly but can stick in then interfere with the movement of less affinity divalent cations (anomalous mole fraction effect, Hess et al., 1986), and also block the permeation pathway for monovalent cations (Lansman et al., 1986). Analysis of the dependence of single channel conductance on divalent cations concentration has led Yue and Marban (1990) to propose that the pore contains at least three binding sites. Key determinant in the molecular site of the selectivity filter of these channels has been demonstrated later; the selectivity filter is formed by parts of the four S5-S6 linker segments (P loop). Four conserved glutamate residues in the putative pore-lining regions of repeats I-IV, form the selectivity filter that binds  $Ca^{2+}$  in the pore of  $\alpha_{1C}$  (Tang et al., 1993; Yang et al., 1993; Cibulsky and Sather, 2000; Cibulsky and Sather, 2003). Recently, X-ray crystallographic analysis of the poreforming portions of the bacterial K<sup>+</sup> channels (Doyle et al., 1998; Jiang et al., 2002a; Jiang et al., 2003a) allowed a model of the  $Ca^{2+}$  channel outer vestibule formed by the P loops to be built (Lipkind and Fozzard, 2001) and provided evidences for the gating of voltage gated channels (Bezanilla, 2000; Jiang et al., 2002b; Jiang et al., 2003b). Hodgkin and Huxley, in 1952, hypothesized that a large flow of ions would be the result of displacement of a few charged particles whose distribution is modified by the membrane potential (Hodgkin and Huxley, 1952a). Development of electrophysiological techniques and cloning of voltage gated channels confirmed this hypothesis. The fourth transmembrane segment (S4) of voltage gated channels contains between four and eight basic residues (arginines or lysines) (see Figure 6). The S4 segments containing these positive charges were described as possibly involved in the gating structure of voltage gated channels in 1984 by Noda et al. and this has been confirmed by mutational analysis (Stuhmer et al., 1989; Papazian et al., 1991; Liman et al., 1991; Noceti et al., 1996). Because of their structural homologies, it is assumed that the gating of all voltage-gated channels, including  $Ca^{2+}$  channels, is similar (Bezanilla, 2000). When the membrane is hyperpolarized, the channel is closed and these segments lie deep in the membrane. During a depolarization, the electric field forces the positively charged S4

segments to move by a helical turn to an external position. Acidic residues in S2 and perhaps S3 seem to also be important in the change of protein conformation leading to the opening of the channel (Bezanilla, 2000; Jiang et al., 2003b). A recent study, using a mutational approach, suggests that the domains I, III and IV rather than domain II are critical for channel opening. By swapping  $Ca_V 1.2$  domains by similar  $Ca_V 3.1$  domains (LVA channel), the authors showed that these three domains contribute strongly to the voltage dependence of activation of  $Ca^{2+}$  channels but that the S4 segments did not account for their voltage dependence, suggesting roles for S1-S3 and S5-S6 segments to confer differences in voltage dependence of  $Ca^{2+}$  channels (Li et al., 2004).

The  $\alpha_1$  subunit of HVA Ca<sup>2+</sup> channels presents, in its I-II linker, a highly conserved sequence that binds the  $\beta$  subunit named AID (Alpha Interacting Domain) (Dolphin, 2003a). The AID corresponds to the following consensus sequence **QQLEEDL-GY--WITQ-E** for L-type channels and is absolutely essential for the binding of the auxiliary subunit (Pragnell et al., 1994). Although, it seems possible that  $\alpha_1$  N- and C-termini of various Ca<sup>2+</sup> channels also bind  $\beta$  (Dolphin, 2003a). Furthermore, it has been suggested that the I-II loop of Ca<sup>2+</sup> channels contains an endoplasmic reticulum retention signal that restricts the plasma membrane incorporation of  $\alpha_1$  and that  $\beta$  subunits binding to the I-II loop favors its plasma membrane expression (Bichet et al., 2000). Nevertheless, this hypothesis is controversial, since it has been proposed that  $\beta$  subunits regulate the biophysical properties of the channel without affecting the expression of  $\alpha_{1C}$  at the plasma membrane (Neely et al., 1993). A recent study shows that  $\alpha_{1C}$  surface expression, evaluated by gating currents, is not affected by co-expression of  $\beta$  subunits, suggesting that  $\beta$  binding to the I-II loop of  $\alpha_1$  subunits only affects primarily channel activity (Neely et al., 2004).

Drug binding domains have been investigated and we have now more information about molecular pharmacology of  $Ca^{2+}$  channels (Doering and Zamponi, 2003). Various compounds bind L-type  $Ca^{2+}$  channels, among them the DHPs. Mutational studies and creation of chimerical channels allowed the S5 and S6 and pore lining region S5-S6 of domains III and IV of L-type channels to be identified as the important regions responsible for DHP binding (Ito et al., 1997; Sinnegger et al., 1997; Yamaguchi et al., 2000b; Wappl et al., 2001; Lipkind and Fozzard, 2003; Yamaguchi et al., 2003) (see Figure 6).

Phenylalkylamines (D888, verapamil, D600) also block  $Ca_V 1.2$  channels (as well as  $Ca_V 1.3$  and  $Ca_V 1.4$  channels). Specific binding sites have been identified on the  $\alpha_{1C}$  IIIS6 and IVS6 regions. Mutations in these regions attenuate  $Ca^{2+}$  current block by phenylalkylamines (Hockerman et al., 1995; Johnson et al., 1996; Hockerman et al., 1997).
Benzothiazepines (*e.g.* diltiazem) are used clinically for their antihypertensive and antiarrhythmic effects. Binding to  $\alpha_{1C}$  subunit of L-type Ca<sup>2+</sup> channels involves domains III and IV (Cai et al., 1997) and more specifically the S6 segments of these domains (Kraus et al., 1996).

A depolarization from a negative to a more positive potential induces the opening of the  $Ca^{2+}$  channel. When the membrane is maintained depolarized, the channels enter a nonconductive state, an "inactivated" state. As previously described,  $Ca_V 1.2$  presents two types of inactivation: a voltage-dependent inactivation and a  $Ca^{2+}$ -dependent inactivation. Inactivation of HVA  $Ca^{2+}$  channel is not yet fully understood, but over the past years, some molecular determinants involved in this processes have been discovered. From studies of chimeras between the fast inactivating  $\alpha_{1E}$  and  $\alpha_{1C}$  and achieving point mutations in the S6 II, III, IV it was suggested that the S6 segments of each four domains of  $\alpha_{1C}$  are crucial for voltagedependent inactivation (Hering et al., 1996; Stotz et al., 2000; Berjukow et al., 2001; Stotz and Zamponi, 2001b) (see Figure 6). But transmembrane segments are not the only part of  $\alpha_{1C}$ involved in voltage-dependent inactivation since intracellular regions also appear to be implicated. The I-II loop (Stotz and Zamponi, 2001b) and the C-terminal part of  $\alpha_{1C}$  are also key determinants for voltage-dependent inactivation (Klockner et al., 1995; Soldatov et al., 1998).

Increase of intracellular Ca<sup>2+</sup> concentration accelerates inactivation of most of the HVA  $Ca^{2+}$  channels (Zuhlke et al., 1999; Shirokov, 1999; Liang et al., 2003). The C-terminus of  $\alpha_{1C}$ contains not only important determinants for voltage-dependent inactivation, but also for Ca<sup>2+</sup>dependent inactivation. It contains a putative  $Ca^{2+}$  binding site, an EF-hand region located in the proximal part of the tail, which was first described as the crucial determinant for  $Ca^{2+}$ dependent inactivation (de Leon et al., 1995). Following this study, subsequent work has concentrated on the C-terminal part of  $\alpha_{1C}$ . Zhou *et al.* (1997) identified a downstream region critical for Ca2+-dependent inactivation which does not contain the EF-hand. The major component of the C-terminus of  $\alpha_{1C}$  implicated in the Ca<sup>2+</sup>-dependent inactivation is an IQ motif located downstream of the EF-hand, which is a calmodulin-binding site, the critical Ca<sup>2+</sup> sensor for Ca<sup>2+</sup>-dependent inactivation (Zuhlke and Reuter, 1998) (see Figure 6). Thereafter, it was suggested that the EF-hand region regulates the voltage-dependent inactivation (Bernatchez et al., 1998) and supports the transduction of Ca/Calmodulin binding into channel inactivation (Peterson et al., 2000). A region of more than 100 amino-acids starting from the EF-hand including the IQ motif is implicated in Ca<sup>2+</sup>-dependent inactivation. Two more regions are involved: the first one named LA (Romanin et al., 2000) or peptide A (Pitt et al., 2001) can bind calmodulin in absence of  $Ca^{2+}$ . The preassociation of the calmodulin free of  $Ca^{2+}$  (apocalmodulin) with the L type  $Ca^{2+}$  channel has been confirmed by fluorescence resonance energy transfer (FRET) (Erickson et al., 2001). Hence, apocalmodulin is tethered to the channel facilitating the rapid triggering of  $Ca^{2+}$ -dependent inactivation (Erickson et al., 2003). A very recent report demonstrated that a single tethered calmodulin is both necessary and sufficient to produce  $Ca^{2+}$ -dependent inactivation (Mori et al., 2004).

The second region, peptide CB (Pate et al., 2000) or peptide C (Pitt et al., 2001) at high  $Ca^{2+}$  concentration can form a complex with the calmodulin bound to the IQ motif, which would constitute a minimal switch to induce  $Ca^{2+}$ -dependent inactivation (Mouton et al., 2001a).

Two models have been proposed for inactivation of  $Ca^{2+}$  channels according to these studies: (i) a constriction of the pore by the S6 segments, a C-type inactivation model (ii) and a *hinged-lid* mechanism involving intracellular gates.

Soldatov's group (Shi and Soldatov, 2002) proposed a model of a C-type/slow inactivation similar to the one proposed for K<sup>+</sup> channels (Choi et al., 1991; Lopez et al., 1994) where a constriction of the pore occurs by the S6 segments lining the intracellular part of the pore. Ca<sub>v</sub>1.2 channels reveals two inactivation components of the Ba<sup>2+</sup> current decay. The authors showed that mutations in the S6 of the four domains I-IV impairs the slow voltage-dependent inactivation and accelerate the fast inactivation of Ba<sup>2+</sup> currents, suggesting that the two mechanisms are linked. These residues at the C-terminal region of each S6 form an annular determinant which mediates the slow inactivation component of L-type Ca<sup>2+</sup> channels. The mutated channel does not exhibit Ca<sup>2+</sup>-dependent inactivation or sensitivity to  $\beta$ -subunit induced modulation of inactivation. The voltage dependence of these channels remains identical to a non-mutated channel suggesting that the voltage sensors are not affected (Shi and Soldatov, 2002). This is consistent with a C-type inactivation, where a constriction of the pore then reached by the Ca<sup>2+</sup> sensors of the C-terminus of  $\alpha_{1C}$  would be the mechanism (Shi and Soldatov, 2002; Soldatov, 2003).

In the second model, the conformational changes following a depolarization and channel opening would unmask docking sites and the C-terminus and the I-II loop would occlude the pore (Stotz et al., 2000). Two important facts are in favor of this hypothesis: the implication of the I-II loop of  $\alpha_{1C}$  which governs the voltage-dependent inactivation of Ca<sub>v</sub>1.2 channels (Stotz and Zamponi, 2001b) and the modulation of the inactivation kinetics by  $\beta$  subunits bound to this intracellular loop (Dolphin, 2003a).

It has also been suggested that voltage-dependent inactivation and  $Ca^{2+}$ -dependent inactivation use the same determinants and confirmed that the I-II loop and the C-terminus of  $\alpha_{1C}$  interact with the pore to induce in the inactivated state (Cens et al., 1998; Kim et al.,

2004). Furthermore, the gating currents recorded during depolarization (ON charges) are not impaired by intracellular  $Ca^{2+}$  while  $Ca^{2+}$ -dependent inactivation affects the return (OFF charges) of the voltage sensors to their initial state (Leroy et al., 2002; Isaev et al., 2004), confirming that the docking sites are located in the internal cavity of the channel or formed after rearrangement during a depolarization.

The molecular site responsible for the cardiac EC coupling in  $\alpha_{1C}$  (CICR, see section 2.2, *vs* voltage activated coupling of  $\alpha_{1S}$ ) has been investigated using chimeric cDNA (Tanabe et al., 1990a; Tanabe et al., 1990b). Using a preparation lacking L-type Ca<sup>2+</sup> channel (skeletal muscle from Murine Muscular Dysgenesis, *mdg*), a skeletal coupling can be restored when myotubes are injected with  $\alpha_{1S}$  and a cardiac one when  $\alpha_{1C}$  is injected. They found that replacing the cytoplasmic loop between domains II and III of the cardiac L type Ca<sup>2+</sup> channel (see Figure 6) with the skeletal coupling). This suggests that contrary to the skeletal  $\alpha_{1S}$ , the cardiac isoform does not interact directly with ryanodine receptors (RyRs) via the loop II-III.

However, a possible interaction between  $\alpha_{1C}$  and RyRs has been suggested by a functional study showing that BavK 8644 (a DHPs agonist) increased the frequency of local Ca<sup>2+</sup> release events (Ca<sup>2+</sup> sparks, see section 2.2.1) in ferret ventricular myocytes (Katoh et al., 2000). These experiments were performed in the absence of external Ca2+, ruling out a possible enhancement of  $Ca^{2+}$  sparks frequency due to  $Ca^{2+}$  entry. They proposed that binding of BayK 8644 to  $\alpha_{1C}$  reveals a weak link with RyRs, which may be due to an unknown intermediate protein (Katoh et al., 2000). Interestingly, a possible interaction between the Cterminus of  $\alpha_{1C}$  and particularly of the CB and IQ motifs, with ryanodine receptors has been described (Mouton et al., 2001b). Recently, Woo et al. (2003) showed a possible functional role for this peptide from the C-terminal tail of  $\alpha_{1C}$  (LA peptide, 1571-1599, Romanin *et al.* 2000) using atrial myocytes. In these cells,  $Ca^{2+}$  spark frequency is high near the cell surface membrane and low in the cell center. Infusion of the peptide via a patch pipette increased four fold the Ca<sup>2+</sup> spark frequency in the center (where RyRs are not in close contact with the L type  $Ca^{2+}$  channel, see section 2.2.1). They proposed that a possible interaction between the Cterminal tail of  $\alpha_{1C}$  and RyRs might sensitize RyRs, thereby increasing their open probability; although further investigations are required to explore this novel finding.

 $Ca_V 1.2$  channels are modulated by phosphorylation by various kinases (see Figure 6).  $\alpha_{1C}$  presents numerous putative phosphorylation sites for protein kinases in its N- and C-terminal regions and in the S4-S5 loop of domain II (Soldatov, 1994). Despite the fact that  $\alpha_{1C}$  is a substrate for phosphorylation by PKA in vitro (De Jongh et al., 1996; Puri et al., 1997), several

attempts to mimic the adrenergic stimulation of Ca<sub>v</sub>1.2 channels in expression systems have failed (Perez-Reyes et al., 1994; Singer-Lahat et al., 1994; Zong et al., 1995; Perets et al., 1996). Nevertheless, a few cases, the expected stimulation of Ca<sup>2+</sup> currents after cAMP increases did occur (Yoshida et al., 1992; Yatani et al., 1995). A possible missing link has been then identified as AKAPs (A kinase–anchoring proteins), which bind to the C-terminus of  $\alpha_{1C}$ (Gao et al., 1997b; Hulme et al., 2003) (see Figure 6). Furthermore, AKAPs bind PKA and would allow it to phosphorylate the ser1928 located in the C-terminus of  $\alpha_{1C}$  and to enhance channel activity (De Jongh et al., 1996; Mitterdorfer et al., 1996; Perets et al., 1996; Gao et al., 1997a; Hulme et al., 2003).

Moreover, cardiac L-type channels can be modulated by other kinases and by PKC and PKG in particular. In heterologous systems, PKC has been reported to enhance currents induced by expression of Ca<sub>v</sub>1.2 channels (Singer-Lahat et al., 1992; Bourinet et al., 1992; Shistik et al., 1998) but there are exceptions (McHugh et al., 2000). Nevertheless, the N-terminus of the  $\alpha_{1C}$  subunit seems crucial for the modulation of Ca<sub>v</sub>1.2 by this kinase (Shistik et al., 1998; McHugh et al., 2000).

A direct phosphorylation of  $\alpha_{1C}$  by PKG has been demonstrated from a functional study performed in oocytes. In this study, activation of PKG by 8-Br-cGMP inhibits the Ca<sup>2+</sup> currents and this effect does not depend on co-expression of  $\beta$  subunits. The authors showed that this effect is prevented by the PKG inhibitor KT5823 and they identified the serine 533 in the I-II intracellular linker as the phosphorylated amino acid (Jiang et al., 2000).

Alternative pre-RNA splicing generates functional and structural diversity of  $\alpha_1$  subunits of Ca<sup>2+</sup> channels (Jurkat-Rott and Lehmann-Horn, 2004). To obtain an mRNA from the pre-RNA that can be translated into a protein, the non-coding introns and alternative exons must be removed. Different isoforms of  $\alpha_{1C}$  subunits following this process have been identified. These display dissimilar biophysical and pharmacological properties. The IVS3 and the S3-S4 loop of domain IV of  $\alpha_{1C}$  are subject to alternative splicing, resulting in different subunits whose expression is tissues dependent or regulated during development (Perez-Reyes et al., 1990; Diebold et al., 1992; Feron et al., 1994; Abernethy and Soldatov, 2002). The C-terminus and the S6 segment of domain I are also subject to alternative splicing affecting current density and inactivation (Soldatov et al., 1997; Klockner et al., 1997; Goodwin et al., 1999). Sensitivity to DHPs is also modified by alternative splicing of these regions (Welling et al., 1997) and some modifications occur in IIIS2 (Soldatov et al., 1995) and IVS3 (Safa et al., 2001). Understanding of the function and regulation of splicing isoforms would present opportunities for the design of more selective therapeutic agents in the future (Jurkat-Rott and Lehmann-Horn, 2004).

#### 4.1.2 Structure-function of $\beta$ subunit

Four intracellular  $\beta$  ( $\beta_1-\beta_4$ ) subunits of about 50-70 kDa have been identified and diversity is extended by alternative splicing. The first one,  $\beta_{1a}$  has been co-purified with the DHPs receptor in skeletal muscle (Tanabe et al., 1987) and then cloned (Ruth et al., 1989). Three more genes encoding for  $\beta_2$ ,  $\beta_3$ ,  $\beta_4$  have been cloned (Perez-Reyes et al., 1992; Hullin et al., 1992; Castellano et al., 1993a; Castellano et al., 1993b), located in different chromosomes in humans: 17q21 for  $\beta_1$  (Gregg et al., 1993), 10p12 for  $\beta_2$  (Taviaux et al., 1997), 12q13 for  $\beta_3$  (Collin et al., 1993) and 2q22 for  $\beta_4$  (Taviaux et al., 1997). In human myocardium, the expression of three genes,  $\beta_1$ ,  $\beta_2$ ,  $\beta_3$  has been demonstrated at the mRNA and protein level (Collin et al., 1993; Haase et al., 1996; Freise et al., 1999). Moreover, a more recent study shows that different splice variants for  $\beta_2$  ( $\beta_{2a}$ - $\beta_{2e}$ ) and  $\beta_3$  (full length and a truncated  $\beta_3$ ) are also expressed in human heart cells (Takahashi et al., 2003; Hullin et al., 2003).

Co-expression of  $\beta$  subunits with  $\alpha_1$  subunits in heterologous expression systems, affects the HVA targeting to the plasma membrane as well as HVA biophysical properties (Walker and De Waard, 1998; Herlitze et al., 2003; Dolphin, 2003a). A model for the structure of  $\beta$  that describes the auxiliary subunits as composed of five domains (D<sub>1</sub>-D<sub>5</sub>) has been proposed (Dolphin, 2003a). D<sub>2</sub> and D<sub>4</sub> are highly conserved between the four  $\beta$  subunits but the three other domains are subject to alternative splicing. In many aspects,  $\beta$  subunits contain homologies with members of the MAGUK, a characteristic that has been confirmed by recent crystallographic analysis of the core domains of  $\beta_2$  (Van Petegem et al., 2004; Opatowsky et al., 2004). D<sub>2</sub> is a Src Homology-3 (SH3) (Hanlon et al., 1999) frequently found to bind to proline-rich motifs(McPherson, 1999). This domain is followed by a linker corresponding to D<sub>3</sub>. D<sub>4</sub> has been identified as a Guanylate Kinase (GK) domain (Hanlon et al., 1999) which was said to contain a very important molecular determinant for the binding of the  $\beta$  subunit to the I-II loop of  $\alpha_1$  subunits of HVA (Pragnell et al., 1994; De Waard and Campbell, 1995). This Beta Interaction Domain (BID) is composed of 41 amino acids highly conserved in the four  $\beta$  subunits. The consensus sequence for the BID is :

**K - - E - - - PYDVVPSMRP - - LVGPSLKGYEVTDMMQKALFDF** (De Waard et al., 1994).

However, according to a recent crystallographic analysis of the core domains of  $\beta_2$ , this region is largely buried and therefore, is unavailable for protein-protein interactions (Van Petegem et al., 2004). Previous mutations in the BID might have disrupted the folded structure and abolished AID interaction.

The N-terminal part of  $\beta_{1b}$  also presents low homology with a PDZ (Post-Synaptic Density protein 95 (PSD-95), Discs large protein (*dlg*) and Zona Occludens-1 (ZO-1)) domain present also in MAGUK (Hanlon et al., 1999) although it is not present in other  $\beta$  subunits . Proteins presenting a PDZ domain are known to traffic and cluster ion channels and receptors (Nourry et al., 2003). This suggests that the  $\beta$  subunits could be involved in trafficking of the  $\alpha_1$  subunit to the plasma membrane and would play a role in its localization with other intracellular proteins (Dolphin, 2003a). This requires further investigation since two recent studies present contradictory results. McGee *et al.* (2004) showed that the major property of the SH3-GK module of  $\beta$  subunits is to regulate their inactivation kinetics. On the contrary, Takahashi *et al.* (2004) showed, using point mutation in this module, the role in Ca<sup>2+</sup> channels trafficking.

Nevertheless, their implication in  $Ca^{2+}$  channel targeting has been suggested by studies realized in polarized epithelial cells, showing a selective and differential trafficking of  $Ca^{2+}$ channels to basolateral or apical membranes allowed by different  $\beta$  subunits (Brice and Dolphin, 1999; Bogdanov et al., 2000). The location also depends on  $\alpha_1$  subunits co-expressed, since the same  $\beta_{1b}$  subunit traffics the neuronal  $Ca^{2+}$  channels,  $Ca_V2.1$ , to the apical membrane and  $Ca_V1.2$  to the basolateral membrane (Brice and Dolphin, 1999). When expressed alone,  $\beta_{2a}$ and  $\beta_{1b}$  are associated with the membrane (Chien et al., 1995; Brice et al., 1997; Bogdanov et al., 2000), suggesting that trafficking  $Ca^{2+}$  channels to the membrane is a property of  $\beta$ subunits. However, this is not linked to its membrane association or its palmitoylation, since  $\beta_3$ and  $\beta_4$  which are not associated with the membrane are able to traffic channels (Takahashi et al., 2003). Furthermore, only  $\beta_{2a}$  is palmitoylated at its N-terminal part (Chien et al., 1996) and mutation of its palmitoylation sites does not prevent the trafficking of  $Ca_V1.2$  to the membrane (Birnbaumer et al., 1998). Moreover,  $\beta_{2e}$ , a splice variant of  $\beta_2$  which is not palmitoylated, is still associated to the plasma membrane (Takahashi et al., 2003).

Additionally, it has been proposed that the I-II linker of  $\alpha_1$  subunit contains an endoplasmic reticulum retention signal and that the binding of  $\beta$  to the AID would allow the traffic of  $\alpha_1$  from the reticulum endoplasmic to the plasma membrane (Bichet et al., 2000). Some electrophysiological studies agreed with this finding, showing that co-expression of  $\beta$  subunits increases gating currents of Ca<sub>V</sub>1.2 channels (Josephson and Varadi, 1996; Colecraft et al., 2002). These conclusions are still controversial since this was not observed in *Xenopus* oocytes (Neely et al., 1993; Olcese et al., 1996; Neely et al., 2004), but it is unclear whether this difference is due to the expression of endogenous  $\beta$  subunits expressed by oocytes (Canti et al., 2001) which would be responsible for the trafficking of over-expressed Ca<sup>2+</sup> channels.

Previously, it has been shown that a single point mutation in the  $\beta$  interaction domain of  $\alpha_{1C}$  disrupts the co-localization and plasma membrane targeting of both subunits, without affecting modulation of single channel properties or whole cell currents (Gerster et al., 1999). This suggests two independent effects of  $\beta$  subunits:  $\beta$  subunits are involved in the trafficking of Ca<sup>2+</sup> channels to the plasma membrane with the  $\alpha_1$  subunits and modulate their biophysical properties (Gerster et al., 1999).

The increase of current density can be attributed to the effect of  $\beta$  subunits on channel trafficking as well as to an increase in the open probability of the channels (Wakamori et al., 1993; Neely et al., 1995; Costantin et al., 1998; Dzhura et al., 2000). An extensive amount of work has been published concerning the effects of  $\beta$  on Ca<sup>2+</sup> channel properties and several reviews summarize these studies (Walker and De Waard, 1998; Birnbaumer et al., 1998; Dolphin, 2003a). These effects are more or less pronounced according to the  $\alpha_1$  or the  $\beta$ subunits co-expressed. However, to summarize the effects of  $\beta$  subunits on Ca<sup>2+</sup> channels, it is generally agreed that these auxiliary subunits affect their voltage dependence, hyperpolarizing their activation by increasing the coupling between the charge movement and the pore opening (Olcese et al., 1996) and hyperpolarize their steady-state inactivation. They accelerate channel activation, modify channels inactivation kinetics (either acceleration or slowing) (Walker and De Waard, 1998; Dolphin, 2003a).  $\beta_{1b}$ , and  $\beta_2$  (especially  $\beta_{2a}$  and  $\beta_{2e}$ ) decrease the inactivation rate (Cens et al., 1999; Takahashi et al., 2003) whereas  $\beta_3$  accelerates the inactivation decay of Ca<sub>V</sub>1.2 (Castellano et al., 1993b). β subunits modulate inactivation kinetics because they interact with I-II loop of the  $\alpha_1$  subunit, on a crucial part which could work as an inactivation ball (Cens et al., 1999). The retardation in inactivation kinetics of  $Ca^{2+}$  channels by  $\beta_{2a}$  has been shown to be due to its palmitoylation (Qin et al., 1998b). Palmitoylation induces immobilization of the channel inactivation gate constituted by the I-II loop of the  $\alpha_1$  subunit, through a membrane-anchoring site constituted of two palmitic acids bound to cysteines present in the N-terminus of the auxiliary subunit (Restituito et al., 2000). This work is in agreement with the model for voltage-dependent inactivation presenting the I-II loop as crucial for the voltage-dependent inactivation of  $Ca^{2+}$  channels.

Phosphorylation of  $\beta$  subunits has been suggested as a possible pathway for regulation of Ca<sup>2+</sup> channels.  $\beta_{1b}$  contains a motif that can be phosphorylated by PKA (De Jongh et al., 1989) and that  $\beta_{2a}$  is phosphorylated by the same kinase (Gerhardstein et al., 1999). It has been also suggested that  $\beta$ -adrenergic stimulation of L-type Ca<sup>2+</sup> current in cardiomyocytes was due to a phosphorylation of  $\beta$  subunits by PKA (Haase et al., 1993). Even if the mechanism for this regulation remains unclear, it seems more likely that the  $\alpha_{1C}$  subunit is the target of this kinase

to modulate the channel (Gao et al., 1997b).  $\beta$  subunits in the BID also contain as well predicted phosphorylation sites by PKC whose functional role remains unknown (Walker and De Waard, 1998), but the structure recently proposed of  $\beta$  subunits suggests that they are buried within the molecule, so their phosphorylation is unlikely to occur (Van Petegem et al., 2004).

A role for  $\beta$  subunits in other types of regulation has recently been demonstrated. Members of Ras-related GTPase subfamily (RGK family), Kir/Gem, Rem, Rem2, Rad, expressed in striated muscle cells, such as cardiac cells have been identified as interacting proteins of Ca<sup>2+</sup> channel  $\beta$  subunits (Beguin et al., 2001; Finlin et al., 2003). This interaction is stronger in the presence of GTP $\gamma$ s and antagonized by Ca<sup>2+</sup>/Calmodulin (Beguin et al., 2001). Co-expression of Rad, Rem or Kir/Gem with L-type (Ca<sub>V</sub>1.2 and Ca<sub>V</sub>1.3) Ca<sup>2+</sup> channels inhibits their plasma membrane functional expression (Beguin et al., 2001; Finlin et al., 2003). This property is inherent to  $\beta$  subunits and does not affect T-type Ca<sup>2+</sup> channels (Finlin et al., 2003).

These ancillary subunits are very important because they not only regulate biophysical properties of HVA  $Ca^{2+}$  channels but also their expression at the membrane. A possible regulation of gene silencing by a truncated  $\beta$  subunit splice variant through nuclear relocalization has been recently proposed (Hibino et al., 2003), which suggests the possible involvement of these auxiliary subunits to a new pathway of transcription, adding one more fascinating property to these subunits.

According to all these studies, it is easy to speculate that new studies about  $\beta$  subunits will reveal a specific functional role of  $\beta$  subunits in cardiac cells. For example, using the technology of adenoviral transfection in cultured myocytes, Yue's laboratory showed that overexpression of exogenous  $\beta$  subunits ( $\beta_{1b}$ ,  $\beta_{2a}$ ,  $\beta_3$  and  $\beta_4$ ) enhance I<sub>Ca</sub> density in adult rat ventricular myocytes. This suggests that  $\beta$  subunits are the limiting factor for expression of Ltype Ca<sup>2+</sup> channels in the heart (Wei et al., 2000). They also showed that  $\beta_{2a}$  is not the main  $\beta$  subunit in ventricular rat myocytes since the reconstituted currents have markedly different inactivation kinetics than in control (Colecraft et al., 2002). Such diversity of  $\beta$  subunits in the rat myocytes has been confirmed by RT PCR recently (Chu et al., 2004) and might have clear impact on electrophysiological properties of I<sub>CaL</sub>.

#### 4.1.3 Structure-function of $\alpha_2 \delta$

The  $\alpha_2\delta$  subunit was initially co-purified with  $\alpha_1$  from skeletal muscle (Nakayama et al., 1987; Catterall et al., 1988), its cDNA sequence has been then described (Ellis et al., 1988).

Our knowledge about the function of these auxiliary subunits (~175 kDa), is not as extensive as for the  $\beta$  subunits. Structurally, they are post-translationaly cleaved to yield disulphidelinked  $\alpha_2$  and  $\delta$  proteins (De Jongh et al., 1990; Jay et al., 1991) (see Figure 6). The transmembrane  $\delta$  part anchors the  $\alpha_2$  protein to the membrane via a single putative transmembrane segment and both proteins are highly glycosylate (Klugbauer et al., 2003; Canti et al., 2003). Four genetically distinct  $\alpha_2\delta$  subunits have been described so far. The gene encoding for the more characterized  $\alpha_2\delta$ -1 has been located to the human chromosomal region 7q21-q22 (Arikkath and Campbell, 2003). Three more genes have been described which encode for  $\alpha_2\delta-2$ ,  $\alpha_2\delta-3$ ,  $\alpha_2\delta-4$  located respectively to the human chromosomal regions 3p21.3, 3p21.1 and 12p13.3 (Arikkath and Campbell, 2003).  $\alpha_2\delta$ -1 demonstrates alternative splicing, resulting in five isoforms,  $\alpha_2\delta$ -1a, b, c, d, e expressed in a tissue-specific manner (Angelotti and Hofmann, 1996). Similarly, three splice variants for  $\alpha_2\delta-2$  (a, b, c) have been described (Hobom et al., 2000).  $\alpha_2\delta$ -1 is expressed in heart (Gong et al., 2001) and it has been shown that the five splice variants for  $\alpha_2\delta$ -1 are found in cardiovascular system, but  $\alpha_2\delta$ -1c and  $\alpha_2\delta$ -1d are the main isoforms expressed in mouse heart (Angelotti and Hofmann, 1996). While  $\alpha_2\delta$ -2 and  $\alpha_2\delta$ -4 are present in heart (Gao et al., 2000; Qin et al., 2002),  $\alpha_2\delta$ -3 is only found in brain (Klugbauer et al., 1999a; Gong et al., 2001). Like the  $\beta$  subunits,  $\alpha_2 \delta$  subunits are also known to exhibit two major regulatory roles. They modulate the targeting of Ca<sup>2+</sup> channels to the plasma membrane as well as their biophysical properties (Klugbauer et al., 2003; Herlitze et al., 2003). The role of  $\alpha_2\delta$  for Ca<sup>2+</sup> channels targeting to the plasma membrane is still controversial. Gao et al. (1999) suggested that its co-expression did not target  $\alpha_{1C}$  to the plasma membrane. But more studies in different heterologous expression systems demonstrated that  $\alpha_2 \delta$  increases the current density of HVA  $Ca^{2+}$  channels such as Ca<sub>V</sub>1.2 channels (Singer et al., 1991; Bangalore et al., 1996; Felix et al., 1997; Klugbauer et al., 1999a) or Ca<sub>v</sub>2.3 (Jones et al., 1998; Klugbauer et al., 1999a) and possibly of T-type channels (Dolphin et al., 1999; Dubel et al., 2004). This increase of the current density would be due to an increase of targeting of the channels to the plasma membrane (Shistik et al., 1995; Dolphin et al., 1999; Yamaguchi et al., 2000a) or to a stabilization of the channels in the membrane (Dolphin et al., 1999). Co-expression of  $\alpha_2\delta$  also modulates the biophysical properties of HVA (Klugbauer et al., 2003). These auxiliary subunits shift their voltage dependence of activation and inactivation to more negative potentials (Felix et al., 1997; Shirokov et al., 1998; Hobom et al., 2000) and promote inactivation of gating currents (Shirokov et al., 1998). The region of the  $\alpha_2\delta$  subunit involved in the modulation of the gating properties of the HVA  $Ca^{2+}$  channels is localized in the  $\delta$  domain of the protein (Felix et al., 1997). They increase the gating current size for Ca<sub>V</sub>1.2 (Bangalore et al., 1996) and Ca<sub>V</sub>2.3 (Jones et al., 1998). This increase is correlated to the enhancement of whole cell ionic current, suggesting that these subunits induce Ca<sup>2+</sup> channel targeting to the plasma membrane. They also accelerate their activation and inactivation rate (Bangalore et al., 1996; Shirokov et al., 1998; Qin et al., 1998a). A possible role of modulation of T-type channels by  $\alpha_2\delta$  has been also suggested. Its co-expression with  $\alpha_{1G}$  hyperpolarizes its voltage dependence and accelerates the decay of the currents and shifts the steady-state inactivation curve (Hobom et al., 2000).

Even if its modulatory role is still controversial,  $\alpha_2\delta$  (Klugbauer et al., 2003; Herlitze et al., 2003) might play a major role to lead to the correct targeting of Ca<sup>2+</sup> channels to the plasma membrane and modulate, in cooperation with  $\beta$  subunits, the Ca<sup>2+</sup> entry in cardiac cells.

#### 4.1.4 Structure-function of $\gamma$ .

 $\gamma$  subunit was originally thought only to be associated with L-type Ca<sup>2+</sup> channels in skeletal muscle where it was first identified (Glossmann et al., 1987). Its primary structure was then described (Jay et al., 1990). This first member of  $\gamma$  subunits,  $\gamma$ 1, due to low level of expression of  $\alpha_{1S}$ , has been co-expressed with Ca<sub>V</sub>1.2 channels in heterologous systems. It shifts the voltage dependence of inactivation to more negative potentials and also accelerates inactivation kinetics (Singer et al., 1991; Eberst et al., 1997). So far, 8  $\gamma$  subunit ( $\gamma$ 1-8) isoforms have been identified (Kang and Campbell, 2003).  $\gamma$ 1 and  $\gamma$ 6 are expressed in skeletal muscle and  $\gamma$ 2-5 and  $\gamma$ 7-8 are mainly expressed in brain (Burgess et al., 1999; Chu et al., 2001). The presence of  $\gamma 4$ ,  $\gamma 6$  and  $\gamma 7$  has been reported in heart cells (Chu et al., 2001). More recently, a new  $\gamma$ 7 has been cloned, and found to exhibit a longer C-terminus than previously predicted and shows striking properties (Moss et al., 2002). Co-expression of this neuronal protein with Ca<sup>2+</sup> channels reduces expression of channel proteins, specifically abolishes Ntype currents but also affects  $Ca_V 1.2$  channels (Moss et al., 2002). This suggests that this  $\gamma$ -like protein would not be associated with a functional channel at the plasma membrane but regulates their expression by a mechanism which remains to be elucidated. Accordingly, a recent study of the three-dimensional structure of cardiac L-type channels determined by single particle electron microscopy reveals differences in the putative transmembrane region with the structure of skeletal L-type channels (Wang et al., 2004a). This is consistent with the lack of an associated  $\gamma$  subunit (Wang et al., 2004a). So far, the expression of  $\gamma$  subunits in heart cell is not well established and their possible physiological role is not clear.

### 4.2 T-type Ca<sup>2+</sup> channels

To date, three T-type  $Ca^{2+}$  channels have been cloned (Perez-Reyes, 2003). After many unsuccessful attempts to clone T-type channels by PCR, progress in the sequencing of human, yeast and C. elegans genomes provided a novel library that could be screened with a computer (*i.e. in silico*). This led to the cloning of the first T-type channel, Ca<sub>v</sub>3.1, by Perez-Reves *et al.* (1998) which exhibited similar biophysical properties to T-type currents recorded in neurons. The full length human Ca<sub>V</sub>3.1 was subsequently cloned (Monteil et al., 2000a). The gene encoding this channel, CACNA1G has been mapped in the human to chromosome 17q22 (Perez-Reyes, 1998). The screening of a human heart library allowed the cloning of Ca<sub>V</sub>3.2, whose gene, CACNA1H, has been mapped to human chromosome 16p13.3 (Cribbs et al., 1998). Screening of rat brain led to identification of the rat version of Ca<sub>v</sub>3.1, Ca<sub>v</sub>3.2 and of Ca<sub>v</sub>3.3 (Lee et al., 1999a). The gene encoding for Ca<sub>v</sub>3.3, CACNA11, has been located in the human on chromosome 22q12.3-13.2 (Mittman et al., 1999b). The structure of the  $\alpha_1$  subunit for T-type channels is similar to those of HVA  $Ca^{2+}$  channels. They are proteins of 24 transmembrane segments arranged in four domains of six segments with a positively charged S4. The S4 segments confer to the channels, their sensitivity to variation of the membrane potential (Perez-Reves, 2003). In HVA Ca<sup>2+</sup> channels, four glutamate residues in each Pregion form the selectivity filter that binds  $Ca^{2+}$  (Tang et al., 1993; Yang et al., 1993; Cibulsky and Sather, 2000; Cibulsky and Sather, 2003). For LVA Ca<sup>2+</sup> channels, the selectivity filter consists of an arrangement of two glutamate and two aspartate residues in the P-loops of domains I-IV of the  $\alpha_1$  subunit (Talavera et al., 2001). Point mutations of these divergent residues led to the conclusion that they are critical for the selectivity and the permeation properties of Ca<sub>v</sub>3.1 (Talavera et al., 2001). The activation kinetics and their sensitivity to protonation is also dependent on these residues (Talavera et al., 2003a; Talavera et al., 2003b). Of course, as with Ca<sub>V</sub>1.2, S4 segments in domains I, III, IV are critical for Ca<sub>V</sub>3.1 opening (Li et al., 2004). Furthermore, S6 segments are implicated in the voltage-dependent inactivation (Marksteiner et al., 2001) as well as the carboxy-terminus (Staes et al., 2001; Park et al., 2004), the I-II and II-III intracellular loops (Park et al., 2004). A recent study on Ca<sub>V</sub>3.2 Ca<sup>2+</sup> channels underlined the importance of several domains, IS2-S3, IS5-S6 and IIS2, in the activation processes of these channels (Khosravani et al., 2004). Point mutations in the IS2-S3 and IS5-S6 hyperpolarized the activation and a mutation in the IIS2 slowed their activation. The S2 of domain II is also implicated in the voltage dependent inactivation of LVA Ca<sup>2+</sup> channels (Khosravani et al., 2004).

The currents mediated by rat and human isoforms of the three Cav3.x channels exhibit different sensitivity to nickel. Nickel is ten fold more potent to inhibit T-type currents conducted through Ca<sub>V</sub>3.2 channels than currents conducted through Ca<sub>V</sub>3.1 or Ca<sub>V</sub>3.3 (Lee et al., 1999b). These three channels begin to activate at very negative potentials, around -70 mV (Perez-Reyes, 2003) (see Figure 1). Currents mediated by Ca<sub>V</sub>3.1 show the fastest activation and inactivation kinetics (Klockner et al., 1999; Monteil et al., 2000a; Chemin et al., 2002). The three channels exhibit no differences in the voltage dependence of their steady-state inactivation or their activation (Klockner et al., 1999). Nevertheless, a study comparing the three human variants of T-type channels showed that these properties are slightly hyperpolarized for Ca<sub>v</sub>3.1 and Ca<sub>v</sub>3.2 in comparison to Ca<sub>v</sub>3.3 (Chemin et al., 2002). Ca<sub>v</sub>3.3 deactivates faster than the other two channels and recovery from short term inactivation is three-fold slower for currents mediated by Cav3.3 and Cav3.2 than for Cav3.1 (Klockner et al., 1999). In contrast, reactivation after a long prepulse is faster for  $Ca_V 3.3$  and slowest for  $Ca_V 3.2$ (Klockner et al., 1999; Chemin et al., 2002). Furthermore, Ca<sub>v</sub>3.3 is the only channel exhibiting voltage-dependent facilitation (Klockner et al., 1999; Chemin et al., 2002). These LVA Ca<sup>2+</sup> channels, unlike the L-type channels have been reported to lack Ca<sup>2+</sup> binding domains in their C-terminal region so they do not exhibit Ca<sup>2+</sup>-dependent inactivation (Perez-Reves, 1998). Nevertheless, Ca<sub>v</sub>3.2 but not Ca<sub>v</sub>3.1 or Ca<sub>v</sub>3.3 exhibits slower activation and inactivation kinetics when  $Ca^{2+}$  is the charge carrier, suggesting a possible modulation by  $Ca^{2+}$ of this channel (Klockner et al., 1999). They differ also from HVA Ca<sup>2+</sup> channels in what their intracellular I-II loop does not contain an AID responsible for binding  $\beta$  subunits (Dolphin, 2003a). Co-expression of cloned  $\beta$  subunits has little or no effect on cloned T-type channel activity, unlike  $\alpha_2 \delta$  which can induce an increase of the current density (Dolphin et al., 1999; Hobom et al., 2000; Gao et al., 2000; Dubel et al., 2004). y subunits have very little effect on T-type channels.  $\gamma_2$  but not  $\gamma_3$  or  $\gamma_4$  slows the decay of the tail currents recorded after coexpression with Ca<sub>V</sub>3.3 (Green et al., 2001). It has also been reported that  $\gamma_2$ ,  $\gamma_4$  and  $\gamma_5$ accelerate inactivation of Ca<sub>V</sub>3.1 and hyperpolarize its steady-state inactivation (Klugbauer et al., 2000). More recently, it has been proposed that  $\alpha_2\delta$ -2a and  $\gamma_5$  improve the coupling between the charge movement and the opening of Ca<sub>v</sub>3.1 expressed in HEK293 cells, supporting the hypothesis that T-type channels may interact with auxiliary subunits (Lacinova and Klugbauer, 2004).

Heart cells express  $Ca_V 3.1$  channels as well as  $Ca_V 3.2$  (Cribbs et al., 1998; Monteil et al., 2000b), while  $Ca_V 3.3$  is mainly expressed in brain (Lee et al., 1999a). Based on nickel sensitivity, it has been proposed that the T-type  $Ca^{2+}$  current in cardiac cells is mainly carried through  $Ca_V 3.2$  channels (Hagiwara et al., 1988; Pascarel et al., 2001). However, this issue

remains controversial and recent reports present evidence supporting  $Ca_V3.1$  as the functional cardiac T-type  $Ca^{2+}$  channel during murine embryonic period (Cribbs et al., 2001) whereas another one suggests this role for  $Ca_V3.2$  (Niwa et al., 2004). As for HVA  $Ca^{2+}$  channels, alternative splicing increases functional and structural diversity of LVA  $Ca^{2+}$  channels (Jurkat-Rott and Lehmann-Horn, 2004). Splice variants of  $Ca_V3.1$  channels differ in their II-III loop which modifies their inactivation kinetics and steady-state inactivation (Mittman et al., 1999a; Chemin et al., 2001). Deletion of seven amino acids residues in the linker between domain III and IV of  $Ca_V3.1$ , by alternative splicing, induces a positive shift of its activation and inactivation and slows its activation kinetics (Monteil et al., 2000a; Chemin et al., 2002). The C-terminus of  $Ca_V3.3$  is also modified by alternative splicing, leading to channels with truncated C-termini and with altered inactivation properties (Murbartian et al., 2002).

It is clear that endogenous T-type currents are subject to hormonal regulation and are modulated by guanine nucleotides and protein kinases in various type of cells (Perez-Reves, 2003; Yunker, 2003). However, less is known about the molecular determinants involved in their modulation. Cloned T-type Ca<sup>2+</sup> channels harbor many consensus phosphorylation sites for different kinases (Cribbs et al., 1998; Lee et al., 1999a; Mittman et al., 1999a; Klugbauer et al., 1999b; Mittman et al., 1999b; Chemin et al., 2001). So far, only the Ca<sub>V</sub>3.2 channels have been identified as subject to modulation by intracellular messengers when expressed in heterologous systems. They are modulated by CAMKII which phosphorylates a serine present in their II-III intracellular loop, which is absent from unregulated Ca<sub>v</sub>3.1 channels (Wolfe et al., 2002; Welsby et al., 2003). Phosphorylation by CAMKII increases activity of Ca<sub>V</sub>3.2 channels at negative potentials, hyperpolarizing their activation (Wolfe et al., 2002; Welsby et al., 2003). These T-type channels are also modulated by PKC which leads to a three-fold increase in the amplitude of barium currents recorded in oocytes (Park et al., 2003). They are also inhibited by G-protein βγ subunits that bind to their II-III loop (Wolfe et al., 2003). Future studies on cloned LVA Ca<sup>2+</sup> channels will certainly reveal more ways to regulate T-type channels and therefore, the activity of pace-maker cells and cardiac function.

## 4.3 Molecular structure of others $Ca^{2+}$ channels ( $\alpha_{1D}$ , $\alpha_{1E}$ )

 $Ca_V 1.2$  channels are known to be predominantly expressed in heart cells, although,  $\alpha_{1C}$  is not the only  $\alpha_1$  subunit present in cardiomyocytes, since two more subunits,  $\alpha_{1D}$  (Ca<sub>V</sub>1.3) and  $\alpha_{1E}$  (Ca<sub>V</sub>2.3) have been detected in heart tissues. RNA encoding  $\alpha_{1D}$  has been detected in these cells and expression of Ca<sub>V</sub>1.3 channels at the plasma membrane was confirmed using antibodies (Takimoto et al., 1997; Wyatt et al., 1997; Xu et al., 2003). It has been suggested that they play a role in cardiac pacemaker activity (see section 2.1.2) and during development of embryonic hearts (Xu et al., 2003). Ca<sub>V</sub>2.3 channels, have also been detected at RNA and protein levels by RT-PCR and immunodetection in rat atrial and ventricular myocytes (Weiergraber et al., 2000; Mitchell et al., 2002). The functional significance in cardiac myocytes (if any) of this Ca<sup>2+</sup> channels remains undeterminated. Transgenic mice lacking Ca<sub>V</sub>2.3 (-/-) did not show heart defects at the adult stage (*e.g.* bradychardia), although an increase in the variation of beating frequency lead to the suggestion that Ca<sub>V</sub>2.3 might be implicated in stabilizing the heart beat of prenatal murine hearts (Lu et al., 2004).

Like  $\alpha_1$  subunits from other Ca<sup>2+</sup> channels,  $\alpha_{1D}$  and  $\alpha_{1E}$  are proteins of about 2000 amino acid residues organized into four domains of six transmembrane segments, with the positively charged S4 conferring sensitivity to potential changes.

Only a few studies describing the biophysical properties and modulation of Ca<sub>V</sub>1.3 channels have been published. These channels activates at relatively hyperpolarized potentials, depending on the studies. The use of different extracellular concentrations of divalent ions, the expression of different cloned channels from different types of cells could explain these differences as well as the co-expression of different or no auxiliary subunits (Xu and Lipscombe, 2001; Koschak et al., 2001; Bell et al., 2001; Scholze et al., 2001; Xu et al., 2003). But several groups confirmed that they activate at more negative potentials than Ca<sub>V</sub>1.2 (Xu and Lipscombe, 2001; Koschak et al., 2001; Zhang et al., 2002b; Kasielke et al., 2003) (see Figure 1). A hyperpolarizing shift of the steady-state inactivation curve for Ca<sub>v</sub>1.3 compared with Ca<sub>v</sub>1.2 currents has also been observed (Koschak et al., 2001; Scholze et al., 2001). The inactivation kinetics of barium currents induced by expression of Cav1.3 channels are slower when  $Ca^{2+}$  was used as charge carrier, so these channels exhibit a  $Ca^{2+}$  dependent inactivation (Xu and Lipscombe, 2001; Bell et al., 2001; Koschak et al., 2003). Not much is known about the molecular determinants implicated in inactivation of these channels. Nevertheless, it is easy to speculate that the mechanism leading to  $Ca^{2+}$ -dependent inactivation is similar to that for Ca<sub>V</sub>1.2 channels since the  $\alpha_{1D}$  C-terminus is highly homologous to the one for Ca<sub>V</sub>1.2 channels (Koschak et al., 2001).

Like other L-type channels,  $\alpha_{1D}$  has an AID in its I-II loop (Dolphin, 2003a) and coexpression of a  $\beta$  and  $\alpha_2$ - $\delta$  subunits increases the current density but does not modify the biophysical properties of the channel (Williams et al., 1992; Xu and Lipscombe, 2001).

These L-type channels exhibit sensitivity to DHPs antagonists. They are blocked by nimodipine (Safa et al., 2001) and nitrendipine, but the neuronal isoform is less sensitive than  $Ca_V 1.2$  channels (Xu and Lipscombe, 2001). They are also blocked by nifedipine and activated by the DHPs agonist BayK8644 (Koschak et al., 2001; Bell et al., 2001). Isradipine block of a pancreatic human  $Ca_V 1.3$  is 8.5-fold lower than that for  $Ca_V 1.2$ , which can be attributed to

differences in the voltage-dependent interaction of DHPs antagonists (Koschak et al., 2001). Different sensitivity to DHPs has also been described for two splice variants cloned from embryonic cardiomyocytes (Xu et al., 2003).

Expression of Ca<sub>V</sub>2.3 in heterologous systems produces current activating and inactivating at more negative potentials than Ca<sub>V</sub>1.2 channels (Berrou et al., 2001). The biophysical properties of Ca<sub>v</sub>2.3 channels are intermediate between those of T-type and L-type channels. They activate at more positive potentials than T-type channels and their steady-state inactivation occurs at more positive potentials than for Ca<sub>V</sub>3.2 but more negative than for Ca<sub>v</sub>3.1 and Ca<sub>v</sub>3.3 (Lee et al., 1999a). Inactivation kinetics of barium currents induced by Ca<sub>V</sub>2.3 expression is faster than for Ca<sub>V</sub>1.2 or Ca<sub>V</sub>3.1 but slower than for Ca<sub>V</sub>3.2 and Ca<sub>V</sub>3.3 (Lee et al., 1999a; Berrou et al., 2001). As for  $\alpha_{1C}$ -containing channels, multiple sites within  $\alpha_{1E}$  are implicated in the voltage-dependent inactivation of Ca<sub>V</sub>2.3 channels. The voltagedependent inactivation involves the I-II loop of  $\alpha_{1E}$  (Stotz et al., 2000; Berrou et al., 2001) as well as the S6 segments of the domain II and III suggesting that as for L-type currents, the I-II linker forms a hinged lid that may dock to the S6 segments of the pore forming subunit (Stotz et al., 2000; Stotz and Zamponi, 2001a). Co-expression of auxiliary subunits modulates Cav2.3 channels. Like other HVA  $Ca^{2+}$  channels, they contain an AID which binds  $\beta$  subunits (Dolphin, 2003a). The main effect of  $\alpha_2\delta$  subunit is to increase current density and  $\beta$  subunits also increase the number of channels expressed at the plasma membrane, hyperpolarize the voltage-dependence of their activation and inactivation and slows their inactivation kinetics (Olcese et al., 1996; Parent et al., 1997; Jones et al., 1998). Mutations of the AID of  $\alpha_{1E}$  affects the kinetics and the voltage-dependence of inactivation of  $Ca_V 2.3$ , reinforcing the hypothesis of involvement of the I-II loop and beta subunits as crucial determinants for inactivation of Ca<sup>2+</sup> channels (Berrou et al., 2001; Berrou et al., 2002). Since it was first cloned (Schneider et al., 1994; Williams et al., 1994), it has been suggested that the inactivation of Cav2.3 is not modulated by intracellular Ca<sup>2+</sup>. Therefore,  $\alpha_{1E}$  has been used to generate  $\alpha_{1C}$ - $\alpha_{1E}$  chimera to study the Ca<sup>2+</sup>-dependent inactivation of L-type channels (de Leon et al., 1995; Parent et al., 1997; Zhou et al., 1997). However, more recently, different studies describe opposite effects of  $Ca^{2+}$  on  $Ca_V 2.3$  activity. At concentrations between 0.1 and 1 µmol/L,  $Ca^{2+}$  increases the current density and slows inactivation kinetics by activating a PKC (Leroy et al., 2003; Klockner et al., 2004). This modulation occurs only for neuronal splice variants that contain an arginine-rich region in their intracellular II-III loop, while neuroendocrine/heart isoforms are not regulated by Ca<sup>2+</sup> or PKC (Pereverzev et al., 2002; Klockner et al., 2004). Another group showed that calmodulin binds to the IQ motif present in the C-terminus of Ca<sub>V</sub>2.3 (Erickson et al., 2001) and that these channels undergo  $Ca^{2+}$ -dependent inactivation, which is mediated by 50 the calmodulin N-terminal lobe, when a low concentration of  $Ca^{2+}$  chelator is present (Liang et al., 2003). Activity of  $Ca_V 2.3$  channels, at low  $Ca^{2+}$  concentration (between 0.1 and 1 µmol/L) are potentiated by activation of a PKC by  $Ca^{2+}$  while  $Ca^{2+}$ -dependent inactivation occurs at higher internal  $Ca^{2+}$  concentrations.

Unlike the other HVA  $Ca^{2+}$  channels, no pharmacological compounds have been identified as specific blockers of the  $Ca_V 2.3$  channels. Expression of  $\alpha_{1E}$  appears to correlate in part to that which has been described as R-type currents ("R" for resistant to known pharmacological tools) (Soong et al., 1993; Schneider et al., 1994; Williams et al., 1994). More recently, a toxin has been isolated from the tarantula *Hysterocrates gigas*, named SNX482, described as a selective blocker for  $\alpha_{1E}$  containing  $Ca^{2+}$  channels (Newcomb et al., 1998). This toxin blocks the channel by interacting with the domains III and IV of  $\alpha_{1E}$  but seems to block L-type channels as well (Bourinet et al., 2001).

# 5 CARDIAC Ca<sup>2+</sup> CHANNELS AND PHYSIO-PATHOLOGICAL CONDITIONS

Although several of human genetic disease associated with mutations in ionic channels (channelopathies) exist in the heart and has been linked to the long Q-T syndrome (mutations in Na<sup>+</sup> or K<sup>+</sup> channels, see for review (Marban, 2002)) and inherited channelopathies exist for Ca<sup>2+</sup> channels (*e.g.* hemiplegic migraine, point mutation in  $\alpha_{1A}$ , hypokalaemic periodic paralysis mutation in  $\alpha_{1S}$ , see for review (Lorenzon and Beam, 2000)), so far no cardiac disease is known to be linked to mutations in the cardiac  $\alpha_1$  or  $\beta$  subunits. However, change of I<sub>Ca</sub> expression and function has been described in disease and I<sub>Ca</sub> can be linked to cardiac pathology. For example, it is well kwown that arrhythmogenic EADs are potentiated by I<sub>CaL</sub> agonist (Bay K8644) whereas I<sub>CaL</sub> blocker (Nitrendipine) abolished EADs in ferret ventricular muscle (Marban et al., 1986). We will now describe several cardiac diseases where I<sub>Ca</sub> has been implicated.

#### 5.1 Diseases with auto-immune components

#### 5.1.1 Congenital heart block

Congenital heart block in a structurally normal heart is strongly associated with autoantibodies reactive with ribonucleoproteins SSA/Ro and SSB/La. This is presumed to be due to the trans-placental passage of auto-antibodies from the mother into the fetal circulation, which can be asymptomatic or exhibit rheumatologic disease such as systemic lupus erythematosus (for review see Boutjdir 2000).

An early study demonstrated that these antibodies inhibit  $I_{CaL}$  in rabbit ventricular myocytes (Garcia et al., 1994). Subsequently, Boutjdir and colleagues provided compelling evidence that these autoimmune antibodies can block  $I_{CaL}$ . They found that positive IgG fraction (whole IgG containing anti-SSA/Ro and SSB/La antibodies purified from sera) and/or affinity purified anti-52 kD SSA/Ro antibodies, from mothers with CHB children, were able to decrease  $I_{CaL}$  amplitude in human fetal myocytes (Boutjdir et al., 1997). At the single channel level, the open probability of L-type Ca<sup>2+</sup> channels was decreased (Boutjdir et al., 1997). These auto-antibodies produced a similar decrease in  $I_{CaL}$  amplitude in rat ventricular cardiomyocytes (Boutjdir et al., 1998), canine Purkinje myocytes (Qu et al., 2001), and rabbit sinoatrial cells (Hu et al., 2004). Inhibition of  $I_{Ca}$  was confirmed in expression system (*Xenopus* oocytes expressing  $\alpha_{IC}$  or  $\alpha_{1H}$ , (Xiao et al., 2001a)). These data provide compelling

evidence that auto-antibodies can block  $I_{Ca}$ , and may thus play a role in the pathogenesis of bradycardia and atrioventricular block in congenital heart block.

An intriguing question is how intracellular antibodies can recognize an extracellular  $Ca^{2+}$  channel epitope. Numerous homologous epitopes were found in 3D representations of human  $\alpha_{1C}$ , especially with the S5-S6 linker of domain VI (Qu et al., 2001). Furthermore,  $\alpha_{1C}$  binding by positive maternal IgG has been confirmed by immunoprecipitation and immunohistochemical study in human fetal cardiomyocytes (Qu et al., 2001). However, the exact site(s) of interaction is not yet determined.

Although an acute effect of these antibodies has been shown on  $I_{CaL}$ , a chronic effect is also suspected.  $I_{CaL}$  density is decreased in ventricular cells from mice and rabbit pups born from mothers immunized with recombinant 52kD-SSA/Ro protein suggesting that prolonged exposure of pups' heart to maternal antibodies during pregnancy could lead to reduced channel expression (Qu et al., 2001; Xiao et al., 2001b). This hypothesis is supported by a lower amount of L-type Ca<sup>2+</sup> channel protein assessed by ELISA and Western blot in pup hearts from immunized mothers. A down-regulation by internalization of L-type Ca<sup>2+</sup> channel induced by anti-52kD-SSA/Ro antibodies has also been suggested (Xiao et al., 2001b).

There is striking evidence that auto-antibodies can directly affect  $I_{Ca}$ , but an alternative mechanism has also been described. The L-type  $Ca^{2+}$  channel may not be the only target for maternal anti-SSA/Ro and/or SSB/La antibodies, because a specific cross-reaction between the recombinant 52kDa SSA/Ro protein and the second extracellular loop of serotoninergic 5-HT<sub>4</sub> receptor expressed in human atrial cells has been identified (sequence G21V) (Eftekhari et al., 2000). In contrast to the results describe above, anti-G21V affinity-purified autoantibodies from lupus patients or from rabbits immunized with the G21V peptide do not inhibit basal  $I_{CaL}$  in adult human atrial myocytes. However, these antibodies antagonize serotonin-induced activation of  $I_{CaL}$  (Eftekhari et al., 2000; Salle et al., 2001); the authors hypothesized that during the early phase of fetal development the serotoninic pathway is more prominent than the  $\beta$ -adrenergic pathway so that 5-HT<sub>4</sub> receptor block leads to a reduction of  $I_{Ca}$  which could in turn induce bradycardia and atrioventricular block (Eftekhari et al., 2001).

Taken together, these results suggest that a mother's antibodies can either directly interact with  $Ca^{2+}$  channels, or can alter channel function indirectly via receptor modulation, and thus contribute to sinus bradycardia and atrioventricular block of their children. But an intriguing question remains: why do autoantibodies have deleterious effects on children while most mothers do not exhibit cardiac symptoms?

A number of antibodies have been identified in sera from patients with myocarditis/dilated cardiomyopathy (for reviews see Malkiel et al., 1996; Caforio et al., 2002) and myocarditis/Chagas' cardiomyopathy (reviewed by Kierszenbaum, 2003).

Myocarditis is an inflammatory disease of the myocardium, and idiopathic, autoimmune, and infectious forms of inflammatory cardiomyopathy are recognized. Inflammatory myocardial disease is involved in the pathogenesis of dilated cardiomyopathy and Chagas' cardiomyopathy (Richardson et al., 1996). Thus, myocarditis and dilated cardiomyopathy DCM and Chagas' cardiomyopathy are thought to represent the acute and chronic phases of an inflammatory process of the myocardium that can be viral, post-infectious immune or organ-specific autoimmune. The infectious agent in Chagas' disease is the protozoan *Trypanosoma cruzi*. According to the Word Health Organization, 16–18 million people are infected by the parasite in South America, and another 90 million are at risk of becoming infected. Chagas' disease is a complex, multifaceted and widespread disease in which, during the chronic phase, a slowly evolving cardiomyopathy affecting about a third of infected people can be observed that leads to severe cardiac dilatation, congestive heart failure, arrhythmias, and death (Higuchi et al., 2003).

The sera of patients with myocarditis and dilated cardiomyopathy shows high frequencies of antibodies directed against the adenine nucleotide translocator (ANT) of the inner mitochondrial membrane (Kuhl et al., 1991), and specific binding by cross-reaction to the L type  $Ca^{2+}$  channel has been demonstrated (Schultheiss et al., 1988). Furthermore, it has been shown that anti-ANT antibody increases  $I_{Ca}$  amplitude in frog and rat myocytes (Morad et al., 1988; Schultheiss et al., 1988), which will contribute to increased intracellular  $Ca^{2+}$  and thus could impair cardiac function (see section 5.2.2).

The sera of patients with idiopathic dilated cardiomyopathy and Chagas' disease also contains autoantibodies directed against the  $\beta_1$ -,  $\beta_2$ .adrenoceptor (Borda et al., 1984; Wallukat and Wollenberger, 1987; Limas et al., 1989; Magnusson et al., 1994; Wallukat et al., 1995; Elies et al., 1996) and/or the muscarinic M<sub>2</sub>-receptor (Fu et al., 1993; Goin et al., 1994; Elies et al., 1996). Immunoglobulins purified from sera of mice experimentally infected with *T. cruzi* during the acute or chronic phase of chagasic cardiopathy can respectively enhance or reduce L-type Ca<sup>2+</sup> current in myocytes isolated from guinea-pig control hearts (Mijares et al., 1996a). These effects were inhibited by propranolol and atropine respectively, suggesting a preferential activation of  $\beta$ -adrenoceptors by antibodies during the acute phase and of M<sub>2</sub>-receptors during the chronic one (Mijares et al., 1996a). These experiments functionally demonstrated the agonist-like activity of Chagasic antibodies on  $\beta$ -adrenorgic and muscarinic receptors.

The second extracellular loop of cardiovascular G-protein coupled receptor (GPCR) appears to be one of the main targets of these antibodies (see for review see Hoebeke 2001). Hence an alternative approach to investigate the functional role of these anti-GPCR antibodies is to investigate the effect of antibodies raised against a peptide corresponding to this loop (Magnusson et al., 1994; Fu, 1995; Mijares et al., 1996b; Matsui et al., 1997; Elies et al., 1998; Lebesgue et al., 1998).

Monoclonal anti- $\beta_2$  adrenoceptor antibodies raised against the second extracellular loop increase I<sub>CaL</sub> in guinea-pig ventricular cardiomyocytes (Mijares et al., 1996b; Lebesgue et al., 1998). Monoclonal or polyclonal antibodies generated against the second extracellular loop of the human muscarinic receptor mimic those found in Chagas' disease, decreasing the amplitude of I<sub>CaL</sub> that has been prestimulated by isoproterenol (Zhao et al., 1996; Nascimento et al., 2001; Hernandez et al., 2003). In addition to  $\beta$ -adrenergic and muscarinic stimulation acting via the cAMP pathway (Wallukat et al., 1995; Fu, 1995; Borda and Sterin-Borda, 1996; Wallukat et al., 1999b; Chiale et al., 2001), it has been shown that antibodies directed against the second extracellular loop of the M<sub>2</sub>-muscarinic receptor are also able to activate the NO/cGMP pathway (Sterin-Borda et al., 1997; Nascimento et al., 2001; Sterin-Borda et al., 2003) (see also section 3.3.2).

Importantly, in contrast to the classical agonists isoproterenol and carbachol, these antibodies fail to produce desensitization of the reaction cascade within 6 hours in cultured neonatal myocytes, inducing chronic stimulation of the GPCR (Magnusson et al., 1994; Wallukat et al., 1995; Wallukat et al., 1999a; Wallukat et al., 1999b). Such chronic stimulation of  $\beta$ -adrenergic and/or muscarinic cascades may have profound consequence on Ca<sup>2+</sup> signaling, which may impair cardiac function (see section 5.2.2) and be pro-arrhythmic. For example, it has been shown that overexpression of  $\beta_2$ -adrenergic receptors in mice accelerates the progression of myocardial fibrosis and heart failure (Liggett et al., 2000). More recently, Jahns et al., (2004) have demonstrated that agonistic anti- $\beta_1$ -adrenoceptor antibodies alone are sufficient to engender myocardial dysfunction characteristic of heart failure.

Thus data are emerging that suggest physiopathologic roles for antibodies, mediated by direct and/or indirect modulation of  $I_{Ca}$ . However it is unclear whether autoreactivity to cardiac antigens initiates heart injury, amplifies ongoing disease, or is merely a sign of prior myocyte damage.

#### 5.2 Cardiac remodeling

The diseased heart undergoes biochemical, electrophysiological and structural changes globally defined as remodeling (Swynghedauw, 1999). In this section we will review three types of remodeling in which modification of  $I_{Ca}$  has been implicated.

#### 5.2.1 Cardiac memory

Memory is a form of remodeling common to many organ and cell systems. In the heart, cardiac memory is characterized by an altered T-wave on the electrocardiogram during sinus rhythm, induced by a preceding period of abnormal ventricular activation (Chatterjee et al., 1969). The direction of the change in the T-wave is in that of the vector of the inciting and abnormally activated QRS complex, hence the term "cardiac memory" was suggested: the Twave "remembered" the ectopic QRS (Rosenbaum et al., 1982). Importantly, this occurs in hearts that have no demonstrable hemodynamic or structural abnormalities or ischemia. A variety of pathologies that alter ventricular activation pattern induce cardiac memory in humans including intermittent left bundle branch block, post-tachycardia syndrome, ventricular pre-excitation and extrasystoles (reviewed in (Patberg and Rosen, 2004)). Rosen's laboratory developed a canine model of ventricular pacing to induce cardiac memory and showed that cardiac memory was associated with altered action potential characteristics (smaller notch and longer action potential). They first showed that the transient outward current density was reduced and that this was associated a decrease at the mRNA level, possibly accounting for the change observed in the action potential (Yu et al., 1999). Further investigation showed that cardiac memory was attenuated by I<sub>CaL</sub> blockade (nifidipine), suggesting that I<sub>CaL</sub> is important in the development of cardiac memory (Plotnikov et al., 2003). Indeed, a short-term memory protocol (2 hours ventricular pacing) showed a significant decrease in nuclear CREB, that did not occur in control dogs or dogs treated with nifidipine (Patberg et al., 2003). These changes implicate effects of I<sub>CaL</sub> at the transcription level (see section 2.3). At the electrophysiological level, no difference was found in the peak density of I<sub>CaL</sub> in cardiac memory myocytes compared to control, although activation was more positive and  $I_{CaL}$  inactivation slower (Plotnikov et al., 2003), which may contribute to the longer action potential. Although the process leading to cardiac memory is far from being completely understood, there is a clear implication of I<sub>CaL</sub> at the electrophysiological and transcriptional levels.

#### 5.2.2 Heart failure

Heart failure is the leading heart disease in the Occidental world. It is characterized by a decrease in ejection fraction, and can result in death in two main ways: a progressive decline in contractile function or sudden death due to arrhythmias. At the cellular level, ventricular myocytes from failing hearts are hypertrophied and show an increase in action potential duration, a decrease in contractility and a blunted  $\beta$ -adrenergic responsiveness (for review see (Tomaselli and Marban, 1999; Towbin and Bowles, 2002)). The mechanisms behind these observations are still unclear although the decrease in contractility might be linked to a decrease in the amplitude and the rate of decline of the systolic Ca<sup>2+</sup> transient (Houser and Margulies, 2003).

Animal models provide useful tools to gain insight into the mechanisms of heart failure, which can be induced by a variety of means (*e.g.* rapid pacing, aortic constriction, infusion of angiotensin II or catecholamines). Over the past 15 years, investigations of  $I_{CaL}$  in animal models of hypertrophy or heart failure have led to a profusion of data with either decreased, increased, or unchanged  $I_{CaL}$ . This variety of results might reflect the different degree of heart failure, species, and experimental conditions. However, in the majority of studies  $I_{CaL}$  amplitude, expressed as density, is unchanged (see (Tomaselli and Marban, 1999) for extensive review); this has been shown in ventricular myocytes from failing rat (Gomez et al., 2001), dog (O'Rourke et al., 1999), rabbit (Pogwizd et al., 1999) and guinea pig (Ahmmed et al., 2000) hearts: although cell surface area approximately doubles,  $I_{Ca}$  density is unchanged

Less data are available from human ventricular myocytes because of the obvious problem of obtaining "healthy" controls. However, several groups have studied myocytes from donor hearts unsuitable for transplantation, and in human studies peak I<sub>Ca</sub> density is also unchanged in myocytes from failing heart (Beuckelmann et al., 1991; Schroder et al., 1998; Chen et al., 2002; Piacentino, III et al., 2003). At the single channel level, failing human myocytes show increased open probability and availability of L-type Ca<sup>2+</sup> channels compared to non-failing myocytes (Schroder et al., 1998). This resembles the effect of  $\beta$ -adrenergic stimulation on the L-type Ca<sup>2+</sup> channel (see section 3.3.1), and it has been proposed that channel phosphorylation is increased in the failing heart because of impaired dephosphorylation (Schroder et al., 1998). This idea has recently been supported by the observation that although global I<sub>Ca</sub> amplitude is the same in failing and non-failing heart, the activation curve of I<sub>Ca</sub> is shifted to the left, suggesting  $\beta$ -adrenergic stimulation (Chen et al., 2002), and I<sub>Ca</sub> from failing heart shows less response to isoproterenol ( $\beta$ -adrenergic agonist). Conversely, PP2A decreases I<sub>Ca</sub> density in the failing heart but not in normal heart, suggesting an increase in basal phosphorylation. Interestingly, these authors also showed that support using a left ventricular assist device restored Ca<sup>2+</sup> channel density and regulation (Chen et al., 2002), consistent with a recent clinical study showing that such support also induces clinical recovery of heart failure by modification of sarcoplasmic reticulum Ca<sup>2+</sup> handling (Terracciano et al., 2004). Ca<sup>2+</sup> channel density, assessed using Bay K8644, is decreased in myocytes from failing heart (Chen et al., 2002); this agrees with a recent study in an animal model, which showed a reduction in Ca<sup>2+</sup> channel density (assessed by recording charge movement, Balijepalli et al., 2003) in failure, although the density of I<sub>Ca</sub> was unaltered, suggesting upregulation of existing channel function. Studies at the protein expression level show conflicting results, one showing unchanged mRNA levels for  $\alpha_{1C}$  and  $\alpha_{2}$ - $\delta$  in heart failure, but with  $\beta$  subunit expression reduced by ~75% (Hullin et al., 1999), and another showing that  $\alpha_{1C}$  and  $\beta$  subunit expression are unchanged (Schroder et al., 1998).

These alterations in the density and regulation of the L type  $Ca^{2+}$  channel might contribute to the abnormal contractility and blunted adrenergic responsiveness of the failing heart. The mechanisms underlying the altered density and regulation of the L type  $Ca^{2+}$ channel are not yet known. Intriguingly, 4 isoforms of  $\alpha_{1C}$  of the L-type  $Ca^{2+}$  channel are expressed in the human heart, and isoform switching occurs in failing human myocytes (Yang et al., 2000). This may also contribute to the observed change in L type  $Ca^{2+}$  channel density during heart failure. An alternative possibility to explain the reduction in L type  $Ca^{2+}$  channel density is a reduction in the density of the t-tubules, where the L-type  $Ca^{2+}$  channel and current are concentrated (see section 2.1.1). It has been shown in the canine pacing model of heart failure that t-tubule density decreases, leading to a decrease in peak  $I_{Ca}$  density (Balijepalli et al., 2003). A similar decrease in t-tubule density has been observed in failing rabbit heart (Quinn et al., 2003). To date, only preliminary reports of t-tubule structure in failing human myocytes have appeared, and show conflicting results, one study showing a decrease in density (Wong et al., 2002) and the other no change (Ohler et al., 2002). Clearly, this issue requires a full investigation.

In summary, in ventricular myocytes from failing heart, L type  $Ca^{2+}$  channel density appears to be reduced, although other changes, for example increased basal phosphorylation and isoform switching, may help to maintain  $I_{Ca}$  density.

Given the absence of changes in  $I_{CaL}$  amplitude but a decrease in sarcoplasmic reticulum  $Ca^{2+}$  release, it is tempting to speculate that a decrease in EC coupling gain (see section 2.2.1) can explain  $Ca^{2+}$  dysfunction in heart failure. However, no changes in EC coupling gain have been shown in heart failure in rabbit (Pogwizd et al., 2001), dog (Hobai and O'Rourke, 2001) or human (Piacentino, III et al., 2003). In these cases, a decrease in sarcoplasmic reticulum  $Ca^{2+}$  content appears to be responsible for defective EC coupling. In contrast, in rat a reduction

in the ability of  $I_{CaL}$  to trigger sarcoplasmic reticulum  $Ca^{2+}$  release was found (reduced EC coupling gain), without altered sarcoplasmic reticulum  $Ca^{2+}$  load (Gomez et al., 1997; Gomez et al., 2001). Three hypothesis has been proposed to explain this altered EC coupling gain during heart failure: (i) a change in the co-localization of L type  $Ca^{2+}$  channels and RyRs; (ii) an increase in the gap between the t-tubule and sarcoplasmic reticulum membranes; (iii) t-tubule remodeling (Gomez et al., 2001).

Given the reduction in sarcoplasmic reticulum  $Ca^{2+}$  release (either by reduced sarcoplasmic reticulum  $Ca^{2+}$  load or change in EC coupling), it might be expected that  $I_{CaL}$ inactivation would be slowed because of less  $Ca^{2+}$ -dependent inactivation. This has been observed in some (*e.g.* (Ahmmed et al., 2000; Gomez et al., 2001)), but not all, studies, although contaminating currents (*e.g.* Na/Ca exchanger) might explain this discrepancy. Such slowing of  $I_{CaL}$  inactivation will lead to greater  $Ca^{2+}$  entry during the action potential (see section 2.1.1) and will promote the occurrence of arrhythmias (EADs).

It is also noteworthy that frequency-dependent facilitation of  $I_{Ca}$  is blunted in myocytes from failing human heart (Piot et al., 1996; Sipido et al., 1998b; Barrere-Lemaire et al., 2000). Thus increasing stimulation rate induces a decrease of  $I_{CaL}$  which is associated with a decrease in action potential duration (Sipido et al., 1998b; Li et al., 1999). This might help explain the negative staircase (decrease of contraction with increase in rate stimulation) and the beneficial effects of slowing heart rate observed in heart failure.

Although  $I_{CaL}$  amplitude is unchanged in ventricular myocytes from failing hearts (above), a decrease has been described in atrial myocytes from two animal models (Li et al., 2000; Boixel et al., 2001), although heart failure induces a smaller decrease of  $I_{Ca}$  amplitude (~30%) than atrial fibrillation (~65%) and does not change action potential duration (Li et al., 2000) (see also section 5.2.3). However, in human failing atrial myocytes,  $I_{CaL}$  density appears unchanged (Mewes and Ravens, 1994).

No change in  $I_{CaL}$  and  $I_{CaT}$  density has been observed in sinoatrial node myocytes from a rabbit model of heart failure (Verkerk et al., 2003). However,  $I_{CaT}$  density has been reported to be increased in ventricular cells isolated from animal models of heart failure (e.g. cat (Nuss and Houser, 1993); hamster (Sen and Smith, 1994), rat (Martinez et al., 1999; Izumi et al., 2003; Ferron et al., 2003)) This increase is reminiscent of the electrophysiological phenotype observed in fetal cells and may therefore represent a reversion to the fetal phenotype: the balance between  $I_{CaL}$  and  $I_{CaT}$  density changes during development, with a progressive decrease of  $I_{CaT}$  density which is not detectable in most adult ventricular myocytes (*e.g.* ventricle (Cohen and Lederer, 1988); rabbit (Wetzel et al., 1993)). Interestingly,  $I_{CaT}$  can be

reexpressed when ventricular adult rat myocytes dedifferentiate during cell culture (Fares et al., 1996).

In contrast with animal models, no significant  $I_{CaT}$  has been detected in atrial or ventricular myocytes isolated from human hearts showing hypertrophy or dilated cardiomyopathy; *e.g.* (Beuckelmann et al., 1991; Le Grand et al., 1991; Chen et al., 2002; Piacentino, III et al., 2003). Therefore, a role for  $I_{CaT}$  in human heart failure and associated arrhythmogenesis appears unlikely.

The development of heart failure is poorly understood. It arises after cardiac hypertrophy, which is an adaptive response that helps the heart deal with an increase in workload. If this increase in workload has a pathophysiological origin (hemodynamic load, neurohormonal stimulus *e.g.* renin-angiotensin II, endothelin1, catecholamines), the hypertrophic response may progress from compensated (heart meets the need of the body) to decompensated hypertrophy, which leads to cardiac dysfunction and overt failure.

Work in transgenic mice suggests that  $Ca^{2+}$  signaling is a likely contributor to this process. Transgenic mice overexpressing the L type  $Ca^{2+}$  channel develop cardiac hypertrophy and cardiomyopathy and die of congestive heart failure (Muth et al., 2001). In contrast, the L type  $Ca^{2+}$  channel inhibitor diltiazem has been reported to blunt cardiac hypertrophy in a mouse model of familial hypertrophic cardiomyopathy (Semsarian et al., 2002).  $Ca^{2+}$  signaling can occur at different levels, and transgenic mice overexpressing calmodulin also show severe cardiac hypertrophy (Gruver et al., 1993). Similarly, transgenic mice overexpressing nuclear CaMKII- $\delta_B$  show cardiac hypertrophy and dilatation (Zhang et al., 2002a), and cardiac specific activation of calcineurin in transgenic mice induces a robust hypertrophic response that progresses to dilated heart failure (Molkentin et al., 1998). Conversely, calcineurin null mice fail to undergo cardiac hypertrophy in response to pressure overload, isoproterenol infusion or angiotensin II infusion (Bueno et al., 2002). Interestingly, NFAT3 null mice showed similar resistance to hypertrophy, establishing the critical downstream role of NFAT3 in mediating calcineurin-regulated hypertrophy (Wilkins et al., 2002) (see also section 2.3 and Figure 4).

In summary, it seems unlikely that  $I_{CaL}$  contributes to the defects in EC coupling observed in heart failure in large mammals (rabbit, dog, human). However, it may contribute to the development of arrhythmias (EADs) and to the negative staircase observed in heart failure. The role of  $I_{CaL}$  in the processes leading to heart failure via ET coupling requires further investigation.

#### 5.2.3 Atrial fibrillation

Atrial fibrillation is the most frequently encountered arrhythmia in clinical practice (Kannel et al., 1982). Chronic atrial fibrillation is characterized by a disorganized high rate (atrial cells fire at 400-600 times per minutes) instead of the normal heart rate (~60 beats per minutes in human). An important advance in the understanding of atrial fibrillation was made with the recognition that atrial fibrillation, once initiated, alters electrophysiological properties in a manner that favours the ease of inducing and maintaining the arrhythmia: "atrial fibrillation begets atrial fibrillation" (Wijffels et al., 1995). This process is called "electrical remodeling". Sustained atrial fibrillation can cause heart failure after several weeks or months (Fenelon et al., 1996). Conversely, heart failure promotes atrial fibrillation and produces atrial ionic remodeling (Li et al., 2000).

At the cellular level, atrial fibrillation is characterized by a depolarized resting potential, a decrease of the action potential plateau, decreased action potential duration and loss of effective atrial contraction (see for review (Nattel, 2002)). Involvement of  $Ca^{2+}$  channels was suggested by an early study showing that the  $Ca^{2+}$  channel antagonist verapamil depresses the AP plateau of normal atria but not in atria from patients with atrial fibrillation (Hordof et al., 1976).

Animal models for atrial fibrillation have confirmed this suggestion. In a rapid pacing dog model of atrial fibrillation,  $I_{CaL}$  density was reduced without concomitant change in its kinetics or voltage dependence, compared with sham-operated dogs (Yue et al., 1997). In contrast,  $I_{CaT}$  density was not altered in this model. Rapid pacing also decreased action potential duration and action potential adaptation to rate. In contrast to to sham operated animals, nifedipine did not alter the action potential duration in the paced dog model suggesting that  $I_{CaL}$  depression was responsible for the action potential shortening in these dogs (Yue et al., 1997). Recognition that reduction of  $I_{CaL}$  is an important factor in atrial fibrillation was strengthened by the observation that  $I_{CaL}$  current density is decreased by ~ 65% in myocytes from patients with chronic atrial fibrillation (Van Wagoner et al., 1999). In contrast to heart failure,  $\beta$ -adrenergic responsiveness of atrial myocytes from patients with chronic atrial fibrillation (~4 fold) compared with patients with normal sinus rhythm (~3 fold) (Van Wagoner et al., 1999).

Although a decrease in  $I_{CaL}$  amplitude is assumed to be the most important factor in altering the duration of the action potential and its loss of adaptation to rate in atrial fibrillation, a decrease in the amplitude of the transient outward potassium current ( $I_{to}$ ) has also been observed (Van Wagoner et al., 1999). The decrease in density of both currents has been

linked to transcriptional down regulation of  $\alpha$  and  $\beta$  subunits of these channels (Bosch et al., 2003), although the molecular events leading to such ionic remodeling remain incompletely understood.

Thus although it is clear that the properties of  $I_{CaL}$  change during atrial fibrillation, it remains to determine whether  $I_{CaL}$  change is a cause or consequence of this pathology.

### 6 CONCLUSIONS AND FUTURES DIRECTIONS

The story of  $Ca^{2+}$  currents in cardiac myocytes is not over yet. The translation research, where the use of transgenic mice addresses integrative normal and pathological questions provides a helpful tool for a better understanding of  $Ca^{2+}$  current's function but also leads to conflicting results. The observation that knock out mice for Ca<sub>v</sub>1.3, Ca<sub>v</sub>2.3, Ca<sub>v</sub>3.1, Ca<sub>v</sub>3.2 are healthy, however Ca<sub>V</sub>1.2 -/- transgenic mice die at the embryonic stage clearly emphasizes the crucial role of I<sub>CaL</sub> in the heart. This is consistent with the observation that no cardiac diseases are linked to mutations of  $Ca^{2+}$  channels. It appears clear that we have to think in terms of microdomains for studying Ca<sup>2+</sup> channels. This concept of microdomain is clear in neurons, and requires further investigation in cardiac cells. We described that ET coupling due to Ca<sup>2+</sup> microdomain is important in neurons, however in cardiac myocytes it is still unclear how cells distinguish between regular global dynamic Ca<sup>2+</sup> signals involved in EC coupling and Ca<sup>2+</sup> involved in ET coupling. Furthermore, it is well established that I<sub>Ca</sub> participates in the excitation-secretion coupling in neuronal cells, e.g. the release mechanism of neurotransmitter vesicles at the synapse, due to  $Ca^{2+}$  microdomains (See for review Neher, 1998). In cardiac myocytes excitation-secretion coupling concept has not yet emerged and to date only one report shows the importance of I<sub>CaT</sub> in the secretion of atrial natriuretic factor by atrial cells (Leuranguer et al., 2000). The development of new microscopic techniques such as FRET, should provide new information regarding inter- and intra-molecular arrangements and compartmentalization. Hence, it appears that the term " $Ca^{2+}$  currents story" in the heart is not so appropriate and perhaps should be renamed "the  $Ca^{2+}$  currents saga".

# ACKNOWLEDGEMENTS

We would like to thank Dr Fay Heblich, Professor Clive Orchard and Professor Annette Dolphin for her helpful comments on the manuscript. Fabien Brette and Jérôme Leroy are also grateful to the Wellcome Trust for his financial support.

# REFERENCES

Abernethy, D. R. and Soldatov, N. M. 2002. Structure-Functional Diversity of Human L-Type Ca2+ Channel: Perspectives for New Pharmacological Targets. J Pharmacol Exp Ther 300, 724-728.

Abi-Gerges, N., Fischmeister, R., and Mery, P. F. 2001. G protein-mediated inhibitory effect of a nitric oxide donor on the L-type Ca2+ current in rat ventricular myocytes. J Physiol (Lond) 531, 117-130.

Abi-Gerges, N., Szabo, G., Otero, A. S., Fischmeister, R., and Mery, P. F. 2002. NO donors potentiate the beta-adrenergic stimulation of ICa,L and the muscarinic activation of IK,ACh in rat cardiac myocytes. J Physiol (Lond) 540, 411-424.

Ahmmed, G. U., Dong, P. H., Song, G., Ball, N. A., Xu, Y., Walsh, R. A., and Chiamvimonvat, N. 2000. Changes in Ca(2+) cycling proteins underlie cardiac action potential prolongation in a pressure-overloaded guinea pig model with cardiac hypertrophy and failure. Circ Res 86, 558-570.

Almers, W. and McCleskey, E. W. 1984. Non-selective conductance in calcium channels of frog muscle: calcium selectivity in a single-file pore. J Physiol 353, 585-608.

Alseikhan, B. A., DeMaria, C. D., Colecraft, H. M., and Yue, D. T. 2002. Engineered calmodulins reveal the unexpected eminence of Ca2+ channel inactivation in controlling heart excitation. Proc Natl Acad Sci U S A 99, 17185-17190.

Alvarez, J. L., Rubio, L. S., and Vassort, G. 1996. Facilitation of T-type calcium current in bullfrog atrial cells: voltage-dependent relief of a G protein inhibitory tone. J Physiol 491 (Pt 2), 321-334.

Anderson, M. E., Braun, A. P., Schulman, H., and Premack, B. A. 1994. Multifunctional Ca2+/calmodulin-dependent protein kinase mediates Ca(2+)-induced enhancement of the L-type Ca2+ current in rabbit ventricular myocytes. Circ Res 75, 854-861.

Angelotti, T. and Hofmann, F. 1996. Tissue-specific expression of splice variants of the mouse voltage-gated calcium channel alpha2/delta subunit. FEBS Lett 397, 331-337.

Aprigliano, O., Rybin, V. O., Pak, E., Robinson, R. B., and Steinberg, S. F. 1997. beta 1-and beta 2adrenergic receptors exhibit differing susceptibility to muscarinic accentuated antagonism. Am J Physiol 272, H2726-H2735.

Argibay, J. A., Fischmeister, R., and Hartzell, H. C. 1988. Inactivation, reactivation and pacing dependence of calcium current in frog cardiocytes: correlation with current density. J Physiol (Lond) 401, 201-226.

Arikkath, J. and Campbell, K. P. 2003. Auxiliary subunits: essential components of the voltagegated calcium channel complex. Curr Opin Neurobiol 13, 298-307.

Artalejo, C. R., Ariano, M. A., Perlman, R. L., and Fox, A. P. 1990. Activation of facilitation calcium channels in chromaffin cells by D1 dopamine receptors through a cAMP/protein kinase A-dependent mechanism. Nature 348, 239-242.

Artalejo, C. R., Rossie, S., Perlman, R. L., and Fox, A. P. 1992. Voltage-dependent phosphorylation may recruit Ca2+ current facilitation in chromaffin cells. Nature 358, 63-66.

Ashley, E. A., Sears, C. E., Bryant, S. M., Watkins, H. C., and Casadei, B. 2002. Cardiac nitric oxide synthase 1 regulates basal and beta-adrenergic contractility in murine ventricular myocytes. Circulation 105, 3011-3016.

Balijepalli, R. C., Lokuta, A. J., Maertz, N. A., Buck, J. M., Haworth, R. A., Valdivia, H. H., and Kamp, T. J. 2003. Depletion of T-tubules and specific subcellular changes in sarcolemmal proteins in tachycardia-induced heart failure. Cardiovasc Res 59, 67-77.

Bangalore, R., Mehrke, G., Gingrich, K., Hofmann, F., and Kass, R. S. 1996. Influence of L-type Ca channel alpha 2/delta-subunit on ionic and gating current in transiently transfected HEK 293 cells. Am J Physiol 270, H1521-H1528.

Barouch, L. A., Harrison, R. W., Skaf, M. W., Rosas, G. O., Cappola, T. P., Kobeissi, Z. A., Hobai, I. A., Lemmon, C. A., Burnett, A. L., O'Rourke, B., Rodriguez, E. R., Huang, P. L., Lima, J. A., Berkowitz, D. E., and Hare, J. M. 2002. Nitric oxide regulates the heart by spatial confinement of nitric oxide synthase isoforms. Nature 416, 337-339.

Barrere-Lemaire, S., Piot, C., Leclercq, F., Nargeot, J., and Richard, S. 2000. Facilitation of L-type calcium currents by diastolic depolarization in cardiac cells: impairment in heart failure. Cardiovasc Res 47, 336-349.

Bates, S. E. and Gurney, A. M. 1999. Use-dependent facilitation and depression of L-type Ca2+ current in guinea-pig ventricular myocytes: modulation by Ca2+ and isoprenaline. Cardiovasc Res 44, 381-389.

Bean, B. P. 1985. Two kinds of calcium channels in canine atrial cells. Differences in kinetics, selectivity, and pharmacology. J Gen Physiol 86, 1-30.

Bean, B. P. 1989. Classes of calcium channels in vertebrate cells. Annu Rev Physiol 51, 367-384.

Bean, B. P. and Rios, E. 1989. Nonlinear charge movement in mammalian cardiac ventricular cells. Components from Na and Ca channel gating. J Gen Physiol 94, 65-93.

Beeler, G. W. Jr. and Reuter, H. 1970. The relation between membrane potential, membrane currents and activation of contraction in ventricular myocardial fibres. J Physiol 207, 211-229.

Beguin, P., Nagashima, K., Gonoi, T., Shibasaki, T., Takahashi, K., Kashima, Y., Ozaki, N., Geering, K., Iwanaga, T., and Seino, S. 2001. Regulation of Ca2+ channel expression at the cell surface by the small G-protein kir/Gem. Nature 411, 701-706.

Bell, D. C., Butcher, A. J., Berrow, N. S., Page, K. M., Brust, P. F., Nesterova, A., Stauderman, K. A., Seabrook, G. R., Nurnberg, B., and Dolphin, A. C. 2001. Biophysical properties, pharmacology, and modulation of human, neuronal L-type (alpha(1D), Ca(V)1.3) voltage-dependent calcium currents. J Neurophysiol 85, 816-827.

Berjukow, S., Marksteiner, R., Sokolov, S., Weiss, R. G., Margreiter, E., and Hering, S. 2001. Amino acids in segment IVS6 and beta-subunit interaction support distinct conformational changes during Ca(v)2.1 inactivation. J Biol Chem 276, 17076-17082.

Bernatchez, G., Talwar, D., and Parent, L. 1998. Mutations in the EF-hand motif impair the inactivation of barium currents of the cardiac alpha1C channel. Biophys J 75, 1727-1739.

Berrou, L., Bernatchez, G., and Parent, L. 2001. Molecular determinants of inactivation within the I-II linker of alpha1E (CaV2.3) calcium channels. Biophys J 80, 215-228.

Berrou, L., Klein, H., Bernatchez, G., and Parent, L. 2002. A specific tryptophan in the I-II linker is a key determinant of beta-subunit binding and modulation in Ca(V)2.3 calcium channels. Biophys J 83, 1429-1442.

Bers, D. M. (2001) *Excitation-contraction coupling and cardiac contractile force*. 2nd ed. Dordrecht, Netherlands: Kluwer Academic.

Beuckelmann, D. J., Nabauer, M., and Erdmann, E. 1991. Characteristics of calcium-current in isolated human ventricular myocytes from patients with terminal heart failure. J Mol Cell Cardiol 23, 929-937.

Beuckelmann, D. J. and Wier, W. G. 1988. Mechanism of release of calcium from sarcoplasmic reticulum of guinea-pig cardiac cells. J Physiol 405, 233-255.

Bezanilla, F. 2000. The voltage sensor in voltage-dependent ion channels. Physiol Rev 80, 555-592.

Bichet, D., Cornet, V., Geib, S., Carlier, E., Volsen, S., Hoshi, T., Mori, Y., and De Waard, M. 2000. The I-II loop of the Ca2+ channel alpha1 subunit contains an endoplasmic reticulum retention signal antagonized by the beta subunit. Neuron 25, 177-190.

Birnbaumer, L., Qin, N., Olcese, R., Tareilus, E., Platano, D., Costantin, J., and Stefani, E. 1998. Structures and functions of calcium channel beta subunits. J Bioenerg Biomembr 30, 357-375.

Black, J. L., III 2003. The voltage-gated calcium channel gamma subunits: a review of the literature. J Bioenerg Biomembr 35, 649-660.

Bogdanov, Y., Brice, N. L., Canti, C., Page, K. M., Li, M., Volsen, S. G., and Dolphin, A. C. 2000. Acidic motif responsible for plasma membrane association of the voltage-dependent calcium channel beta1b subunit. Eur J Neurosci 12, 894-902.

Bohn, G., Moosmang, S., Conrad, H., Ludwig, A., Hofmann, F., and Klugbauer, N. 2000. Expression of T- and L-type calcium channel mRNA in murine sinoatrial node. FEBS Lett 481, 73-76.

Boixel, C., Gonzalez, W., Louedec, L., and Hatem, S. N. 2001. Mechanisms of L-type Ca(2+) current downregulation in rat atrial myocytes during heart failure. Circ Res 89, 607-613.

Borda, E., Pascual, J., Cossio, P., De, L., V, Arana, R., and Sterin-Borda, L. 1984. A circulating IgG in Chagas' disease which binds to beta-adrenoceptors of myocardium and modulates their activity. Clin Exp Immunol 57, 679-686.

Borda, E. S. and Sterin-Borda, L. 1996. Antiadrenergic and muscarinic receptor antibodies in Chagas' cardiomyopathy. International Journal of Cardiology 54, 149-156.

Bosch, R. F., Scherer, C. R., Rub, N., Wohrl, S., Steinmeyer, K., Haase, H., Busch, A. E., Seipel, L., and Kuhlkamp, V. 2003. Molecular mechanisms of early electrical remodeling: transcriptional downregulation of ion channel subunits reduces I(Ca,L) and I(to) in rapid atrial pacing in rabbits. J Am Coll Cardiol 41, 858-869.

Bourinet, E., Fournier, F., Lory, P., Charnet, P., and Nargeot, J. 1992. Protein kinase C regulation of cardiac calcium channels expressed in Xenopus oocytes. Pflugers Arch 421, 247-255.

Bourinet, E., Stotz, S. C., Spaetgens, R. L., Dayanithi, G., Lemos, J., Nargeot, J., and Zamponi, G. W. 2001. Interaction of SNX482 with domains III and IV inhibits activation gating of alpha(1E) (Ca(V)2.3) calcium channels. Biophys J 81, 79-88.

Boutjdir, M. 2000. Molecular and ionic basis of congenital complete heart block. Trends Cardiovasc Med 10, 114-122.

Boutjdir, M., Chen, L., Zhang, Z. H., Tseng, C. E., DiDonato, F., Rashbaum, W., Morris, A., el Sherif, N., and Buyon, J. P. 1997. Arrhythmogenicity of IgG and anti-52-kD SSA/Ro affinity-purified antibodies from mothers of children with congenital heart block. Circ Res 80, 354-362.

Boutjdir, M., Chen, L., Zhang, Z. H., Tseng, C. E., el Sherif, N., and Buyon, J. P. 1998. Serum and immunoglobulin G from the mother of a child with congenital heart block induce conduction abnormalities and inhibit L-type calcium channels in a rat heart model. Pediatr Res 44, 11-19.

Boyett, M. R., Honjo, H., Harrison, S. M., Zang, W. J., and Kirby, M. S. 1994. Ultra-slow voltagedependent inactivation of the calcium current in guinea-pig and ferret ventricular myocytes. Pflugers Arch 428, 39-50.

Brandt, N. 1985. Identification of two populations of cardiac microsomes with nitrendipine receptors: correlation of the distribution of dihydropyridine receptors with organelle specific markers. Arch Biochem Biophys 242, 306-319.

Brehm, P. and Eckert, R. 1978. Calcium entry leads to inactivation of calcium channel in Paramecium. Science 202, 1203-1206.

Brette, F., Lacampagne, A., Salle, L., Findlay, I., and Le Guennec, J. Y. 2003a. Intracellular Cs+ activates the PKA pathway, revealing a fast, reversible, Ca2+-dependent inactivation of L-type Ca2+ current. Am J Physiol Cell Physiol 285, C310-C318.

Brette, F. and Orchard, C. H. 2003. T-tubule function in mammalian cardiac myocytes. Circ Res 92, 1182-1192.

Brette, F., Rodriguez, P., Komukai, K., Colyer, J., and Orchard, C. H. 2004a. beta-adrenergic stimulation restores the Ca transient of ventricular myocytes lacking t-tubules. J Mol Cell Cardiol 36, 265-275.

Brette, F., Salle, L., and Orchard, C. H. 2004b. Differential modulation of L-type Ca2+ current by SR Ca2+ release at the T-tubules and surface membrane of rat ventricular myocytes. Circ Res 95, e1-e7.

Brette, F., Le Guennec, J. Y., and Findlay, I. 2003b. Low-voltage triggering of Ca2+ release from the sarcoplasmic reticulum in cardiac muscle cells. Am J Physiol 285, C1544-C1552.

Brice, N. L., Berrow, N. S., Campbell, V., Page, K. M., Brickley, K., Tedder, I., and Dolphin, A. C. 1997. Importance of the different beta subunits in the membrane expression of the alpha1A and alpha2 calcium channel subunits: studies using a depolarization-sensitive alpha1A antibody. Eur J Neurosci 9, 749-759.

Brice, N. L. and Dolphin, A. C. 1999. Differential plasma membrane targeting of voltage-dependent calcium channel subunits expressed in a polarized epithelial cell line. J Physiol 515 (Pt 3), 685-694.

Brodde, O. E. and Michel, M. C. 1999. Adrenergic and muscarinic receptors in the human heart. Pharmacol Rev 51, 651-690.

Bueno, O. F., Wilkins, B. J., Tymitz, K. M., Glascock, B. J., Kimball, T. F., Lorenz, J. N., and Molkentin, J. D. 2002. Impaired cardiac hypertrophic response in Calcineurin Abeta -deficient mice. Proc Natl Acad Sci U S A 99, 4586-4591.

Burgess, D. L., Davis, C. F., Gefrides, L. A., and Noebels, J. L. 1999. Identification of three novel Ca(2+) channel gamma subunit genes reveals molecular diversification by tandem and chromosome duplication. Genome Res 9, 1204-1213.

Caforio, A. L. P., Mahon, N. J., Tona, F., and McKenna, W. J. 2002. Circulating cardiac autoantibodies in dilated cardiomyopathy and myocarditis: pathogenetic and clinical significance. European Journal of Heart Failure 4, 411-417.

Cai, D., Mulle, J. G., and Yue, D. T. 1997. Inhibition of recombinant Ca2+ channels by benzothiazepines and phenylalkylamines: class-specific pharmacology and underlying molecular determinants. Mol Pharmacol 51, 872-881.

Callewaert, G., Cleemann, L., and Morad, M. 1988. Epinephrine enhances Ca2+ current-regulated Ca2+ release and Ca2+ reuptake in rat ventricular myocytes. Proc Natl Acad Sci U S A 85, 2009-2013.

Campbell, D. L., Stamler, J. S., and Strauss, H. C. 1996. Redox modulation of L-type calcium channels in ferret ventricular myocytes. Dual mechanism regulation by nitric oxide and S-nitrosothiols. J Gen PHYSIOL 108, 277-293.

Cannell, M. B., Berlin, J. R., and Lederer, W. J. 1987. Effect of membrane potential changes on the calcium transient in single rat cardiac muscle cells. Science 238, 1419-1423.

Canti, C., Davies, A., Berrow, N. S., Butcher, A. J., Page, K. M., and Dolphin, A. C. 2001. Evidence for two concentration-dependent processes for beta-subunit effects on alpha1B calcium channels. Biophys J 81, 1439-1451.

Canti, C., Davies, A., and Dolphin, A. C. 2003. Calcium channel Alpha2 Delta Subunits: Structure, Functions and Target Site for Drug. Current Neuropharmacology 1, 209-217.

Carafoli, E. 2002. Calcium signaling: a tale for all seasons. Proc Natl Acad Sci U S A 99, 1115-1122.

Carl, S. L., Felix, K., Caswell, A. H., Brandt, N. R., Ball, W. J., Jr., Vaghy, P. L., Meissner, G., and Ferguson, D. G. 1995. Immunolocalization of sarcolemmal dihydropyridine receptor and sarcoplasmic reticular triadin and ryanodine receptor in rabbit ventricle and atrium. J Cell Biol 129, 672-682.

Carrion, A. M., Link, W. A., Ledo, F., Mellstrom, B., and Naranjo, J. R. 1999. DREAM is a Ca2+-regulated transcriptional repressor. Nature 398, 80-84.

Castellano, A., Wei, X., Birnbaumer, L., and Perez-Reyes, E. 1993a. Cloning and expression of a neuronal calcium channel beta subunit. J Biol Chem 268, 12359-12366.

Castellano, A., Wei, X., Birnbaumer, L., and Perez-Reyes, E. 1993b. Cloning and expression of a third calcium channel beta subunit. J Biol Chem 268, 3450-3455.

Catterall, W. A. 2000. Structure and regulation of voltage-gated Ca2+ channels. Annu Rev Cell Dev Biol 16, 521-555.

Catterall, W. A., Seagar, M. J., and Takahashi, M. 1988. Molecular properties of dihydropyridinesensitive calcium channels in skeletal muscle. J Biol Chem 263, 3535-3538.

Cens, T., Restituito, S., and Charnet, P. 1999. Regulation of Ca-sensitive inactivation of a 1-type Ca2+ channel by specific domains of beta subunits. FEBS Lett 450, 17-22.

Cens, T., Restituito, S., Vallentin, A., and Charnet, P. 1998. Promotion and inhibition of L-type Ca2+ channel facilitation by distinct domains of the subunit. J Biol Chem 273, 18308-18315.

Chatterjee, K., Harris, A. M., Davies, J. G., and Leatham, A. 1969. T-wave changes after artificial pacing. Lancet 1, 759-760.

Chemin, J., Monteil, A., Bourinet, E., Nargeot, J., and Lory, P. 2001. Alternatively Spliced {alpha}1G (CaV3.1) Intracellular Loops Promote Specific T-Type Ca2+ Channel Gating Properties. Biophys J 80, 1238-1250.

Chemin, J., Monteil, A., Perez-Reyes, E., Bourinet, E., Nargeot, J., and Lory, P. 2002. Specific contribution of human T-type calcium channel isotypes (alpha1G, alpha1H and alpha1I) to neuronal excitability. J Physiol (Lond) 540, 3-14.

Chen, C. C., Lamping, K. G., Nuno, D. W., Barresi, R., Prouty, S. J., Lavoie, J. L., Cribbs, L. L., England, S. K., Sigmund, C. D., Weiss, R. M., Williamson, R. A., Hill, J. A., and Campbell, K. P. 2003. Abnormal coronary function in mice deficient in alpha1H T-type Ca2+ channels. Science 302, 1416-1418.

Chen, X., Piacentino, V., III, Furukawa, S., Goldman, B., Margulies, K. B., and Houser, S. R. 2002. L-type Ca2+ channel density and regulation are altered in failing human ventricular myocytes and recover after support with mechanical assist devices. Circ Res 91, 517-524.

Chen-Izu, Y., Xiao, R. P., Izu, L. T., Cheng, H., Kuschel, M., Spurgeon, H., and Lakatta, E. G. 2000. G(i)-dependent localization of beta(2)-adrenergic receptor signaling to L-type Ca(2+) channels. Biophys J 79, 2547-2556.

Cheng, H., Lederer, W. J., and Cannell, M. B. 1993. Calcium sparks: elementary events underlying excitation-contraction coupling in heart muscle. Science 262, 740-744.

Chiale, P. A., Ferrari, I., Mahler, E., Vallazza, M. A., Elizari, M. V., Rosenbaum, M. B., and Levin, M. J. 2001. Differential Profile and Biochemical Effects of Antiautonomic Membrane Receptor Antibodies in Ventricular Arrhythmias and Sinus Node Dysfunction. Circulation 103, 1765-1771.

Chien, A. J., Carr, K. M., Shirokov, R. E., Rios, E., and Hosey, M. M. 1996. Identification of palmitoylation sites within the L-type calcium channel beta2a subunit and effects on channel function. J Biol Chem 271, 26465-26468.

Chien, A. J., Zhao, X., Shirokov, R. E., Puri, T. S., Chang, C. F., Sun, D., Rios, E., and Hosey, M. M. 1995. Roles of a membrane-localized beta subunit in the formation and targeting of functional L-type Ca2+ channels. J Biol Chem 270, 30036-30044.

Choi, K. L., Aldrich, R. W., and Yellen, G. 1991. Tetraethylammonium Blockade Distinguishes Two Inactivation Mechanisms in Voltage-Activated K+ Channels. PNAS 88, 5092-5095.

Chu, P. J., Larsen, J. K., Chen, C. C., and Best, P. M. 2004. Distribution and relative expression levels of calcium channel beta subunits within the chambers of the rat heart. J Mol Cell Cardiol 36, 423-434.

Chu, P. J., Robertson, H. M., and Best, P. M. 2001. Calcium channel gamma subunits provide insights into the evolution of this gene family. Gene 280, 37-48.

Cibulsky, S. M. and Sather, W. A. 2000. The EEEE locus is the sole high-affinity Ca(2+) binding structure in the pore of a voltage-gated Ca(2+) channel: block by ca(2+) entering from the intracellular pore entrance. J Gen Physiol 116, 349-362.

Cibulsky, S. M. and Sather, W. A. 2003. Control of ion conduction in L-type Ca2+ channels by the concerted action of S5-6 regions. Biophys J 84, 1709-1719.

Clipstone, N. A. and Crabtree, G. R. 1992. Identification of calcineurin as a key signalling enzyme in T-lymphocyte activation. Nature 357, 695-697.

Cohen, N. M. and Lederer, W. J. 1988. Changes in the calcium current of rat heart ventricular myocytes during development. J Physiol 406, 115-146.

Colecraft, H. M., Alseikhan, B., Takahashi, S. X., Chaudhuri, D., Mittman, S., Yegnasubramanian, V., Alvania, R. S., Johns, D. C., Marban, E., and Yue, D. T. 2002. Novel functional properties of Ca(2+) channel beta subunits revealed by their expression in adult rat heart cells. J Physiol 541, 435-452.

Collin, T., Wang, J. J., Nargeot, J., and Schwartz, A. 1993. Molecular cloning of three isoforms of the L-type voltage-dependent calcium channel beta subunit from normal human heart. Circ Res 72, 1337-1344.

Costantin, J. L., Qin, N., Zhou, J., Platano, D., Birnbaumer, L., and Stefani, E. 1998. Long lasting facilitation of the rabbit cardiac Ca2+ channel: correlation with the coupling efficiency between charge movement and pore opening. FEBS Lett 423, 213-217.

Crabtree, G. R. 2001. Calcium, calcineurin, and the control of transcription. J Biol Chem 276, 2313-2316.

Cribbs, L. L., Lee, J. H., Yang, J., Satin, J., Zhang, Y., Daud, A., Barclay, J., Williamson, M. P., Fox, M., Rees, M., and Perez-Reyes, E. 1998. Cloning and characterization of alpha1H from human heart, a member of the T-type Ca2+ channel gene family. Circ Res 83, 103-109.

Cribbs, L. L., Martin, B. L., Schroder, E. A., Keller, B. B., Delisle, B. P., and Satin, J. 2001. Identification of the t-type calcium channel (Ca(v)3.1d) in developing mouse heart. Circ Res 88, 403-407.

Dai, S., Klugbauer, N., Zong, X., Seisenberger, C., and Hofmann, F. 1999. The role of subunit composition on prepulse facilitation of the cardiac L-type calcium channel. FEBS Lett 442, 70-74.

Davare, M. A., Avdonin, V., Hall, D. D., Peden, E. M., Burette, A., Weinberg, R. J., Horne, M. C., Hoshi, T., and Hell, J. W. 2001. A beta2 adrenergic receptor signaling complex assembled with the Ca2+ channel Cav1.2. Science 293, 98-101.

De Jongh, K. S., Merrick, D. K., and Catterall, W. A. 1989. Subunits of purified calcium channels: a 212-kDa form of alpha 1 and partial amino acid sequence of a phosphorylation site of an independent beta subunit. Proc Natl Acad Sci U S A 86, 8585-8589.

De Jongh, K. S., Murphy, B. J., Colvin, A. A., Hell, J. W., Takahashi, M., and Catterall, W. A. 1996. Specific phosphorylation of a site in the full-length form of the alpha 1 subunit of the cardiac
L-type calcium channel by adenosine 3',5'-cyclic monophosphate-dependent protein kinase. Biochemistry 35, 10392-10402.

De Jongh, K. S., Warner, C., and Catterall, W. A. 1990. Subunits of purified calcium channels. Alpha 2 and delta are encoded by the same gene. J Biol Chem 265, 14738-14741.

de Leon, M., Wang, Y., Jones, L., Perez-Reyes, E., Wei, X., Soong, T. W., Snutch, T. P., and Yue, D. T. 1995. Essential Ca(2+)-binding motif for Ca(2+)-sensitive inactivation of L- type Ca2+ channels. Science 270, 1502-1506.

De Waard, M. and Campbell, K. P. 1995. Subunit regulation of the neuronal alpha 1A Ca2+ channel expressed in Xenopus oocytes. J Physiol 485 (Pt 3), 619-634.

De Waard, M., Pragnell, M., and Campbell, K. P. 1994. Ca2+ channel regulation by a conserved beta subunit domain. Neuron 13, 495-503.

Deisseroth, K., Bito, H., and Tsien, R. W. 1996. Signaling from synapse to nucleus: postsynaptic CREB phosphorylation during multiple forms of hippocampal synaptic plasticity. Neuron 16, 89-101.

Deisseroth, K., Heist, E. K., and Tsien, R. W. 1998. Translocation of calmodulin to the nucleus supports CREB phosphorylation in hippocampal neurons. Nature 392, 198-202.

Delbridge, L. M., Bassani, J. W., and Bers, D. M. 1996. Steady-state twitch Ca2+ fluxes and cytosolic Ca2+ buffering in rabbit ventricular myocytes. Am J Physiol 270, C192-C199.

Delbridge, L. M., Satoh, H., Yuan, W., Bassani, J. W., Qi, M., Ginsburg, K. S., Samarel, A. M., and Bers, D. M. 1997. Cardiac myocyte volume, Ca2+ fluxes, and sarcoplasmic reticulum loading in pressure-overload hypertrophy. Am J Physiol 272, H2425-H2435.

Delgado, C., Artiles, A., Gomez, A. M., and Vassort, G. 1999. Frequency-dependent increase in cardiac Ca2+ current is due to reduced Ca2+ release by the sarcoplasmic reticulum. J Mol Cell Cardiol 31, 1783-1793.

DelPrincipe, F., Egger, M., and Niggli, E. 2000. L-type Ca2+ current as the predominant pathway of Ca2+ entry during I(Na) activation in beta-stimulated cardiac myocytes. J Physiol 527 Pt 3, 455-466.

Diebold, R. J., Ellinor, P. T., Wang, J., Muthuchamy, M., Wieczorek, D. F., and Schwartz, A. 1992. Mutually Exclusive Exon Splicing of the Cardiac Calcium Channel {alpha}1 Subunit Gene Generates Developmentally Regulated Isoforms in the Rat Heart. PNAS 89, 1497-1501.

Dittrich, M., Jurevicius, J., Georget, M., Rochais, F., Fleischmann, B., Hescheler, J., and Fischmeister, R. 2001. Local response of L-type Ca(2+) current to nitric oxide in frog ventricular myocytes. J Physiol 534, 109-121.

Doering, C. J. and Zamponi, G. W. 2003. Molecular pharmacology of high voltage-activated calcium channels. J Bioenerg Biomembr 35, 491-505.

Doerr, T., Denger, R., and Trautwein, W. 1989. Calcium currents in single SA nodal cells of the rabbit heart studied with action potential clamp. Pflugers Arch 413, 599-603.

Dolmetsch, R. 2003. Excitation-transcription coupling: signaling by ion channels to the nucleus. Sci STKE 2003, E4.

Dolmetsch, R. E., Lewis, R. S., Goodnow, C. C., and Healy, J. I. 1997. Differential activation of transcription factors induced by Ca2+ response amplitude and duration. Nature 386, 855-858.

Dolmetsch, R. E., Pajvani, U., Fife, K., Spotts, J. M., and Greenberg, M. E. 2001. Signaling to the nucleus by an L-type calcium channel-calmodulin complex through the MAP kinase pathway. Science 294, 333-339.

Dolphin, A. C. 2003a. Beta subunits of voltage-gated calcium channels. J Bioenerg Biomembr 35, 599-620.

Dolphin, A. C. 2003b. G protein modulation of voltage-gated calcium channels. Pharmacol Rev 55, 607-627.

Dolphin, A. C., Wyatt, C. N., Richards, J., Beattie, R. E., Craig, P., Lee, J. H., Cribbs, L. L., Volsen, S. G., and Perez-Reyes, E. 1999. The effect of alpha2-delta and other accessory subunits on expression and properties of the calcium channel alpha1G. J Physiol 519 Pt 1, 35-45.

Doyle, D. A., Morais, C. J., Pfuetzner, R. A., Kuo, A., Gulbis, J. M., Cohen, S. L., Chait, B. T., and MacKinnon, R. 1998. The structure of the potassium channel: molecular basis of K+ conduction and selectivity. Science 280, 69-77.

Draper, M. H. and Weidmann, S. 1951. Cardiac resting and action potentials recorded with an extracellular electrode. J Physiol (Lond) 115, 74-94.

Dubel, S. J., Altier, C., Chaumont, S., Lory, P., Bourinet, E., and Nargeot, J. 2004. Plasma membrane expression of T-type calcium channel alpha 1 subunits is modulated by HVA auxiliary subunits. J Biol Chem.

Dzhura, I., Wu, Y., Colbran, R. J., Balser, J. R., and Anderson, M. E. 2000. Calmodulin kinase determines calcium-dependent facilitation of L-type calcium channels. Nat Cell Biol 2, 173-177.

Eberst, R., Dai, S., Klugbauer, N., and Hofmann, F. 1997. Identification and functional characterization of a calcium channel gamma subunit. Pflugers Arch 433, 633-637.

Eftekhari, P., Roegel, J. C., Lezoualc'h, F., Fischmeister, R., Imbs, J. L., and Hoebeke, J. 2001. Induction of neonatal lupus in pups of mice immunized with synthetic peptides derived from amino acid sequences of the serotoninergic 5-HT4 receptor. Eur J Immunol 31, 573-579.

Eftekhari, P., Salle, L., Lezoualc'h, F., Mialet, J., Gastineau, M., Briand, J. P., Isenberg, D. A., Fournie, G. J., Argibay, J., Fischmeister, R., Muller, S., and Hoebeke, J. 2000. Anti-SSA/Ro52 autoantibodies blocking the cardiac 5-HT4 serotoninergic receptor could explain neonatal lupus congenital heart block. Eur J Immunol 30, 2782-2790.

Eisner, D. A., Choi, H. S., Diaz, M. E., O'Neill, S. C., and Trafford, A. W. 2000. Integrative analysis of calcium cycling in cardiac muscle. Circ Res 87, 1087-1094.

Elies, R., Ferrari, I., Wallukat, G., Lebesgue, D., Chiale, P., Elizari, M., Rosenbaum, M., Hoebeke, J., and Levin, M. J. 1996. Structural and functional analysis of the B cell epitopes recognized by anti-receptor autoantibodies in patients with Chagas' disease. J Immunol 157, 4203-4211.

Elies, R., Fu, L. X. M., Eftekhari, P., Wallukat, G., Schulze, W., Granier, C., Hjalmarson, A., and Hoebeke, J. 1998. Immunochemical and functional characterization of an agonist-like monoclonal antibody against the M2 acetylcholine receptor. European Journal of Biochemistry 251, 659-666.

Ellis, S. B., Williams, M. E., Ways, N. R., Brenner, R., Sharp, A. H., Leung, A. T., Campbell, K. P., McKenna, E., Koch, W. J., Hui, A., and . 1988. Sequence and expression of mRNAs encoding the alpha 1 and alpha 2 subunits of a DHP-sensitive calcium channel. Science 241, 1661-1664.

Endo, M., Tanaka, M., and Ogawa, Y. 1970. Calcium induced release of calcium from the sarcoplasmic reticulum of skinned skeletal muscle fibres. Nature 228, 34-36.

Erickson, M. G., Alseikhan, B. A., Peterson, B. Z., and Yue, D. T. 2001. Preassociation of calmodulin with voltage-gated Ca(2+) channels revealed by FRET in single living cells. Neuron 31, 973-985.

Erickson, M. G., Liang, H., Mori, M. X., and Yue, D. T. 2003. FRET two-hybrid mapping reveals function and location of L-type Ca2+ channel CaM preassociation. Neuron 39, 97-107.

Ertel, E. A., Campbell, K. P., Harpold, M. M., Hofmann, F., Mori, Y., Perez-Reyes, E., Schwartz, A., Snutch, T. P., Tanabe, T., Birnbaumer, L., Tsien, R. W., and Catterall, W. A. 2000. Nomenclature of voltage-gated calcium channels. Neuron 25, 533-535.

Fabiato, A. 1982. Calcium release in skinned cardiac cells: variations with species, tissues, and development. Fed Proc 41, 2238-2244.

Fabiato, A. 1983. Calcium-induced release of calcium from the cardiac sarcoplasmic reticulum. Am J Physiol 245, C1-14.

Fabiato, A. 1985. Simulated calcium current can both cause calcium loading in and trigger calcium release from the sarcoplasmic reticulum of a skinned canine cardiac Purkinje cell. J Gen Physiol 85, 291-320.

Fabiato, A. and Fabiato, F. 1973. Activation of skinned cardiac cells. Subcellular effects of cardioactive drugs. Eur J Cardiol 1, 143-155.

Fabiato, A. and Fabiato, F. 1975. Contractions induced by a calcium-triggered release of calcium from the sarcoplasmic reticulum of single skinned cardiac cells. J Physiol (Lond) 249, 469-495.

Fabiato, A. and Fabiato, F. 1978. Calcium-induced release of calcium from the sarcoplasmic reticulum of skinned cells from adult human, dog, cat, rabbit, rat, and frog hearts and from fetal and new-born rat ventricles. Ann N Y Acad Sci 307, 491-522.

Fares, N., Gomez, J. P., and Potreau, D. 1996. T-type calcium current is expressed in dedifferentiated adult rat ventricular cells in primary culture. C R Acad Sci III 319, 569-576.

Fauconnier, J., Bedut, S., Le Guennec, J. Y., Babuty, D., and Richard, S. 2003. Ca2+ currentmediated regulation of action potential by pacing rate in rat ventricular myocytes. Cardiovasc Res 57, 670-680.

Fedida, D., Noble, D., and Spindler, A. J. 1988a. Mechanism of the use dependence of Ca2+ current in guinea-pig myocytes. J Physiol (Lond) 405, 461-475.

Fedida, D., Noble, D., and Spindler, A. J. 1988b. Use-dependent reduction and facilitation of Ca2+ current in guinea-pig myocytes. J Physiol (Lond) 405, 439-460.

Felix, R., Gurnett, C. A., De Waard, M., and Campbell, K. P. 1997. Dissection of Functional Domains of the Voltage-Dependent Ca2+ Channel alpha 2delta Subunit. J Neurosci 17, 6884-6891.

Fenelon, G., Wijns, W., Andries, E., and Brugada, P. 1996. Tachycardiomyopathy: mechanisms and clinical implications. Pacing Clin Electrophysiol 19, 95-106.

Feron, O., Dessy, C., Opel, D. J., Arstall, M. A., Kelly, R. A., and Michel, T. 1998. Modulation of the endothelial nitric-oxide synthase-caveolin interaction in cardiac myocytes. Implications for the autonomic regulation of heart rate. J Biol Chem 273, 30249-30254.

Feron, O., Octave, J. N., Christen, M. O., and Godfraind, T. 1994. Quantification of two splicing events in the L-type calcium channel alpha-1 subunit of intestinal smooth muscle and other tissues. Eur J Biochem 222, 195-202.

Feron, O., Smith, T. W., Michel, T., and Kelly, R. A. 1997. Dynamic targeting of the agoniststimulated m2 muscarinic acetylcholine receptor to caveolae in cardiac myocytes. J Biol Chem 272, 17744-17748.

Ferrier, G. R. and Howlett, S. E. 1995. Contractions in guinea-pig ventricular myocytes triggered by a calcium-release mechanism separate from Na+ and L-currents. J Physiol 484 (Pt 1), 107-122.

Ferrier, G. R. and Howlett, S. E. 2001. Cardiac excitation-contraction coupling: role of membrane potential in regulation of contraction. Am J Physiol Heart Circ Physiol 280, H1928-H1944.

Ferrier, G. R., Zhu, J., Redondo, I. M., and Howlett, S. E. 1998. Role of cAMP-dependent protein kinase A in activation of a voltage-sensitive release mechanism for cardiac contraction in guinea-pig myocytes. J Physiol 513 (Pt 1), 185-201.

Ferron, L., Capuano, V., Ruchon, Y., Deroubaix, E., Coulombe, A., and Renaud, J. F. 2003. Angiotensin II signaling pathways mediate expression of cardiac T-type calcium channels. Circ Res 93, 1241-1248.

Feske, S., Giltnane, J., Dolmetsch, R., Staudt, L. M., and Rao, A. 2001. Gene regulation mediated by calcium signals in T lymphocytes. Nat Immunol 2, 316-324.

Findlay, I. 2004. Physiological modulation of inactivation in L-type Ca(2+) channels: one switch. J Physiol 554, 275-283.

Finlin, B. S., Crump, S. M., Satin, J., and Andres, D. A. 2003. Regulation of voltage-gated calcium channel activity by the Rem and Rad GTPases. PNAS 100, 14469-14474.

Fischmeister, R. and Hartzell, H. C. 1987. Cyclic guanosine 3',5'-monophosphate regulates the calcium current in single cells from frog ventricle. J Physiol 387, 453-472.

Ford, L. E. and Podolsky, R. J. 1970. Regenerative calcium release within muscle cells. Science 167, 58-59.

Franzini-Armstrong, C., Protasi, F., and Ramesh, V. 1999. Shape, size, and distribution of Ca(2+) release units and couplons in skeletal and cardiac muscles. Biophys J 77, 1528-1539.

Freise, D., Himmerkus, N., Schroth, G., Trost, C., Weissgerber, P., Freichel, M., and Flockerzi, V. 1999. Mutations of calcium channel beta subunit genes in mice. Biol Chem 380, 897-902.

Fu, L. X., Magnusson, Y., Bergh, C. H., Liljeqvist, J. A., Waagstein, F., Hjalmarson, A., and Hoebeke, J. 1993. Localization of a functional autoimmune epitope on the muscarinic acetylcholine receptor-2 in patients with idiopathic dilated cardiomyopathy. J Clin Invest 91, 1964-1968.

Fu, M. L. 1995. Anti-peptide antibodies against an autoimmune epitope on human muscarinic receptor mimic functional autoantibodies against the same epitope in patients with idiopathic dilated cardiomyopathy. Eur Heart J 16 Suppl O, 89-91.

Gallo, M. P., Ghigo, D., Bosia, A., Alloatti, G., Costamagna, C., Penna, C., and Levi, R. C. 1998. Modulation of guinea-pig cardiac L-type calcium current by nitric oxide synthase inhibitors. J Physiol 506 (Pt 3), 639-651.

Gallo, M. P., Malan, D., Bedendi, I., Biasin, C., Alloatti, G., and Levi, R. C. 2001. Regulation of cardiac calcium current by NO and cGMP-modulating agents. Pflugers Arch 441, 621-628.

Gao, B., Sekido, Y., Maximov, A., Saad, M., Forgacs, E., Latif, F., Wei, M. H., Lerman, M., Lee, J. H., Perez-Reyes, E., Bezprozvanny, I., and Minna, J. D. 2000. Functional Properties of a New Voltage-dependent Calcium Channel alpha 2delta Auxiliary Subunit Gene (CACNA2D2). J Biol Chem 275, 12237-12242.

Gao, T., Chien, A. J., and Hosey, M. M. 1999. Complexes of the alpha1C and beta subunits generate the necessary signal for membrane targeting of class C L-type calcium channels. J Biol Chem 274, 2137-2144.

Gao, T., Puri, T. S., Gerhardstein, B. L., Chien, A. J., Green, R. D., and Hosey, M. M. 1997a. Identification and subcellular localization of the subunits of L-type calcium channels and adenylyl cyclase in cardiac myocytes. J Biol Chem 272, 19401-19407.

Gao, T., Yatani, A., Dell'Acqua, M. L., Sako, H., Green, S. A., Dascal, N., Scott, J. D., and Hosey, M. M. 1997b. cAMP-dependent regulation of cardiac L-type Ca2+ channels requires membrane targeting of PKA and phosphorylation of channel subunits. Neuron 19, 185-196.

Garcia, S., Nascimento, J. H., Bonfa, E., Levy, R., Oliveira, S. F., Tavares, A. V., and de Carvalho, A. C. 1994. Cellular mechanism of the conduction abnormalities induced by serum from anti-Ro/SSA-positive patients in rabbit hearts. J Clin Invest 93, 718-724.

Gathercole, D. V., Colling, D. J., Skepper, J. N., Takagishi, Y., Levi, A. J., and Severs, N. J. 2000. Immunogold-labeled L-type calcium channels are clustered in the surface plasma membrane overlying junctional sarcoplasmic reticulum in guinea-pig myocytes-implications for excitationcontraction coupling in cardiac muscle. J Mol Cell Cardiol 32, 1981-1994.

Gerhardstein, B. L., Puri, T. S., Chien, A. J., and Hosey, M. M. 1999. Identification of the sites phosphorylated by cyclic AMP-dependent protein kinase on the beta 2 subunit of L-type voltage-dependent calcium channels. Biochemistry 38, 10361-10370.

Gerster, U., Neuhuber, B., Groschner, K., Striessnig, J., and Flucher, B. E. 1999. Current modulation and membrane targeting of the calcium channel alpha1C subunit are independent functions of the beta subunit. J Physiol (Lond) 517, 353-368.

Glossmann, H., Striessnig, J., Hymel, L., and Schindler, H. 1987. Purified L-type calcium channels: only one single polypeptide (alpha 1-subunit) carries the drug receptor domains and is regulated by protein kinases. Biomed Biochim Acta 46, S351-S356.

Goin, J. C., Borda, E., Perez Leiros, C., Storino, R., and Sterin-Borda, L. 1994. Identification of antibodies with muscarinic cholinergic activity in human Chagas' disease: pathological implications. Journal of the Autonomic Nervous System 47, 45-52.

Goldberg, N. D., Haddox, M. K., Nicol, S. E., Glass, D. B., Sanford, C. H., Kuehl, F. A., Jr., and Estensen, R. 1975. Biologic regulation through opposing influences of cyclic GMP and cyclic AMP: the Yin Yang hypothesis. Adv Cyclic Nucleotide Res 5, 307-330.

Gomez, A. M., Guatimosim, S., Dilly, K. W., Vassort, G., and Lederer, W. J. 2001. Heart failure after myocardial infarction: altered excitation-contraction coupling. Circulation 104, 688-693.

Gomez, A. M., Valdivia, H. H., Cheng, H., Lederer, M. R., Santana, L. F., Cannell, M. B., McCune, S. A., Altschuld, R. A., and Lederer, W. J. 1997. Defective excitation-contraction coupling in experimental cardiac hypertrophy and heart failure [see comments]. Science 276, 800-806.

Gong, H. C., Hang, J., Kohler, W., Li, L., and Su, T. Z. 2001. Tissue-specific expression and gabapentin-binding properties of calcium channel alpha2delta subunit subtypes. J Membr Biol 184, 35-43.

Goodwin, L. O., Leeds, N. B., Guzowski, D., Hurley, I. R., Pergolizzi, R. G., and Benoff, S. 1999. Identification of structural elements of the testis-specific voltage dependent calcium channel that potentially regulate its biophysical properties. Mol Hum Reprod 5, 311-322.

Gorelik, J., Gu, Y., SPOHR, H. A., Shevchuk, A. I., LAB, M. J., HARDING, S. E., Edwards, C. R. W., Whitaker, M., Moss, G. W. J., Benton, D. C. H., Sanchez, D., Darszon, A., Vodyanoy, I., Klenerman, D., and KORCHEV, Y. E. 2002. Ion Channels in Small Cells and Subcellular Structures Can Be Studied with a Smart Patch-Clamp System. Biophys J 83, 3296-3303.

Grantham, C. J. and Cannell, M. B. 1996. Ca2+ influx during the cardiac action potential in guinea pig ventricular myocytes. Circ Res 79, 194-200.

Green, P. J., Warre, R., Hayes, P. D., McNaughton, N. C. L., Medhurst, A. D., Pangalos, M., Duckworth, D. M., and Randall, A. D. 2001. Kinetic modification of the {alpha}11 subunitmediated T-type Ca2+ channel by a human neuronal Ca2+ channel {gamma} subunit. J Physiol (Lond) 533, 467-478.

Gregg, R. G., Powers, P. A., and Hogan, K. 1993. Assignment of the human gene for the beta subunit of the voltage-dependent calcium channel (CACNLB1) to chromosome 17 using somatic cell hybrids and linkage mapping. Genomics 15, 185-187.

Griffiths, H. and MacLeod, K. T. 2003. The Voltage-sensitive Release Mechanism of Excitation Contraction Coupling in Rabbit Cardiac Muscle Is Explained by Calcium-induced Calcium Release. J Gen PHYSIOL 121, 353-373.

Gruver, C. L., DeMayo, F., Goldstein, M. A., and Means, A. R. 1993. Targeted developmental overexpression of calmodulin induces proliferative and hypertrophic growth of cardiomyocytes in transgenic mice. Endocrinology 133, 376-388.

Gu, Y., Gorelik, J., Spohr, H. A., Shevchuk, A., Lab, M. J., Harding, S. E., Vodyanoy, I., Klenerman, D., and Korchev, Y. E. 2002. High-resolution scanning patch-clamp: new insights into cell function. FASEB J 16, 748-750.

Guo, J. and Duff, HJ. 2003. Inactivation of ICa-L is the major determinant of use-dependent facilitation in rat cardiomyocytes. J Physiol 547, 797-805.

Gurney, A. M., Charnet, P., Pye, J. M., and Nargeot, J. 1989. Augmentation of cardiac calcium current by flash photolysis of intracellular caged-Ca2+ molecules. Nature 341, 65-68.

Haase, H., Karczewski, P., Beckert, R., and Krause, E. G. 1993. Phosphorylation of the L-type calcium channel beta subunit is involved in beta-adrenergic signal transduction in canine myocardium. FEBS Lett 335, 217-222.

Haase, H., Kresse, A., Hohaus, A., Schulte, H. D., Maier, M., Osterziel, K. J., Lange, P. E., and Morano, I. 1996. Expression of calcium channel subunits in the normal and diseased human myocardium. J Mol Med 74, 99-104.

Haddad, G. E., Sperelakis, N., and Bkaily, G. 1995. Regulation of the calcium slow channel by cyclic GMP dependent protein kinase in chick heart cells. Mol Cell Biochem 148, 89-94.

Haddock, P. S., Coetzee, W. A., Cho, E., Porter, L., Katoh, H., Bers, D. M., Jafri, M. S., and Artman, M. 1999. Subcellular [Ca2+]i gradients during excitation-contraction coupling in newborn rabbit ventricular myocytes. Circ Res 85, 415-427.

Hadley, R. W. and Lederer, W. J. 1989. Intramembrane charge movement in guinea-pig and rat ventricular myocytes. J Physiol 415, 601-624.

Hadley, R. W. and Lederer, W. J. 1991a. Ca2+ and voltage inactivate Ca2+ channels in guinea-pig ventricular myocytes through independent mechanisms. J Physiol 444, 257-268.

Hadley, R. W. and Lederer, W. J. 1991b. Properties of L-type calcium channel gating current in isolated guinea pig ventricular myocytes. J Gen Physiol 98, 265-285.

Hadley, R. and Hume, J. 1987. An intrinsic potential-dependent inactivation mechanism associated with calcium channels in guinea-pig myocytes. J PHYSIOL 389, 205-222.

Hagiwara, N., Irisawa, H., and Kameyama, M. 1988. Contribution of two types of calcium currents to the pacemaker potentials of rabbit sino-atrial node cells. J Physiol (Lond) 395, 233-253.

Hamill, O. P., Marty, A., Neher, E., Sakmann, B., and Sigworth, F. J. 1981. Improved patch-clamp techniques for high-resolution current recording from cells and cell-free membrane patches. Pflugers Arch 391, 85-100.

Han, J., Kim, E., Lee, S. H., Yoo, S., Ho, W. K., and Earm, Y. E. 1998. cGMP facilitates calcium current via cGMP-dependent protein kinase in isolated rabbit ventricular myocytes. Pflugers Arch 435, 388-393.

Hanlon, M. R., Berrow, N. S., Dolphin, A. C., and Wallace, B. A. 1999. Modelling of a voltagedependent Ca2+ channel beta subunit as a basis for understanding its functional properties. FEBS Lett 445, 366-370.

Hansma, P. K., Drake, B., Marti, O., Gould, S. A., and Prater, C. B. 1989. The scanning ion-conductance microscope. Science 243, 641-643.

Hardingham, G. E., Arnold, F. J., and Bading, H. 2001. Nuclear calcium signaling controls CREBmediated gene expression triggered by synaptic activity. Nat Neurosci 4, 261-267.

Hare, J. M. 2003. Nitric oxide and excitation-contraction coupling. J Mol Cell Cardiol 35, 719-729.

Hare, J. M. and Stamler, J. S. 1999. NOS: modulator, not mediator of cardiac performance. Nat Med 5, 273-274.

Hartzell, H. C. and Fischmeister, R. 1986. Opposite effects of cyclic GMP and cyclic AMP on Ca2+ current in single heart cells. Nature 323, 273-275.

Hartzell, H. C., Mery, P. F., Fischmeister, R., and Szabo, G. 1991. Sympathetic regulation of cardiac calcium current is due exclusively to cAMP-dependent phosphorylation. Nature 351, 573-576.

Harvey, R. D. and Belevych, A. E. 2003. Muscarinic regulation of cardiac ion channels. Br J Pharmacol 139, 1074-1084.

Hatem, S. N., Benardeau, A., Rucker-Martin, C., Marty, I., de Chamisso, P., Villaz, M., and Mercadier, J. J. 1997. Different compartments of sarcoplasmic reticulum participate in the excitation-contraction coupling process in human atrial myocytes. Circ Res 80, 345-353.

Hering, S., Aczel, S., Grabner, M., Doring, F., Berjukow, S., Mitterdorfer, J., Sinnegger, M. J., Striessnig, J., Degtiar, V. E., Wang, Z., and Glossmann, H. 1996. Transfer of High Sensitivity for Benzothiazepines from L-type to Class A (BI) Calcium Channels. J Biol Chem 271, 24471-24475.

Herlitze, S., Xie, M., Han, J., Hummer, A., Melnik-Martinez, K. V., Moreno, R. L., and Mark, M. D. 2003. Targeting mechanisms of high voltage-activated Ca2+ channels. J Bioenerg Biomembr 35, 621-637.

Hernandez, C. C. Q., Barcellos, L. C., Gimenez, L. E. D., Cabarcas, R. A. B., Garcia, S., Pedrosa, R. C., Nascimento, J. H. M., Kurtenbach, E., Masuda, M. O., and Campos de Carvalho, A. C. 2003. Human chagasic IgGs bind to cardiac muscarinic receptors and impair L-type Ca2+ currents. Cardiovascular Research 58, 55-65.

Hescheler, J., Tang, M., Jastorff, B., and Trautwein, W. 1987. On the mechanism of histamine induced enhancement of the cardiac Ca2+ current. Pflugers Arch 410, 23-29.

Hess, P., Lansman, J. B., and Tsien, R. W. 1984. Different modes of Ca channel gating behaviour favoured by dihydropyridine Ca agonists and antagonists. Nature 311, 538-544.

Hess, P., Lansman, J. B., and Tsien, R. W. 1986. Calcium channel selectivity for divalent and monovalent cations. Voltage and concentration dependence of single channel current in ventricular heart cells. J Gen Physiol 88, 293-319.

Hess, P. and Tsien, R. W. 1984. Mechanism of ion permeation through calcium channels. Nature 309, 453-456.

Hibino, H., Pironkova, R., Onwumere, O., Rousset, M., Charnet, P., Hudspeth, A. J., and Lesage, F. 2003. Direct interaction with a nuclear protein and regulation of gene silencing by a variant of the Ca2+-channel beta 4 subunit. PNAS 100, 307-312.

Higuchi, M. d. L., Benvenuti, L. A., Martins Reis, M., and Metzger, M. 2003. Pathophysiology of the heart in Chagas' disease: current status and new developments. Cardiovascular Research 60, 96-107.

Hirano, Y., Fozzard, H. A., and January, C. T. 1989. Characteristics of L- and T-type Ca2+ currents in canine cardiac Purkinje cells. Am J Physiol 256, H1478-H1492.

Hirano, Y., Yoshinaga, T., Murata, M., and Hiraoka, M. 1999. Prepulse-induced mode 2 gating behavior with and without beta-adrenergic stimulation in cardiac L-type Ca channels. Am J Physiol 276, C1338-C1345.

Hobai, I. A., Howarth, F. C., Pabbathi, V. K., Dalton, G. R., Hancox, J. C., Zhu, J. Q., Howlett, S. E., Ferrier, G. R., and Levi, A. J. 1997. "Voltage-activated Ca release" in rabbit, rat and guinea-pig cardiac myocytes, and modulation by internal cAMP. Pflugers Arch 435, 164-173.

Hobai, I. A. and O'Rourke, B. 2001. Decreased sarcoplasmic reticulum calcium content is responsible for defective excitation-contraction coupling in canine heart failure. Circulation 103, 1577-1584.

Hobom, M., Dai, S., Marais, E., Lacinova, L., Hofmann, F., and Klugbauer, N. 2000. Neuronal distribution and functional characterization of the calcium channel alpha2delta-2 subunit. Eur J Neurosci 12, 1217-1226.

Hoch, B., Meyer, R., Hetzer, R., Krause, E. G., and Karczewski, P. 1999. Identification and expression of delta-isoforms of the multifunctional Ca2+/calmodulin-dependent protein kinase in failing and nonfailing human myocardium. Circ Res 84, 713-721.

Hockerman, G. H., Johnson, B. D., Abbott, M. R., Scheuer, T., and Catterall, W. A. 1997. Molecular determinants of high affinity phenylalkylamine block of L-type calcium channels in transmembrane segment IIIS6 and the pore region of the alpha1 subunit. J Biol Chem 272, 18759-18765.

Hockerman, G. H., Johnson, B. D., Scheuer, T., and Catterall, W. A. 1995. Molecular determinants of high affinity phenylalkylamine block of L-type calcium channels. J Biol Chem 270, 22119-22122.

Hodgkin, A. L. and Huxley, A. F. 1952a. A quantitative description of membrane current and its application to conduction and excitation in nerve. J Physiol 117, 500-544.

Hodgkin, A. L. and Huxley, A. F. 1952b. Currents carried by sodium and potassium ions through the membrane of the giant axon of Loligo. J Physiol 116, 449-472.

Hoebeke, J. 2001. Molecular mechanisms of anti-G-protein-coupled receptor autoantibodies. Autoimmunity 34, 161-164.

Hordof, A. J., Edie, R., Malm, J. R., Hoffman, B. F., and Rosen, M. R. 1976. Electrophysiologic properties and response to pharmacologic agents of fibers from diseased human atria. Circulation 54, 774-779.

Houser, S. R. and Margulies, K. B. 2003. Is depressed myocyte contractility centrally involved in heart failure? Circ Res 92, 350-358.

Howlett, S. E., Zhu, J. Q., and Ferrier, G. R. 1998. Contribution of a voltage-sensitive calcium release mechanism to contraction in cardiac ventricular myocytes. Am J Physiol 274, H155-H170.

Hryshko, L. V. and Bers, D. M. 1990. Ca current facilitation during postrest recovery depends on Ca entry. Am J Physiol 259, H951-H961.

Hu, H., Chiamvimonvat, N., Yamagishi, T., and Marban, E. 1997. Direct inhibition of expressed cardiac L-type Ca2+ channels by S-nitrosothiol nitric oxide donors. Circ Res 81, 742-752.

Hu, K., Qu, Y., Yue, Y., and Boutjdir, M. 2004. Functional basis of sinus bradycardia in congenital heart block. Circ Res 94, e32-e38.

Hullin, R., Asmus, F., Ludwig, A., Hersel, J., and Boekstegers, P. 1999. Subunit expression of the cardiac L-type calcium channel is differentially regulated in diastolic heart failure of the cardiac allograft. Circulation 100, 155-163.

Hullin, R., Khan, I. F., Wirtz, S., Mohacsi, P., Varadi, G., Schwartz, A., and Herzig, S. 2003. Cardiac L-type calcium channel beta-subunits expressed in human heart have differential effects on single channel characteristics. J Biol Chem 278, 21623-21630.

Hullin, R., Singer-Lahat, D., Freichel, M., Biel, M., Dascal, N., Hofmann, F., and Flockerzi, V. 1992. Calcium channel beta subunit heterogeneity: functional expression of cloned cDNA from heart, aorta and brain. EMBO J 11, 885-890.

Hulme, J. T., Lin, T. W., Westenbroek, R. E., Scheuer, T., and Catterall, W. A. 2003. Betaadrenergic regulation requires direct anchoring of PKA to cardiac CaV1.2 channels via a leucine zipper interaction with A kinase-anchoring protein 15. Proc Natl Acad Sci U S A 100, 13093-13098.

Hume, J. R. and Uehara, A. 1985. Ionic basis of the different action potential configurations of single guinea-pig atrial and ventricular myocytes. J Physiol 368, 525-544.

Huser, J., Blatter, L. A., and Lipsius, S. L. 2000. Intracellular Ca2+ release contributes to automaticity in cat atrial pacemaker cells. J Physiol 524 Pt 2, 415-422.

Imai, Y., Jiang, B., and Pappano, A. J. 2001. Mechanism for muscarinic inhibition of ICa(L) is determined by the path for elevating cyclic AMP in cardiac myocytes. Cardiovascular Research 51, 331-343.

Isaev, D., Solt, K., Gurtovaya, O., Reeves, J. P., and Shirokov, R. 2004. Modulation of the voltage sensor of L-type Ca2+ channels by intracellular Ca2+. J Gen Physiol 123, 555-571.

Isenberg, G. and Klockner, U. 1982. Calcium currents of isolated bovine ventricular myocytes are fast and of large amplitude. Pflugers Arch 395, 30-41.

Ito, H., Klugbauer, N., and Hofmann, F. 1997. Transfer of the high affinity dihydropyridine sensitivity from L-type To non-L-type calcium channel. Mol Pharmacol 52, 735-740.

Izumi, T., Kihara, Y., Sarai, N., Yoneda, T., Iwanaga, Y., Inagaki, K., Onozawa, Y., Takenaka, H., Kita, T., and Noma, A. 2003. Reinduction of T-type calcium channels by endothelin-1 in failing hearts in vivo and in adult rat ventricular myocytes in vitro. Circulation 108, 2530-2535.

Jahns, R., Boivin, V., Hein, L., Triebel, S., Angermann, C. E., Ertl, G., and Lohse, M. J. 2004. Direct evidence for a {beta}1-adrenergic receptor-directed autoimmune attack as a cause of idiopathic dilated cardiomyopathy. J Clin Invest 113, 1419-1429.

January, C. T. and Riddle, J. M. 1989. Early afterdepolarizations: mechanism of induction and block. A role for L-type Ca2+ current. Circ Res 64, 977-990.

Jay, S. D., Ellis, S. B., McCue, A. F., Williams, M. E., Vedvick, T. S., Harpold, M. M., and Campbell, K. P. 1990. Primary structure of the gamma subunit of the DHP-sensitive calcium channel from skeletal muscle. Science 248, 490-492.

Jay, S. D., Sharp, A. H., Kahl, S. D., Vedvick, T. S., Harpold, M. M., and Campbell, K. P. 1991. Structural characterization of the dihydropyridine-sensitive calcium channel alpha 2-subunit and the associated delta peptides. J Biol Chem 266, 3287-3293. Jiang, L. H., Gawler, D. J., Hodson, N., Milligan, C. J., Pearson, H. A., Porter, V., and Wray, D. 2000. Regulation of Cloned Cardiac L-type Calcium Channels by cGMP-dependent Protein Kinase. J Biol Chem 275, 6135-6143.

Jiang, Y., Lee, A., Chen, J., Cadene, M., Chait, B. T., and MacKinnon, R. 2002a. Crystal structure and mechanism of a calcium-gated potassium channel. Nature 417, 515-522.

Jiang, Y., Lee, A., Chen, J., Cadene, M., Chait, B. T., and MacKinnon, R. 2002b. The open pore conformation of potassium channels. Nature 417, 523-526.

Jiang, Y., Lee, A., Chen, J., Ruta, V., Cadene, M., Chait, B. T., and MacKinnon, R. 2003a. X-ray structure of a voltage-dependent K+ channel. Nature 423, 33-41.

Jiang, Y., Ruta, V., Chen, J., Lee, A., and MacKinnon, R. 2003b. The principle of gating charge movement in a voltage-dependent K+ channel. Nature 423, 42-48.

Johnson, B. D., Hockerman, G. H., Scheuer, T., and Catterall, W. A. 1996. Distinct effects of mutations in transmembrane segment IVS6 on block of L-type calcium channels by structurally similar phenylalkylamines. Mol Pharmacol 50, 1388-1400.

Jones, L. P., Wei, S. K., and Yue, D. T. 1998. Mechanism of auxiliary subunit modulation of neuronal alpha1E calcium channels. J Gen Physiol 112, 125-143.

Josephson, I. R., Sanchez-Chapula, J., and Brown, A. M. 1984. A comparison of calcium currents in rat and guinea pig single ventricular cells. Circ Res 54, 144-156.

Josephson, I. R. and Varadi, G. 1996. The beta subunit increases Ca2+ currents and gating charge movements of human cardiac L-type Ca2+ channels. Biophys J 70, 1285-1293.

Jurevicius, J. and Fischmeister, R. 1996. cAMP compartmentation is responsible for a local activation of cardiac Ca2+ channels by beta-adrenergic agonists. Proc Natl Acad Sci U S A 93, 295-299.

Jurevicius, J. and Fischmeister, R. 1997. Longitudinal distribution of Na+ and Ca2+ channels and beta- adrenoceptors on the sarcolemmal membrane of frog cardiomyocytes. J Physiol (Lond) 503, 471-477.

Jurkat-Rott, K. and Lehmann-Horn, F. 2004. The impact of splice isoforms on voltage-gated calcium channel {alpha}1 subunits. J Physiol (Lond) 554, 609-619.

Kamp, T. J., Hu, H., and Marban, E. 2000. Voltage-dependent facilitation of cardiac L-type Ca channels expressed in HEK-293 cells requires beta-subunit. Am J Physiol Heart Circ Physiol 278, H126-H136.

Kang, M. G. and Campbell, K. P. 2003. Gamma subunit of voltage-activated calcium channels. J Biol Chem 278, 21315-21318.

Kannel, W. B., Abbott, R. D., Savage, D. D., and McNamara, P. M. 1982. Epidemiologic features of chronic atrial fibrillation: the Framingham study. N Engl J Med 306, 1018-1022.

Kasielke, N., Obermair, G. J., Kugler, G., Grabner, M., and Flucher, B. E. 2003. Cardiac-type ECcoupling in dysgenic myotubes restored with Ca2+ channel subunit isoforms alpha1C and alpha1D does not correlate with current density. Biophys J 84, 3816-3828. Kass, R. and Sanguinetti, M. 1984. Inactivation of calcium channel current in the calf cardiac Purkinje fiber. Evidence for voltage- and calcium-mediated mechanisms. J Gen PHYSIOL 84, 705-726.

Katoh, H., Schlotthauer, K., and Bers, D. M. 2000. Transmission of information from cardiac dihydropyridine receptor to ryanodine receptor: evidence from BayK 8644 effects on resting Ca(2+) sparks. Circ Res 87, 106-111.

Kawai, M., Hussain, M., and Orchard, C. H. 1999. Excitation-contraction coupling in rat ventricular myocytes after formamide-induced detubulation. Am J Physiol 277, H603-H609.

Keely, S. L. 1979. Prostaglandin E1 activation of heart cAMP-dependent protein kinase: apparent dissociation of protein kinase activation from increases in phosphorylase activity and contractile force. Mol Pharmacol 15, 235-245.

Khosravani, H., Altier, C., Simms, B., Hamming, K. S., Snutch, T. P., Mezeyova, J., McRory, J. E., and Zamponi, G. W. 2004. Gating effects of mutations in the Cav3.2 T-type calcium channel associated with childhood absence epilepsy. J Biol Chem 279, 9681-9684.

Kierszenbaum, F. 2003. Views on the autoimmunity hypothesis for Chagas disease pathogenesis. FEMS Immunology and Medical Microbiology 37, 1-11.

Kim, D., Song, I., Keum, S., Lee, T., Jeong, M. J., Kim, S. S., McEnery, M. W., and Shin, H. S. 2001. Lack of the burst firing of thalamocortical relay neurons and resistance to absence seizures in mice lacking alpha(1G) T-type Ca(2+) channels. Neuron 31, 35-45.

Kim, J., Ghosh, S., Nunziato, D. A., and Pitt, G. S. 2004. Identification of the components controlling inactivation of voltage-gated Ca2+ channels. Neuron 41, 745-754.

Kirstein, M., Rivet-Bastide, M., Hatem, S., Benardeau, A., Mercadier, J. J., and Fischmeister, R. 1995. Nitric oxide regulates the calcium current in isolated human atrial myocytes. J Clin Invest 95, 794-802.

Klein, G., Drexler, H., and Schroder, F. 2000. Protein kinase G reverses all isoproterenol induced changes of cardiac single L-type calcium channel gating. Cardiovascular Research 48, 367-374.

Klockner, U., Lee, J. H., Cribbs, L. L., Daud, A., Hescheler, J., Pereverzev, A., Perez-Reyes, E., and Schneider, T. 1999. Comparison of the Ca2 + currents induced by expression of three cloned alpha1 subunits, alpha1G, alpha1H and alpha1I, of low-voltage-activated T-type Ca2 + channels. Eur J Neurosci 11, 4171-4178.

Klockner, U., Mikala, G., Eisfeld, J., Iles, D. E., Strobeck, M., Mershon, J. L., Schwartz, A., and Varadi, G. 1997. Properties of three COOH-terminal splice variants of a human cardiac L-type Ca2+-channel alpha1-subunit. Am J Physiol 272, H1372-H1381.

Klockner, U., Mikala, G., Varadi, M., Varadi, G., and Schwartz, A. 1995. Involvement of the carboxyl-terminal region of the alpha 1 subunit in voltage-dependent inactivation of cardiac calcium channels. J Biol Chem 270, 17306-17310.

Klockner, U., Pereverzev, A., Leroy, J., Krieger, A., Vajna, R., Pfitzer, G., Hescheler, J., Malecot, C. O., and Schneider, T. 2004. The cytosolic II-III loop of Ca2.3 provides an essential determinant for the phorbol ester-mediated stimulation of E-type Ca channel activity. Eur J Neurosci 19, 2659-2668.

Klugbauer, N., Dai, S., Specht, V., Lacinova, L., Marais, E., Bohn, G., and Hofmann, F. 2000. A family of gamma-like calcium channel subunits. FEBS Lett 470, 189-197.

Klugbauer, N., Lacinova, L., Marais, E., Hobom, M., and Hofmann, F. 1999a. Molecular diversity of the calcium channel alpha2delta subunit. J Neurosci 19, 684-691.

Klugbauer, N., Marais, E., and Hofmann, F. 2003. Calcium channel alpha2delta subunits: differential expression, function, and drug binding. J Bioenerg Biomembr 35, 639-647.

Klugbauer, N., Marais, E., Lacinova, L., and Hofmann, F. 1999b. A T-type calcium channel from mouse brain. Pflugers Arch 437, 710-715.

Kockskamper, J., Sheehan, K. A., Bare, D. J., Lipsius, S. L., Mignery, G. A., and Blatter, L. A. 2001. Activation and propagation of Ca(2+) release during excitation- contraction coupling in atrial myocytes. Biophys J 81, 2590-2605.

Koschak, A., Reimer, D., Huber, I., Grabner, M., Glossmann, H., Engel, J., and Striessnig, J. 2001. alpha 1D (Cav1.3) subunits can form l-type Ca2+ channels activating at negative voltages. J Biol Chem 276, 22100-22106.

Koschak, A., Reimer, D., Walter, D., Hoda, J. C., Heinzle, T., Grabner, M., and Striessnig, J. 2003. Cav1.4alpha1 subunits can form slowly inactivating dihydropyridine-sensitive L-type Ca2+ channels lacking Ca2+-dependent inactivation. J Neurosci 23, 6041-6049.

Kraus, R., Reichl, B., Kimball, S. D., Grabner, M., Murphy, B. J., Catterall, W. A., and Striessnig, J. 1996. Identification of benz(othi)azepine-binding regions within L-type calcium channel alpha1 subunits. J Biol Chem 271, 20113-20118.

Kuhl, U., Melzner, B., Schafer, B., Schultheiss, H. P., and Strauer, B. E. 1991. The Ca-channel as cardiac autoantigen. Eur Heart J 12 Suppl D, 99-104.

Kumar, R., Joyner, R. W., Komalavilas, P., and Lincoln, T. M. 1999. Analysis of Expression of cGMP-Dependent Protein Kinase in Rabbit Heart Cells. J Pharmacol Exp Ther 291, 967-975.

Kumar, R., Namiki, T., and Joyner, R. W. 1997. Effects of cGMP on L-type calcium current of adult and newborn rabbit ventricular cells. Cardiovascular Research 33, 573-582.

Kuschel, M., Zhou, Y. Y., Cheng, H., Zhang, S. J., Chen, Y., Lakatta, E. G., and Xiao, R. P. 1999. G(i) protein-mediated functional compartmentalization of cardiac beta(2)-adrenergic signaling. J Biol Chem 274, 22048-22052.

Kuznetsov, V., Pak, E., Robinson, R. B., and Steinberg, S. F. 1995. Beta 2-adrenergic receptor actions in neonatal and adult rat ventricular myocytes. Circ Res 76, 40-52.

Lacinova, L. and Klugbauer, N. 2004. Modulation of gating currents of the Ca(v)3.1 calcium channel by alpha 2 delta 2 and gamma 5 subunits. Arch Biochem Biophys 425, 207-213.

Laflamme, M. A. and Becker, P. L. 1998. Do beta 2-adrenergic receptors modulate Ca2+ in adult rat ventricular myocytes? Am J Physiol 274, H1308-H1314.

Laflamme, M. A. and Becker, P. L. 1999. G(s) and adenylyl cyclase in transverse tubules of heart: implications for cAMP-dependent signaling. Am J Physiol 277, H1841-H1848.

Lansman, J. B., Hess, P., and Tsien, R. W. 1986. Blockade of current through single calcium channels by Cd2+, Mg2+, and Ca2+. Voltage and concentration dependence of calcium entry into the pore. J Gen Physiol 88, 321-347.

Le Grand, B., Hatem, S., Deroubaix, E., Couetil, J. P., and Coraboeuf, E. 1991. Calcium current depression in isolated human atrial myocytes after cessation of chronic treatment with calcium antagonists. Circ Res 69, 292-300.

Le Guennec, J. V. and Noble, D. 1994. Effects of rapid changes of external Na+ concentration at different moments during the action potential in guinea-pig myocytes. J Physiol 478 Pt 3, 493-504.

Lebesgue, D., Wallukat, G., Mijares, A., Granier, C., Argibay, J., and Hoebeke, J. 1998. An agonistlike monoclonal antibody against the human [beta]2-adrenoceptor. European Journal of Pharmacology 348, 123-133.

Lederer, W. J., Niggli, E., and Hadley, R. W. 1990. Sodium-calcium exchange in excitable cells: fuzzy space. Science 248, 283.

Ledo, F., Kremer, L., Mellstrom, B., and Naranjo, J. R. 2002. Ca2+-dependent block of CREB-CBP transcription by repressor DREAM. EMBO J 21, 4583-4592.

Lee, J. H., Daud, A. N., Cribbs, L. L., Lacerda, A. E., Pereverzev, A., Klockner, U., Schneider, T., and Perez-Reyes, E. 1999a. Cloning and expression of a novel member of the low voltage-activated T-type calcium channel family. J Neurosci 19, 1912-1921.

Lee, J. H., Gomora, J. C., Cribbs, L. L., and Perez-Reyes, E. 1999b. Nickel block of three cloned T-type calcium channels: low concentrations selectively block alpha1H. Biophys J 77, 3034-3042.

Lee, K. S. 1987. Potentiation of the calcium-channel currents of internally perfused mammalian heart cells by repetitive depolarization. Proc Natl Acad Sci U S A 84, 3941-3945.

Lee, K. S., Marban, E., and Tsien, R. W. 1985. Inactivation of calcium channels in mammalian heart cells: joint dependence on membrane potential and intracellular calcium. J Physiol 364, 395-411.

Leroy, J., Lignon, J. M., Gannier, F., Argibay, J. A., and Malecot, C. O. 2002. Caffeine-induced immobilization of gating charges in isolated guinea-pig ventricular heart cells. Br J Pharmacol 135, 721-734.

Leroy, J., Pereverzev, A., Vajna, R., Qin, N., Pfitzer, G., Hescheler, J., Malecot, C. O., Schneider, T., and Klockner, U. 2003. Ca2+-sensitive regulation of E-type Ca2+ channel activity depends on an arginine-rich region in the cytosolic II-III loop. Eur J Neurosci 18, 841-855.

Leuranguer, V., Monteil, A., Bourinet, E., Dayanithi, G., and Nargeot, J. 2000. T-type calcium currents in rat cardiomyocytes during postnatal development: contribution to hormone secretion. Am J Physiol Heart Circ Physiol 279, H2540-H2548.

Levi, R. C., Alloatti, G., and Fischmeister, R. 1989. Cyclic GMP regulates the Ca-channel current in guinea pig ventricular myocytes. Pflugers Arch 413, 685-687.

Levi, R. C., Alloatti, G., Penna, C., and Gallo, M. P. 1994. Guanylate-cyclase-mediated inhibition of cardiac ICa by carbachol and sodium nitroprusside. Pflugers Arch 426, 419-426.

Li, D., Melnyk, P., Feng, J., Wang, Z., Petrecca, K., Shrier, A., and Nattel, S. 2000. Effects of experimental heart failure on atrial cellular and ionic electrophysiology. Circulation 101, 2631-2638.

Li, G. R., Yang, B., Feng, J., Bosch, R. F., Carrier, M., and Nattel, S. 1999. Transmembrane ICa contributes to rate-dependent changes of action potentials in human ventricular myocytes. Am J Physiol 276, H98-H106.

Li, J., Stevens, L., Klugbauer, N., and Wray, D. 2004. Roles of molecular regions in determining differences between voltage-dependence of activation of CaV3.1 and CaV1.2 calcium channels. J Biol Chem.

Liang, H., DeMaria, C. D., Erickson, M. G., Mori, M. X., Alseikhan, B. A., and Yue, D. T. 2003. Unified mechanisms of Ca2+ regulation across the Ca2+ channel family. Neuron 39, 951-960.

Liggett, S. B., Tepe, N. M., Lorenz, J. N., Canning, A. M., Jantz, T. D., Mitarai, S., Yatani, A., and Dorn, G. W., II 2000. Early and Delayed Consequences of {beta}2-Adrenergic Receptor Overexpression in Mouse Hearts : Critical Role for Expression Level. Circulation 101, 1707-1714.

Liman, E. R., Hess, P., Weaver, F., and Koren, G. 1991. Voltage-sensing residues in the S4 region of a mammalian K+ channel. Nature 353, 752-756.

Limas, C. J., Goldenberg, I. F., and Limas, C. 1989. Autoantibodies against beta-adrenoceptors in human idiopathic dilated cardiomyopathy. Circ Res 64, 97-103.

Linz, K. W. and Meyer, R. 1998a. Control of L-type calcium current during the action potential of guinea-pig ventricular myocytes. J Physiol 513 (Pt 2), 425-442.

Linz, K. W. and Meyer, R. 1998b. The late component of L-type calcium current during guinea-pig cardiac action potentials and its contribution to contraction. Pflugers Arch 436, 679-688.

Linz, K. W. and Meyer, R. 2000. Profile and kinetics of L-type calcium current during the cardiac ventricular action potential compared in guinea-pigs, rats and rabbits. Pflugers Arch 439, 588-599.

Lipkind, G. M. and Fozzard, H. A. 2001. Modeling of the outer vestibule and selectivity filter of the L-type Ca2+ channel. Biochemistry 40, 6786-6794.

Lipkind, G. M. and Fozzard, H. A. 2003. Molecular modeling of interactions of dihydropyridines and phenylalkylamines with the inner pore of the L-type Ca2+ channel. Mol Pharmacol 63, 499-511.

Lipscombe, D. 2002. L-Type Calcium Channels: Highs and New Lows. Circ Res 90, 933-935.

Lipsius, S. L., Huser, J., and Blatter, L. A. 2001. Intracellular Ca2+ release sparks atrial pacemaker activity. News Physiol Sci 16, 101-106.

Liu, Y., Zeng, W., Delmar, M., and Jalife, J. 1993. Ionic mechanisms of electronic inhibition and concealed conduction in rabbit atrioventricular nodal myocytes. Circulation 88, 1634-1646.

Lopez, G. A., Jan, Y. N., and Jan, L. Y. 1994. Evidence that the S6 segment of the Shaker voltagegated K+ channel comprises part of the pore. Nature 367, 179-182.

Lopez-Lopez, J. R., Shacklock, P. S., Balke, C. W., and Wier, W. G. 1995. Local calcium transients triggered by single L-type calcium channel currents in cardiac cells. Science 268, 1042-1045.

Lorenzon, N. M. and Beam, K. G. 2000. Calcium channelopathies. Kidney Int 57, 794-802.

Lu, Z. J., Pereverzev, A., Liu, H. L., WeiergrΣber, M., Henry, M., Krieger, A., Smyth, N., Hescheler, J., and Schneider, T. 2004. Arrhythmia in Isolated Prenatal Hearts after Ablation of the Ca v 2.3 (a1E) Subunit of Voltage-gated Ca 2+ Channels. Cellular Physiology and Biochemistry 14, 11-22.

Magnusson, Y., Wallukat, G., Waagstein, F., Hjalmarson, A., and Hoebeke, J. 1994. Autoimmunity in idiopathic dilated cardiomyopathy. Characterization of antibodies against the beta 1-adrenoceptor with positive chronotropic effect. Circulation 89, 2760-2767.

Malkiel, S., Kuan, A. P., and Diamond, B. 1996. Autoimmunity in heart disease: mechanisms and genetic susceptibility. Molecular Medicine Today 2, 336-342.

Mangoni, M. E., Fontanaud, P., Noble, P. J., Noble, D., Benkemoun, H., Nargeot, J., and Richard, S. 2000. Facilitation of the L-type calcium current in rabbit sino-atrial cells: effect on cardiac automaticity. Cardiovasc Res 48, 375-392.

Mangoni, M. E., Couette, B., Bourinet, E., Platzer, J., Reimer, D., Striessnig, J., and Nargeot, J. 2003. Functional role of L-type Cav1.3 Ca2+ channels in cardiac pacemaker activity. PNAS 100, 5543-5548.

Marban, E. 2002. Cardiac channelopathies. Nature 415, 213-218.

Marban, E., Robinson, S. W., and Wier, W. G. 1986. Mechanisms of arrhythmogenic delayed and early afterdepolarizations in ferret ventricular muscle. J Clin Invest 78, 1185-1192.

Marban, E. and Tsien, R. W. 1982. Enhancement of calcium current during digitalis inotropy in mammalian heart: positive feed-back regulation by intracellular calcium? J Physiol 329, 589-614.

Marks, A. R. 2001. Ryanodine receptors/calcium release channels in heart failure and sudden cardiac death. J Mol Cell Cardiol 33, 615-624.

Marksteiner, R., Schurr, P., Berjukow, S., Margreiter, E., Perez-Reyes, E., and Hering, S. 2001. Inactivation determinants in segment IIIS6 of Ca(v)3.1. J Physiol 537, 27-34.

Martinez, M. L., Heredia, M. P., and Delgado, C. 1999. Expression of T-type Ca(2+) channels in ventricular cells from hypertrophied rat hearts. J Mol Cell Cardiol 31, 1617-1625.

Masaki, H., Sato, Y., Luo, W., Kranias, E. G., and Yatani, A. 1997. Phospholamban deficiency alters inactivation kinetics of L-type Ca2+ channels in mouse ventricular myocytes. Am J Physiol 272, H606-H612.

Matsui, S., Fu, M. L. X., Katsuda, S., Hayase, M., Yamaguchi, N., Teraoka, K., Kurihara, T., Takekoshi, N., Murakami, E., Hoebeke, J., and Hjalmarson, A. 1997. Peptides Derived from Cardiovascular G-protein-coupled Receptors Induce Morphological Cardiomyopathic Changes in Immunized Rabbits. Journal of Molecular and Cellular Cardiology 29, 641-655.

Maurice, D. H., Palmer, D., Tilley, D. G., Dunkerley, H. A., Netherton, S. J., Raymond, D. R., Elbatarny, H. S., and Jimmo, S. L. 2003. Cyclic Nucleotide Phosphodiesterase Activity, Expression, and Targeting in Cells of the Cardiovascular System. Mol Pharmacol 64, 533-546.

McDonald, T. F., Pelzer, S., Trautwein, W., and Pelzer, D. J. 1994. Regulation and modulation of calcium channels in cardiac, skeletal, and smooth muscle cells. Physiol Rev 74, 365-507.

McGee, A. W., Nunziato, D. A., Maltez, J. M., Prehoda, K. E., Pitt, G. S., and Bredt, D. S. 2004. Calcium channel function regulated by the SH3-GK module in beta subunits. Neuron 42, 89-99.

McHugh, D., Sharp, E. M., Scheuer, T., and Catterall, W. A. 2000. Inhibition of cardiac L-type calcium channels by protein kinase C phosphorylation of two sites in the N-terminal domain. PNAS 97, 12334-12338.

McMorn, S. O., Harrison, S. M., Zang, W. J., and Boyett, M. R. 1996. Comparison of ultra-slow, voltage-dependent inactivation of the cardiac L-type Ca2+ channel with Ca2+ or Ba2+ as the charge carrier in ferret ventricular myocytes. Exp Physiol 81, 565-575.

McPherson, P. S. 1999. Regulatory role of SH3 domain-mediated protein-protein interactions in synaptic vesicle endocytosis. Cell Signal 11, 229-238.

Mentrard, D., Vassort, G., and Fischmeister, R. 1984. Calcium-mediated inactivation of the calcium conductance in cesium- loaded frog heart cells. J Gen Physiol 83, 105-131.

Mery, P. F., Brechler, V., Pavoine, C., Pecker, F., and Fischmeister, R. 1990. Glucagon stimulates the cardiac Ca2+ current by activation of adenylyl cyclase and inhibition of phosphodiesterase. Nature 345, 158-161.

Mery, P. F., Lohmann, S. M., Walter, U., and Fischmeister, R. 1991. Ca2+ current is regulated by cyclic GMP-dependent protein kinase in mammalian cardiac myocytes. Proc Natl Acad Sci U S A 88, 1197-1201.

Mery, P. F., Pavoine, C., Belhassen, L., Pecker, F., and Fischmeister, R. 1993. Nitric oxide regulates cardiac Ca2+ current. Involvement of cGMP- inhibited and cGMP-stimulated phosphodiesterases through guanylyl cyclase activation. J Biol Chem 268, 26286-26295.

Mewes, T. and Ravens, U. 1994. L-type calcium currents of human myocytes from ventricle of non-failing and failing hearts and from atrium. J Mol Cell Cardiol 26, 1307-1320.

Mijares, A., Verdot, L., Peineau, N., Vray, B., Hoebeke, J., and Argibay, J. 1996a. Antibodies from Trypanosoma cruzi infected mice recognize the second extracellular loop of the beta 1-adrenergic and M2-muscarinic receptors and regulate calcium channels in isolated cardiomyocytes. Mol Cell Biochem 163-164, 107-112.

Mijares, A., Lebesgue, D., Argibay, J., and Hoebeke, J. 1996b. Anti-peptide antibodies sensitive to the 'active' state of the [beta]2-adrenergic receptor. FEBS Letters 399, 188-191.

Mikami, A., Imoto, K., Tanabe, T., Niidome, T., Mori, Y., Takeshima, H., Narumiya, S., and Numa, S. 1989. Primary structure and functional expression of the cardiac dihydropyridine-sensitive calcium channel. Nature 340, 230-233.

Ming, Z., Nordin, C., and Aronson, R. S. 1994. Role of L-type calcium channel window current in generating current-induced early afterdepolarizations. J Cardiovasc Electrophysiol 5, 323-334.

Mitchell, J. W., Larsen, J. K., and Best, P. M. 2002. Identification of the calcium channel alpha 1E (Ca(v)2.3) isoform expressed in atrial myocytes. Biochim Biophys Acta 1577, 17-26.

Mitra, R. and Morad, M. 1986. Two types of calcium channels in guinea pig ventricular myocytes. Proc Natl Acad Sci U S A 83, 5340-5344.

Mitterdorfer, J., Froschmayr, M., Grabner, M., Moebius, F. F., Glossmann, H., and Striessnig, J. 1996. Identification of PK-A phosphorylation sites in the carboxyl terminus of L-type calcium channel alpha 1 subunits. Biochemistry 35, 9400-9406.

Mittman, S., Guo, J., and Agnew, W. S. 1999a. Structure and alternative splicing of the gene encoding alpha1G, a human brain T calcium channel alpha1 subunit. Neurosci Lett 274, 143-146.

Mittman, S., Guo, J., Emerick, M. C., and Agnew, W. S. 1999b. Structure and alternative splicing of the gene encoding alpha1I, a human brain T calcium channel alpha1 subunit. Neurosci Lett 269, 121-124.

Molkentin, J. D., Lu, J. R., Antos, C. L., Markham, B., Richardson, J., Robbins, J., Grant, S. R., and Olson, E. N. 1998. A calcineurin-dependent transcriptional pathway for cardiac hypertrophy. Cell 93, 215-228.

Monteil, A., Chemin, J., Bourinet, E., Mennessier, G., Lory, P., and Nargeot, J. 2000a. Molecular and functional properties of the human alpha(1G) subunit that forms T-type calcium channels. J Biol Chem 275, 6090-6100.

Monteil, A., Chemin, J., Leuranguer, V., Altier, C., Mennessier, G., Bourinet, E., Lory, P., and Nargeot, J. 2000b. Specific properties of T-type calcium channels generated by the human alpha 11 subunit. J Biol Chem 275, 16530-16535.

Morad, M., Davies, N. W., Ulrich, G., and Schultheiss, H. P. 1988. Antibodies against ADP-ATP carrier enhance Ca2+ current in isolated cardiac myocytes. Am J Physiol Heart Circ Physiol 255, H960-H964.

Mori, M. X., Erickson, M. G., and Yue, D. T. 2004. Functional stoichiometry and local enrichment of calmodulin interacting with Ca2+ channels. Science 304, 432-435.

Moss, F. J., Viard, P., Davies, A., Bertaso, F., Page, K. M., Graham, A., Canti, C., Plumpton, M., Plumpton, C., Clare, J. J., and Dolphin, A. C. 2002. The novel product of a five-exon stargazin-related gene abolishes Ca(V)2.2 calcium channel expression. EMBO J 21, 1514-1523.

Mouton, J., Feltz, A., and Maulet, Y. 2001a. Interactions of calmodulin with two peptides derived from the c-terminal cytoplasmic domain of the Ca(v)1.2 Ca2+ channel provide evidence for a molecular switch involved in Ca2+-induced inactivation. J Biol Chem 276, 22359-22367.

Mouton, J., Ronjat, M., Jona, I., Villaz, M., Feltz, A., and Maulet, Y. 2001b. Skeletal and cardiac ryanodine receptors bind to the Ca(2+)-sensor region of dihydropyridine receptor alpha(1C) subunit. FEBS Lett 505, 441-444.

Munk, A. A., Adjemian, R. A., Zhao, J., Ogbaghebriel, A., and Shrier, A. 1996. Electrophysiological properties of morphologically distinct cells isolated from the rabbit atrioventricular node. J Physiol 493 (Pt 3), 801-818.

Murbartian, J., Arias, J. M., Lee, J. H., Gomora, J. C., and Perez-Reyes, E. 2002. Alternative splicing of the rat Ca(v)3.3 T-type calcium channel gene produces variants with distinct functional properties(1). FEBS Lett 528, 272-278.

Murphy, T. H., Worley, P. F., and Baraban, J. M. 1991. L-type voltage-sensitive calcium channels mediate synaptic activation of immediate early genes. Neuron 7, 625-635.

Musa, H., Lei, M., Honjo, H., Jones, S. A., Dobrzynski, H., Lancaster, M. K., Takagishi, Y., Henderson, Z., Kodama, I., and Boyett, M. R. 2002. Heterogeneous expression of Ca(2+) handling proteins in rabbit sinoatrial node. J Histochem Cytochem 50, 311-324.

Muth, J. N., Bodi, I., Lewis, W., Varadi, G., and Schwartz, A. 2001. A Ca2+-Dependent Transgenic Model of Cardiac Hypertrophy : A Role for Protein Kinase C{{alpha}}. Circulation 103, 140-147.

Muth, J. N., Yamaguchi, H., Mikala, G., Grupp, I. L., Lewis, W., Cheng, H., Song, L. S., Lakatta, E. G., Varadi, G., and Schwartz, A. 1999. Cardiac-specific Overexpression of the alpha 1 Subunit of the L-type Voltage-dependent Ca2+ Channel in Transgenic Mice. LOSS OF ISOPROTERENOL-INDUCED CONTRACTION. J Biol Chem 274, 21503-21506.

Nabauer, M. and Morad, M. 1990. Ca2(+)-induced Ca2+ release as examined by photolysis of caged Ca2+ in single ventricular myocytes. Am J Physiol 258, C189-C193.

Nakayama, N., Kirley, T. L., Vaghy, P. L., McKenna, E., and Schwartz, A. 1987. Purification of putative Ca2+ channel protein from rabbit skeletal muscle. Determination of the amino-terminal sequence. J Biol Chem 262, 6572-6576.

Naraghi, M. and Neher, E. 1997. Linearized buffered Ca2+ diffusion in microdomains and its implications for calculation of [Ca2+] at the mouth of a calcium channel. J Neurosci 17, 6961-6973.

Nascimento, J. H. 1996. Electrophysiological properties of L-type Ca2+ currents in isolated adult mouse ventricular myocytes. Braz J Med Biol Res 29, 1397-1405.

Nascimento, J. H., Salle, L., Hoebeke, J., Argibay, J., and Peineau, N. 2001. cGMP-mediated inhibition of cardiac L-type Ca(2+) current by a monoclonal antibody against the M(2) ACh receptor. Am J Physiol Cell Physiol 281, C1251-C1258.

Nattel, S. 2002. New ideas about atrial fibrillation 50 years on. Nature 415, 219-226.

Neely, A., Garcia-Olivares, J., Voswinkel, S., Horstkott, H., and Hidalgo, P. 2004. Folding of active calcium channel beta(1b) -subunit by size-exclusion chromatography and its role on channel function. J Biol Chem 279, 21689-21694.

Neely, A., Olcese, R., Baldelli, P., Wei, X., Birnbaumer, L., and Stefani, E. 1995. Dual activation of the cardiac Ca2+ channel alpha 1C-subunit and its modulation by the beta-subunit. Am J Physiol 268, C732-C740.

Neely, A., Wei, X., Olcese, R., Birnbaumer, L., and Stefani, E. 1993. Potentiation by the beta subunit of the ratio of the ionic current to the charge movement in the cardiac calcium channel. Science 262, 575-578.

Neher, E. 1998. Vesicle pools and Ca2+ microdomains: new tools for understanding their roles in neurotransmitter release. Neuron 20, 389-399.

Newcomb, R., Szoke, B., Palma, A., Wang, G., Chen, X., Hopkins, W., Cong, R., Miller, J., Urge, L., Tarczy-Hornoch, K., Loo, J. A., Dooley, D. J., Nadasdi, L., Tsien, R. W., Lemos, J., and Miljanich, G. 1998. Selective peptide antagonist of the class E calcium channel from the venom of the tarantula Hysterocrates gigas. Biochemistry 37, 15353-15362.

Niggli, E. and Lederer, W. J. 1990. Voltage-independent calcium release in heart muscle. Science 250, 565-568.

Niwa, N., Yasui, K., Opthof, T., Takemura, H., Shimizu, A., Horiba, M., Lee, J. K., Honjo, H., Kamiya, K., and Kodama, I. 2004. Cav3.2 subunit underlies the functional T-type Ca2+ channel in murine hearts during the embryonic period. Am J Physiol Heart Circ Physiol 286, H2257-H2263.

Noble, S. and Shimoni, Y. 1981. The calcium and frequency dependence of the slow inward current 'staircase' in frog atrium. J Physiol 310, 57-75.

Noceti, F., Baldelli, P., Wei, X., Qin, N., Toro, L., Birnbaumer, L., and Stefani, E. 1996. Effective gating charges per channel in voltage-dependent K+ and Ca2+ channels. J Gen Physiol 108, 143-155.

Noma, A., Irisawa, H., Kokobun, S., Kotake, H., Nishimura, M., and Watanabe, Y. 1980. Slow current systems in the A-V node of the rabbit heart. Nature 285, 228-229.

Nourry, C., Grant, S. G., and Borg, J. P. 2003. PDZ domain proteins: plug and play! Sci STKE 2003, RE7.

Nowycky, M. C., Fox, A. P., and Tsien, R. W. 1985. Three types of neuronal calcium channel with different calcium agonist sensitivity. Nature 316, 440-443.

Nuss, H. B. and Houser, S. R. 1993. T-type Ca2+ current is expressed in hypertrophied adult feline left ventricular myocytes. Circ Res 73, 777-782.

O'Rourke, B., Kass, D. A., Tomaselli, G. F., Kaab, S., Tunin, R., and Marban, E. 1999. Mechanisms of altered excitation-contraction coupling in canine tachycardia-induced heart failure, I: experimental studies. Circ Res 84, 562-570.

Ohler, A., Houser, S. R., Tomaselli, G. F., and Rourke, B. 2002. Transverse tubules are unchanged in myocytes from failing human heart. Biophys J 82, a590. Abstract.

Olcese, R., Neely, A., Qin, N., Wei, X., Birnbaumer, L., and Stefani, E. 1996. Coupling between charge movement and pore opening in vertebrate neuronal alpha 1E calcium channels. J Physiol 497 (Pt 3), 675-686.

Ono, K. and Trautwein, W. 1991. Potentiation by cyclic GMP of beta-adrenergic effect on Ca2+ current in guinea-pig ventricular cells. J Physiol 443, 387-404.

Opatowsky, Y., Chen, C. C., Campbell, K. P., and Hirsch, J. A. 2004. Structural analysis of the voltage-dependent calcium channel beta subunit functional core and its complex with the alpha1 interaction domain. Neuron 42, 387-399.

Ostrom, R. S., Gregorian, C., Drenan, R. M., Xiang, Y., Regan, J. W., and Insel, P. A. 2001. Receptor number and caveolar co-localization determine receptor coupling efficiency to adenylyl cyclase. J Biol Chem 276, 42063-42069.

Ouadid, H., Seguin, J., Dumuis, A., Bockaert, J., and Nargeot, J. 1992. Serotonin increases calcium current in human atrial myocytes via the newly described 5-hydroxytryptamine4 receptors. Mol Pharmacol 41, 346-351.

Ouadid, H., Seguin, J., Richard, S., Chaptal, P. A., and Nargeot, J. 1991. Properties and Modulation of Ca channels in adult human atrial cells. J Mol Cell Cardiol 23, 41-54.

Page, E. 1978. Quantitative ultrastructural analysis in cardiac membrane physiology. Am J Physiol 235, C147-C158.

91

Papazian, D. M., Timpe, L. C., Jan, Y. N., and Jan, L. Y. 1991. Alteration of voltage-dependence of Shaker potassium channel by mutations in the S4 sequence. Nature 349, 305-310.

Parent, L., Schneider, T., Moore, C. P., and Talwar, D. 1997. Subunit regulation of the human brain alpha 1E calcium channel. J Membr Biol 160, 127-140.

Park, J. Y., Jeong, S. W., Perez-Reyes, E., and Lee, J. H. 2003. Modulation of Ca(v)3.2 T-type Ca2+ channels by protein kinase C. FEBS Lett 547, 37-42.

Park, J. Y., Kang, H. W., Jeong, S. W., and Lee, J. H. 2004. Multiple structural elements contribute to the slow kinetics of the Cav3.3 T-type channel. J Biol Chem 279, 21707-21713.

Parker, I., Zang, W. J., and Wier, W. G. 1996. Ca2+ sparks involving multiple Ca2+ release sites along Z-lines in rat heart cells. J Physiol 497, 31-38.

Pascarel, C., Brette, F., and Le Guennec, J. Y. 2001. Enhancement of the T-type calcium current by hyposmotic shock in isolated guinea-pig ventricular myocytes. J Mol Cell Cardiol 33, 1363-1369.

Patberg, K. W., Plotnikov, A. N., Quamina, A., Gainullin, R. Z., Rybin, A., Danilo, P., Jr., Sun, L. S., and Rosen, M. R. 2003. Cardiac memory is associated with decreased levels of the transcriptional factor CREB modulated by angiotensin II and calcium. Circ Res 93, 472-478.

Patberg, K. W. and Rosen, M. R. 2004. Molecular determinants of cardiac memory and their regulation. J Mol Cell Cardiol 36, 195-204.

Pate, P., Mochca-Morales, J., Wu, Y., Zhang, J. Z., Rodney, G. G., Serysheva, I. I., Williams, B. Y., Anderson, M. E., and Hamilton, S. L. 2000. Determinants for Calmodulin Binding on Voltage-dependent Ca2+ Channels. J Biol Chem 275, 39786-39792.

Pawson, T. and Scott, J. D. 1997. Signaling through scaffold, anchoring, and adaptor proteins. Science 278, 2075-2080.

Perets, T., Blumenstein, Y., Shistik, E., Lotan, I., and Dascal, N. 1996. A potential site of functional modulation by protein kinase A in the cardiac Ca2+ channel alpha 1C subunit. FEBS Lett 384, 189-192.

Pereverzev, A., Leroy, J., Krieger, A., Malecot, C. O., Hescheler, J., Pfitzer, G., Klockner, U., and Schneider, T. 2002. Alternate splicing in the cytosolic II-III loop and the carboxy terminus of human E-type voltage-gated Ca(2+) channels: electrophysiological characterization of isoforms. Mol Cell Neurosci 21, 352-365.

Perez-Reyes, E. 1998. Molecular characterization of a novel family of low voltage-activated, T-type, calcium channels. J Bioenerg Biomembr 30, 313-318.

Perez-Reyes, E., Castellano, A., Kim, H. S., Bertrand, P., Baggstrom, E., Lacerda, A. E., Wei, X. Y., and Birnbaumer, L. 1992. Cloning and expression of a cardiac/brain beta subunit of the L-type calcium channel. J Biol Chem 267, 1792-1797.

Perez-Reyes, E., Cribbs, L. L., Daud, A., Lacerda, A. E., Barclay, J., Williamson, M. P., Fox, M., Rees, M., and Lee, J. H. 1998. Molecular characterization of a neuronal low-voltage-activated T-type calcium channel. Nature 391, 896-900.

Perez-Reyes, E., Wei, X. Y., Castellano, A., and Birnbaumer, L. 1990. Molecular diversity of Ltype calcium channels. Evidence for alternative splicing of the transcripts of three non-allelic genes. J Biol Chem 265, 20430-20436.

Perez-Reyes, E., Yuan, W., Wei, X., and Bers, D. M. 1994. Regulation of the cloned L-type cardiac calcium channel by cyclic-AMP-dependent protein kinase. FEBS Lett 342, 119-123.

Perez-Reyes, E. 2003. Molecular Physiology of Low-Voltage-Activated T-type Calcium Channels. Physiol Rev 83, 117-161.

Peterson, B. Z., Lee, J. S., Mulle, J. G., Wang, Y., de Leon, M., and Yue, D. T. 2000. Critical determinants of Ca(2+)-dependent inactivation within an EF- hand motif of L-type Ca(2+) channels. Biophys J 78, 1906-1920.

Pfeifer, A., Nurnberg, B., Kamm, S., Uhde, M., Schultz, G., Ruth, P., and Hofmann, F. 1995. Cyclic GMP-dependent protein kinase blocks pertussis toxin-sensitive hormone receptor signaling pathways in Chinese hamster ovary cells. J Biol Chem 270, 9052-9059.

Piacentino, V., III, Dipla, K., Gaughan, J. P., and Houser, S. R. 2000. Voltage-dependent Ca2+ release from the SR of feline ventricular myocytes is explained by Ca2+-induced Ca2+ release. J Physiol 523 Pt 3, 533-548.

Piacentino, V., III, Weber, C. R., Chen, X., Weisser-Thomas, J., Margulies, K. B., Bers, D. M., and Houser, S. R. 2003. Cellular basis of abnormal calcium transients of failing human ventricular myocytes. Circ Res 92, 651-658.

Pietrobon, D. and Hess, P. 1990. Novel mechanism of voltage-dependent gating in L-type calcium channels. Nature 346, 651-655.

Pinto, J. M., Sosunov, E. A., Gainullin, R. Z., Rosen, M. R., and Boyden, P. A. 1999. Effects of mibefradil, a T-type calcium current antagonist, on electrophysiology of Purkinje fibers that survived in the infarcted canine heart. J Cardiovasc Electrophysiol 10, 1224-1235.

Piot, C., Lemaire, S., Albat, B., Seguin, J., Nargeot, J., and Richard, S. 1996. High frequencyinduced upregulation of human cardiac calcium currents. Circulation 93, 120-128.

Pitt, G. S., Zuhlke, R. D., Hudmon, A., Schulman, H., Reuter, H., and Tsien, R. W. 2001. Molecular basis of calmodulin tethering and Ca2+-dependent inactivation of L-type Ca2+ channels. J Biol Chem 276, 30794-30802.

Platzer, J., Engel, J., Schrott-Fischer, A., Stephan, K., Bova, S., Chen, H., Zheng, H., and Striessnig, J. 2000. Congenital deafness and sinoatrial node dysfunction in mice lacking class D L-type Ca2+ channels. Cell 102, 89-97.

Plotnikov, A. N., Yu, H., Geller, J. C., Gainullin, R. Z., Chandra, P., Patberg, K. W., Friezema, S., Danilo, P., Jr., Cohen, I. S., Feinmark, S. J., and Rosen, M. R. 2003. Role of L-type calcium channels in pacing-induced short-term and long-term cardiac memory in canine heart. Circulation 107, 2844-2849.

Pogwizd, S. M., Qi, M., Yuan, W., Samarel, A. M., and Bers, D. M. 1999. Upregulation of Na(+)/Ca(2+) exchanger expression and function in an arrhythmogenic rabbit model of heart failure. Circ Res 85, 1009-1019.

Pogwizd, S. M., Schlotthauer, K., Li, L., Yuan, W., and Bers, D. M. 2001. Arrhythmogenesis and contractile dysfunction in heart failure: Roles of sodium-calcium exchange, inward rectifier potassium current, and residual beta-adrenergic responsiveness. Circ Res 88, 1159-1167.

Pragnell, M., De Waard, M., Mori, Y., Tanabe, T., Snutch, T. P., and Campbell, K. P. 1994. Calcium channel beta-subunit binds to a conserved motif in the I-II cytoplasmic linker of the alpha 1-subunit. Nature 368, 67-70.

Publicover, S. J., Preston, M. R., and El Haj, A. J. 1995. Voltage-dependent potentiation of low-voltage-activated Ca2+ channel currents in cultured rat bone marrow cells. J Physiol 489 (Pt 3), 649-661.

Puglisi, J. L., Yuan, W., Bassani, J. W., and Bers, D. M. 1999. Ca(2+) influx through Ca(2+) channels in rabbit ventricular myocytes during action potential clamp: influence of temperature. Circ Res 85, e7-e16.

Puri, T. S., Gerhardstein, B. L., Zhao, X. L., Ladner, M. B., and Hosey, M. M. 1997. Differential effects of subunit interactions on protein kinase A- and C-mediated phosphorylation of L-type calcium channels. Biochemistry 36, 9605-9615.

Qiao, X. and Meng, H. 2003. Nonchannel functions of the calcium channel gamma subunit: insight from research on the stargazer mutant. J Bioenerg Biomembr 35, 661-670.

Qin, N., Olcese, R., Stefani, E., and Birnbaumer, L. 1998a. Modulation of human neuronal alpha 1E-type calcium channel by alpha 2delta -subunit. Am J Physiol Cell Physiol 274, C1324-C1331.

Qin, N., Platano, D., Olcese, R., Costantin, J. L., Stefani, E., and Birnbaumer, L. 1998b. Unique regulatory properties of the type 2a Ca2+ channel beta ásubunit caused by palmitoylation. PNAS 95, 4690-4695.

Qin, N., Yagel, S., Momplaisir, M. L., Codd, E. E., and D'Andrea, M. R. 2002. Molecular Cloning and Characterization of the Human Voltage-Gated Calcium Channel alpha 2delta -4 Subunit. Mol Pharmacol 62, 485-496.

Qu, Y., Xiao, G. Q., Chen, L., and Boutjdir, M. 2001. Autoantibodies from mothers of children with congenital heart block downregulate cardiac L-type Ca channels. J Mol Cell Cardiol 33, 1153-1163.

Quinn, F. R., Currie, S., Duncan, A. M., Miller, S., Sayeed, R., Cobbe, S. M., and Smith, G. L. 2003. Myocardial infarction causes increased expression but decreased activity of the myocardial Na+-Ca2+ exchanger in the rabbit. J Physiol 553, 229-242.

Ramirez, M. T., Zhao, X. L., Schulman, H., and Brown, J. H. 1997. The nuclear deltaB isoform of Ca2+/calmodulin-dependent protein kinase II regulates atrial natriuretic factor gene expression in ventricular myocytes. J Biol Chem 272, 31203-31208.

Razani, B., Woodman, S. E., and Lisanti, M. P. 2002. Caveolae: From Cell Biology to Animal Physiology. Pharmacol Rev 54, 431-467.

Restituito, S., Cens, T., Barrere, C., Geib, S., Galas, S., De Waard, M., and Charnet, P. 2000. The [beta]2a subunit is a molecular groom for the Ca2+ channel inactivation gate. J Neurosci 20, 9046-9052.

Reuter, H. 1967. The dependence of slow inward current in Purkinje fibres on the extracellular calcium-concentration. J Physiol 192, 479-492.

Rich, T. C., Fagan, K. A., Tse, T. E., Schaack, J., Cooper, D. M., and Karpen, J. W. 2001. A uniform extracellular stimulus triggers distinct cAMP signals in different compartments of a simple cell. Proc Natl Acad Sci U S A 98, 13049-13054.

Richard, S., Charnet, P., and Nerbonne, J. M. 1993. Interconversion between distinct gating pathways of the high threshold calcium channel in rat ventricular myocytes. J Physiol (Lond) 462, 197-228.

Richard, S., Tiaho, F., Charnet, P., Nargeot, J., and Nerbonne, J. M. 1990. Two pathways for Ca2+ channel gating differentially modulated by physiological stimuli. Am J Physiol 258, H1872-H1881.

Richardson, P., McKenna, W., Bristow, M., Maisch, B., Mautner, B., O'Connell, J., Olsen, E., Thiene, G., Goodwin, J., Gyarfas, I., Martin, I., and Nordet, P. 1996. Report of the 1995 World Health Organization/International Society and Federation of Cardiology Task Force on the Definition and Classification of Cardiomyopathies. Circulation 93, 841-842.

Rigg, L. and Terrar, D. A. 1996. Possible role of calcium release from the sarcoplasmic reticulum in pacemaking in guinea-pig sino-atrial node. Exp Physiol 81, 877-880.

Ringer, S. 1883. A further contribution regarding the influence of the different constituents of the blood on the contraction of the heart. J Physiol 4, 29-42.

Rivet-Bastide, M., Vandecasteele, G., Hatem, S., Verde, I., Benardeau, A., Mercadier, J. J., and Fischmeister, R. 1997. cGMP-stimulated Cyclic Nucleotide Phosphodiesterase Regulates the Basal Calcium Current in Human Atrial Myocytes. J Clin Invest 99, 2710-2718.

Romanin, C., Gamsjaeger, R., Kahr, H., Schaufler, D., Carlson, O., Abernethy, D. R., and Soldatov, N. M. 2000. Ca(2+) sensors of L-type Ca(2+) channel. FEBS Lett 487, 301-306.

Romanin, C., Seydl, K., Glossmann, H., and Schindler, H. 1992. The dihydropyridine niguldipine inhibits T-type Ca2+ currents in atrial myocytes. Pflugers Arch 420, 410-412.

Rosenbaum, M. B., Blanco, H. H., Elizari, M. V., Lazzari, J. O., and Davidenko, J. M. 1982. Electrotonic modulation of the T wave and cardiac memory. Am J Cardiol 50, 213-222.

Rougier, O., Vassort, G., Garnier, D., Gargouil, Y. M., and Coraboeuf, E. 1969. Existence and role of a slow inward current during the frog atrial action potential. Pflugers Arch 308, 91-110.

Ruth, P., Rohrkasten, A., Biel, M., Bosse, E., Regulla, S., Meyer, H. E., Flockerzi, V., and Hofmann, F. 1989. Primary structure of the beta subunit of the DHP-sensitive calcium channel from skeletal muscle. Science 245, 1115-1118.

Rybin, V. O., Xu, X., Lisanti, M. P., and Steinberg, S. F. 2000. Differential targeting of beta - adrenergic receptor subtypes and adenylyl cyclase to cardiomyocyte caveolae. A mechanism to functionally regulate the cAMP signaling pathway. J Biol Chem 275, 41447-41457.

Safa, P., Boulter, J., and Hales, T. G. 2001. Functional properties of Cav1.3 (alpha1D) L-type Ca2+ channel splice variants expressed by rat brain and neuroendocrine GH3 cells. J Biol Chem 276, 38727-38737.

Sakai, R., Shen, J. B., and Pappano, A. J. 1999. Elevated cAMP suppresses muscarinic inhibition of L-type calcium current in guinea pig ventricular myocytes. J Cardiovasc Pharmacol 34, 304-315.

Sakmann, B. and Neher, E. (1995) Single-channel recording. 2nd ed. Plenum Press, New York.

Salle, L., Eftekhari, P., Aupart, M., Cosnay, P., Hoebeke, J., and Argibay, J. A. 2001. Inhibitory activity of antibodies against the human atrial 5-HT(4)receptor. J Mol Cell Cardiol 33, 405-417.

Santana, L. F., Chase, E. G., Votaw, V. S., Nelson, M. T., and Greven, R. 2002. Functional coupling of calcineurin and protein kinase A in mouse ventricular myocytes. J Physiol 544, 57-69.

Satoh, H. 1995. Role of T-type Ca2+ channel inhibitors in the pacemaker depolarization in rabbit sino-atrial nodal cells. Gen Pharmacol 26, 581-587.

Schneider, T., Wei, X., Olcese, R., Costantin, J. L., Neely, A., Palade, P., Perez-Reyes, E., Qin, N., Zhou, J., Crawford, G. D., and . 1994. Molecular analysis and functional expression of the human type E neuronal Ca2+ channel alpha 1 subunit. Receptors Channels 2, 255-270.

Scholze, A., Plant, T. D., Dolphin, A. C., and Nurnberg, B. 2001. Functional expression and characterization of a voltage-gated CaV1.3 (alpha1D) calcium channel subunit from an insulin-secreting cell line. Mol Endocrinol 15, 1211-1221.

Schram, G., Pourrier, M., Melnyk, P., and Nattel, S. 2002. Differential distribution of cardiac ion channel expression as a basis for regional specialization in electrical function. Circ Res 90, 939-950.

Schroder, F., Handrock, R., Beuckelmann, D. J., Hirt, S., Hullin, R., Priebe, L., Schwinger, R. H., Weil, J., and Herzig, S. 1998. Increased availability and open probability of single L-type calcium channels from failing compared with nonfailing human ventricle. Circulation 98, 969-976.

Schroder, F., Klein, G., Fiedler, B., Bastein, M., Schnasse, N., Hillmer, A., Ames, S., Gambaryan, S., Drexler, H., and Walter, U. 2003. Single L-type Ca2+ channel regulation by cGMP-dependent protein kinase type I in adult cardiomyocytes from PKG I transgenic mice. Cardiovascular Research 60, 268-277.

Schultheiss, H. P., Kuhl, U., Janda, I., Melzner, B., Ulrich, G., and Morad, M. 1988. Antibodymediated enhancement of calcium permeability in cardiac myocytes. J Exp Med 168, 2105-2119.

Schultz, D., Mikala, G., Yatani, A., Engle, D. B., Iles, D. E., Segers, B., Sinke, R. J., Weghuis, D. O., Klockner, U., Wakamori, M., and . 1993. Cloning, chromosomal localization, and functional expression of the alpha 1 subunit of the L-type voltage-dependent calcium channel from normal human heart. Proc Natl Acad Sci U S A 90, 6228-6232.

Scriven, D. R., Dan, P., and Moore, E. D. 2000. Distribution of proteins implicated in excitationcontraction coupling in rat ventricular myocytes. Biophys J 79, 2682-2691.

Sculptoreanu, A., Rotman, E., Takahashi, M., Scheuer, T., and Catterall, W. A. 1993a. Voltagedependent potentiation of the activity of cardiac L-type calcium channel alpha 1 subunits due to phosphorylation by cAMP-dependent protein kinase. Proc Natl Acad Sci U S A 90, 10135-10139.

Sculptoreanu, A., Scheuer, T., and Catterall, W. A. 1993b. Voltage-dependent potentiation of L-type Ca2+ channels due to phosphorylation by cAMP-dependent protein kinase. Nature 364, 240-243.

Sears, C. E., Bryant, S. M., Ashley, E. A., Lygate, C. A., Rakovic, S., Wallis, H. L., Neubauer, S., Terrar, D. A., and Casadei, B. 2003. Cardiac neuronal nitric oxide synthase isoform regulates myocardial contraction and calcium handling. Circ Res 92, e52-e59.

Seisenberger, C., Specht, V., Welling, A., Platzer, J., Pfeifer, A., Kuhbandner, S., Striessnig, J., Klugbauer, N., Feil, R., and Hofmann, F. 2000. Functional Embryonic Cardiomyocytes after

Disruption of the L-type alpha 1C (Cav1.2) Calcium Channel Gene in the Mouse. J Biol Chem 275, 39193-39199.

Semsarian, C., Ahmad, I., Giewat, M., Georgakopoulos, D., Schmitt, J. P., McConnell, B. K., Reiken, S., Mende, U., Marks, A. R., Kass, D. A., Seidman, C. E., and Seidman, J. G. 2002. The L-type calcium channel inhibitor diltiazem prevents cardiomyopathy in a mouse model. J Clin Invest 109, 1013-1020.

Sen, L. and Smith, T. W. 1994. T-type Ca2+ channels are abnormal in genetically determined cardiomyopathic hamster hearts. Circ Res 75, 149-155.

Shacklock, P. S., Wier, W. G., and Balke, C. W. 1995. Local Ca2+ transients (Ca2+ sparks) originate at transverse tubules in rat heart cells. J Physiol 487 (Pt 3), 601-608.

Sham, J. S. 1997. Ca2+ release-induced inactivation of Ca2+ current in rat ventricular myocytes: evidence for local Ca2+ signalling. J Physiol (Lond) 500 (Pt 2), 285-295.

Sham, J. S., Cleemann, L., and Morad, M. 1995. Functional coupling of Ca2+ channels and ryanodine receptors in cardiac myocytes. Proc Natl Acad Sci U S A 92, 121-125.

Sham, J. S., Song, L. S., Chen, Y., Deng, L. H., Stern, M. D., Lakatta, E. G., and Cheng, H. 1998. Termination of Ca2+ release by a local inactivation of ryanodine receptors in cardiac myocytes. Proc Natl Acad Sci U S A 95, 15096-15101.

Shaywitz, A. J. and Greenberg, M. E. 1999. CREB: a stimulus-induced transcription factor activated by a diverse array of extracellular signals. Annu Rev Biochem 68, 821-861.

Shen, J. B. and Pappano, A. J. 2002. On the Role of Phosphatase in Regulation of Cardiac L-Type Calcium Current by Cyclic GMP. J Pharmacol Exp Ther 301, 501-506.

Shepherd, N. and McDonough, H. B. 1998. Ionic diffusion in transverse tubules of cardiac ventricular myocytes. Am J Physiol Heart Circ Physiol 275, H852-H860.

Shi, C. and Soldatov, N. M. 2002. Molecular determinants of voltage-dependent slow inactivation of the Ca2+ channel. J Biol Chem 277, 6813-6821.

Shirayama, T. and Pappano, A. J. 1996. Biphasic effects of intrapipette cyclic guanosine monophosphate on L-type calcium current and contraction of guinea pig ventricular myocytes. J Pharmacol Exp Ther 279, 1274-1281.

Shirokov, R. 1999. Interaction between permeant ions and voltage sensor during inactivation of N-type Ca2+ channels. J Physiol 518 (Pt 3), 697-703.

Shirokov, R., Ferreira, G., Yi, J., and Rios, E. 1998. Inactivation of gating currents of L-type calcium channels. Specific role of the alpha 2 delta subunit. J Gen Physiol 111, 807-823.

Shistik, E., Ivanina, T., Blumenstein, Y., and Dascal, N. 1998. Crucial role of N terminus in function of cardiac L-type Ca2+ channel and its modulation by protein kinase C. J Biol Chem 273, 17901-17909.

Shistik, E., Ivanina, T., Puri, T., Hosey, M., and Dascal, N. 1995. Ca2+ current enhancement by alpha 2/delta and beta subunits in Xenopus oocytes: contribution of changes in channel gating and alpha 1 protein level. J Physiol 489 (Pt 1), 55-62.

Singer, D., Biel, M., Lotan, I., Flockerzi, V., Hofmann, F., and Dascal, N. 1991. The roles of the subunits in the function of the calcium channel. Science 253, 1553-1557.

Singer-Lahat, D., Gershon, E., Lotan, I., Hullin, R., Biel, M., Flockerzi, V., Hofmann, F., and Dascal, N. 1992. Modulation of cardiac Ca2+ channels in Xenopus oocytes by protein kinase C. FEBS Lett 306, 113-118.

Singer-Lahat, D., Lotan, I., Biel, M., Flockerzi, V., Hofmann, F., and Dascal, N. 1994. Cardiac calcium channels expressed in Xenopus oocytes are modulated by dephosphorylation but not by cAMP-dependent phosphorylation. Receptors Channels 2, 215-226.

Sinnegger, M. J., Wang, Z., Grabner, M., Hering, S., Striessnig, J., Glossmann, H., and Mitterdorfer, J. 1997. Nine L-type Amino Acid Residues Confer Full 1,4-Dihydropyridine Sensitivity to the Neuronal Calcium Channel alpha 1A Subunit. ROLE OF L-TYPE MET1188. J Biol Chem 272, 27686-27693.

Sipido, K. R., Callewaert, G., and Carmeliet, E. 1995. Inhibition and rapid recovery of Ca2+ current during Ca2+ release from sarcoplasmic reticulum in guinea pig ventricular myocytes. Circ Res 76, 102-109.

Sipido, K. R., Carmeliet, E., and Van De, W. F. 1998a. T-type Ca2+ current as a trigger for Ca2+ release from the sarcoplasmic reticulum in guinea-pig ventricular myocytes [see comments]. J Physiol (Lond) 508 (Pt 2), 439-451.

Sipido, K. R., Stankovicova, T., Flameng, W., Vanhaecke, J., and Verdonck, F. 1998b. Frequency dependence of Ca2+ release from the sarcoplasmic reticulum in human ventricular myocytes from end-stage heart failure. Cardiovasc Res 37, 478-488.

Soldatov, N. M. 1994. Genomic structure of human L-type Ca2+ channel. Genomics 22, 77-87.

Soldatov, N. M. 2003. Ca2+ channel moving tail: link between Ca2+-induced inactivation and Ca2+ signal transduction. Trends Pharmacol Sci 24, 167-171.

Soldatov, N. M., Bouron, A., and Reuter, H. 1995. Different voltage-dependent inhibition by dihydropyridines of human Ca2+ channel splice variants. J Biol Chem 270, 10540-10543.

Soldatov, N. M., Oz, M., O'Brien, K. A., Abernethy, D. R., and Morad, M. 1998. Molecular determinants of L-type Ca2+ channel inactivation. Segment exchange analysis of the carboxyl-terminal cytoplasmic motif encoded by exons 40-42 of the human alpha1C subunit gene. J Biol Chem 273, 957-963.

Soldatov, N. M., Zuhlke, R. D., Bouron, A., and Reuter, H. 1997. Molecular structures involved in L-type calcium channel inactivation. Role of the carboxyl-terminal region encoded by exons 40-42 in alpha1C subunit in the kinetics and Ca2+ dependence of inactivation. J Biol Chem 272, 3560-3566.

Song, L. S., Sham, J. S., Stern, M. D., Lakatta, E. G., and Cheng, H. 1998. Direct measurement of SR release flux by tracking 'Ca2+ spikes' in rat cardiac myocytes. J Physiol 512, 677-691.

Song, L. S., Guia, A., Muth, J. N., Rubio, M., Wang, S. Q., Xiao, R. P., Josephson, I. R., Lakatta, E. G., Schwartz, A., and Cheng, H. 2002. Ca2+ Signaling in Cardiac Myocytes Overexpressing the {alpha}1 Subunit of L-Type Ca2+ Channel. Circ Res 90, 174-181.

Soong, T. W., Stea, A., Hodson, C. D., Dubel, S. J., Vincent, S. R., and Snutch, T. P. 1993. Structure and functional expression of a member of the low voltage-activated calcium channel family. Science 260, 1133-1136.

Staes, M., Talavera, K., Klugbauer, N., Prenen, J., Lacinova, L., Droogmans, G., Hofmann, F., and Nilius, B. 2001. The amino side of the C-terminus determines fast inactivation of the T-type calcium channel alpha1G. J Physiol 530, 35-45.

Sterin-Borda, L., Giordanengo, L., Joensen, L., and Gea, S. 2003. Cruzipain induces autoantibodies against cardiac muscarinic acetylcholine receptors. Functional and pathological implications. Eur J Immunol 33, 2459-2468.

Sterin-Borda, L., Leiros, C. P., Goin, J. C., Cremaschi, G., Genaro, A., Echague, A. V., and Borda, E. 1997. Participation of Nitric Oxide Signaling System in the Cardiac Muscarinic Cholinergic Effect of Human Chagasic IgG. Journal of Molecular and Cellular Cardiology 29, 1851-1865.

Stern, M. D. 1992. Theory of excitation-contraction coupling in cardiac muscle. Biophys J 63, 497-517.

Stotz, S. C., Hamid, J., Spaetgens, R. L., Jarvis, S. E., and Zamponi, G. W. 2000. Fast inactivation of voltage-dependent calcium channels. A hinged-lid mechanism? J Biol Chem 275, 24575-24582.

Stotz, S. C. and Zamponi, G. W. 2001a. Identification of inactivation determinants in the domain IIS6 region of high voltage-activated calcium channels. J Biol Chem 276, 33001-33010.

Stotz, S. C. and Zamponi, G. W. 2001b. Structural determinants of fast inactivation of high voltageactivated Ca(2+) channels. Trends Neurosci 24, 176-181.

Stuhmer, W., Conti, F., Suzuki, H., Wang, X. D., Noda, M., Yahagi, N., Kubo, H., and Numa, S. 1989. Structural parts involved in activation and inactivation of the sodium channel. Nature 339, 597-603.

Sumii, K. and Sperelakis, N. 1995. cGMP-Dependent Protein Kinase Regulation of the L-Type Ca2+ Current in Rat Ventricular Myocytes. Circ Res 77, 803-812.

Sun, H., Leblanc, N., and Nattel, S. 1997. Mechanisms of inactivation of L-type calcium channels in human atrial myocytes. Am J Physiol 272, H1625-H1635.

Sun, X. H., Protasi, F., Takahashi, M., Takeshima, H., Ferguson, D. G., and Franzini-Armstrong, C. 1995. Molecular architecture of membranes involved in excitation-contraction coupling of cardiac muscle. J Cell Biol 129, 659-671.

Swynghedauw, B. 1999. Molecular mechanisms of myocardial remodeling. Physiol Rev 79, 215-262.

Takagishi, Y., Yasui, K., Severs, N. J., and Murata, Y. 2000. Species-specific difference in distribution of voltage-gated L-type Ca2+ channels of cardiac myocytes. Am J Physiol Cell Physiol 279, C1963-C1969.

Takahashi, M., Seagar, M. J., Jones, J. F., Reber, B. F., and Catterall, W. A. 1987. Subunit structure of dihydropyridine-sensitive calcium channels from skeletal muscle. Proc Natl Acad Sci U S A 84, 5478-5482.

Takahashi, S. X., Miriyala, J., and Colecraft, H. M. 2004. Membrane-associated guanylate kinaselike properties of beta-subunits required for modulation of voltage-dependent Ca2+ channels. Proc Natl Acad Sci U S A 101, 7193-7198.

Takahashi, S. X., Mittman, S., and Colecraft, H. M. 2003. Distinctive modulatory effects of five human auxiliary beta2 subunit splice variants on L-type calcium channel gating. Biophys J 84, 3007-3021.

Takamatsu, H., Nagao, T., Ichijo, H., and Adachi-Akahane, S. 2003. L-type Ca2+ channels serve as a sensor of the SR Ca2+ for tuning the efficacy of Ca2+-induced Ca2+ release in rat ventricular myocytes. J Physiol 552, 415-424.

Takimoto, K., Li, D., Nerbonne, J. M., and Levitan, E. S. 1997. Distribution, splicing and glucocorticoid-induced expression of cardiac alpha 1C and alpha 1D voltage-gated Ca2+ channel mRNAs. J Mol Cell Cardiol 29, 3035-3042.

Talavera, K., Janssens, A., Klugbauer, N., Droogmans, G., and Nilius, B. 2003a. Extracellular Ca2+ modulates the effects of protons on gating and conduction properties of the T-type Ca2+ channel alpha1G (CaV3.1). J Gen Physiol 121, 511-528.

Talavera, K., Janssens, A., Klugbauer, N., Droogmans, G., and Nilius, B. 2003b. Pore structure influences gating properties of the T-type Ca2+ channel alpha1G. J Gen Physiol 121, 529-540.

Talavera, K., Staes, M., Janssens, A., Klugbauer, N., Droogmans, G., Hofmann, F., and Nilius, B. 2001. Aspartate residues of the Glu-Glu-Asp-Asp (EEDD) pore locus control selectivity and permeation of the T-type Ca(2+) channel alpha(1G). J Biol Chem 276, 45628-45635.

Tanabe, T., Beam, K. G., Adams, B. A., Niidome, T., and Numa, S. 1990a. Regions of the skeletal muscle dihydropyridine receptor critical for excitation-contraction coupling. Nature 346, 567-569.

Tanabe, T., Mikami, A., Numa, S., and Beam, K. G. 1990b. Cardiac-type excitation-contraction coupling in dysgenic skeletal muscle injected with cardiac dihydropyridine receptor cDNA. Nature 344, 451-453.

Tanabe, T., Takeshima, H., Mikami, A., Flockerzi, V., Takahashi, H., Kangawa, K., Kojima, M., Matsuo, H., Hirose, T., and Numa, S. 1987. Primary structure of the receptor for calcium channel blockers from skeletal muscle. Nature 328, 313-318.

Tang, S., Mikala, G., Bahinski, A., Yatani, A., Varadi, G., and Schwartz, A. 1993. Molecular localization of ion selectivity sites within the pore of a human L-type cardiac calcium channel. J Biol Chem 268, 13026-13029.

Taviaux, S., Williams, M. E., Harpold, M. M., Nargeot, J., and Lory, P. 1997. Assignment of human genes for beta 2 and beta 4 subunits of voltage-dependent Ca2+ channels to chromosomes 10p12 and 2q22-q23. Hum Genet 100, 151-154.

Terracciano, C. M., Hardy, J., Birks, E. J., Khaghani, A., Banner, N. R., and Yacoub, M. H. 2004. Clinical recovery from end-stage heart failure using left-ventricular assist device and pharmacological therapy correlates with increased sarcoplasmic reticulum calcium content but not with regression of cellular hypertrophy. Circulation 109, 2263-2265.

Tiaho, F., Piot, C., Nargeot, J., and Richard, S. 1994. Regulation of the frequency-dependent facilitation of L-type Ca2+ currents in rat ventricular myocytes. J Physiol (Lond) 477 (Pt 2), 237-251.

Tohse, N., Nakaya, H., Takeda, Y., and Kanno, M. 1995. Cyclic GMP-mediated inhibition of L-type Ca2+ channel activity by human natriuretic peptide in rabbit heart cells. Br J Pharmacol 114, 1076-1082.

Tohse, N. and Sperelakis, N. 1991. cGMP inhibits the activity of single calcium channels in embryonic chick heart cells. Circ Res 69, 325-331.

Tomaselli, G. F. and Marban, E. 1999. Electrophysiological remodeling in hypertrophy and heart failure. Cardiovasc Res 42, 270-283.

Towbin, J. A. and Bowles, N. E. 2002. The failing heart. Nature 415, 227-233.

Trafford, A. W., Diaz, M. E., Negretti, N., and Eisner, D. A. 1997. Enhanced Ca2+ current and decreased Ca2+ efflux restore sarcoplasmic reticulum Ca2+ content after depletion. Circ Res 81, 477-484.

Trafford, A. W., Diaz, M. E., O'Neill, S. C., and Eisner, D. A. 2002. Integrative analysis of calcium signalling in cardiac muscle. Front Biosci 7, d843-d852.

Trafford, A. W. and Eisner, D. A. 2003. No role for a voltage sensitive release mechanism in cardiac muscle. J Mol Cell Cardiol 35, 145-151.

Tseng, G. N. 1988. Calcium current restitution in mammalian ventricular myocytes is modulated by intracellular calcium. Circ Res 63, 468-482.

Tseng, G. N. and Boyden, P. A. 1989. Multiple types of Ca2+ currents in single canine Purkinje cells. Circ Res 65, 1735-1750.

Tsien, R. W. 1973. Adrenaline-like effects of intracellular iontophoresis of cyclic AMP in cardiac Purkinje fibres. Nat New Biol 245, 120-122.

Tsien, R. W., Bean, B. P., Hess, P., Lansman, J. B., Nilius, B., and Nowycky, M. C. 1986. Mechanisms of calcium channel modulation by beta-adrenergic agents and dihydropyridine calcium agonists. J Mol Cell Cardiol 18, 691-710.

Tsien, R. W., Hess, P., McCleskey, E. W., and Rosenberg, R. L. 1987. Calcium channels: mechanisms of selectivity, permeation, and block. Annu Rev Biophys Biophys Chem 16, 265-290.

Tytgat, J., Nilius, B., Vereecke, J., and Carmeliet, E. 1988. The T-type Ca channel in guinea-pig ventricular myocytes is insensitive to isoproterenol. Pflugers Arch 411, 704-706.

Valdeolmillos, M., O'Neill, S. C., Smith, G. L., and Eisner, D. A. 1989. Calcium-induced calcium release activates contraction in intact cardiac cells. Pflugers Arch 413, 676-678.

Van Petegem, F., Clark, K. A., Chatelain, F. C., and Minor, D. L. 2004. Structure of a complex between a voltage-gated calcium channel beta-subunit and an alpha-subunit domain. Nature 429, 671-675.

Van Wagoner, D. R., Pond, A. L., Lamorgese, M., Rossie, S. S., McCarthy, P. M., and Nerbonne, J. M. 1999. Atrial L-type Ca2+ currents and human atrial fibrillation. Circ Res 85, 428-436.

Vandecasteele, G., Eschenhagen, T., and Fischmeister, R. 1998. Role of the NO-cGMP pathway in the muscarinic regulation of the L-type Ca2+ current in human atrial myocytes. J Physiol 506 (Pt 3), 653-663.

Vandecasteele, G., Verde, I., Rucker-Martin, C., Donzeau-Gouge, P., and Fischmeister, R. 2001. Cyclic GMP regulation of the L-type Ca2+ channel current in human atrial myocytes. J Physiol (Lond) 533, 329-340.

Verheijck, E. E., van Ginneken, A. C., Wilders, R., and Bouman, L. N. 1999. Contribution of L-type Ca2+ current to electrical activity in sinoatrial nodal myocytes of rabbits. Am J Physiol 276, H1064-H1077.

Verkerk, A. O., Wilders, R., Coronel, R., Ravesloot, J. H., and Verheijck, E. E. 2003. Ionic remodeling of sinoatrial node cells by heart failure. Circulation 108, 760-766.

Vila Petroff, M. G., Egan, J. M., Wang, X., and Sollott, S. J. 2001. Glucagon-like peptide-1 increases cAMP but fails to augment contraction in adult rat cardiac myocytes. Circ Res 89, 445-452.

Vinogradova, T. M., Zhou, Y. Y., Maltsev, V., Lyashkov, A., Stern, M., and Lakatta, E. G. 2004. Rhythmic ryanodine receptor Ca2+ releases during diastolic depolarization of sinoatrial pacemaker cells do not require membrane depolarization. Circ Res 94, 802-809.

Vinogradova, T. M., Zhou, Y. Y., Bogdanov, K. Y., Yang, D., Kuschel, M., Cheng, H., and Xiao, R. P. 2000. Sinoatrial Node Pacemaker Activity Requires Ca2+/Calmodulin-Dependent Protein Kinase II Activation. Circ Res 87, 760-767.

Wahler, G. M. and Dollinger, S. J. 1995. Nitric oxide donor SIN-1 inhibits mammalian cardiac calcium current through cGMP-dependent protein kinase. Am J Physiol 268, C45-C54.

Wakamori, M., Mikala, G., Schwartz, A., and Yatani, A. 1993. Single-channel analysis of a cloned human heart L-type Ca2+ channel alpha 1 subunit and the effects of a cardiac beta subunit. Biochem Biophys Res Commun 196, 1170-1176.

Walker, D. and De Waard, M. 1998. Subunit interaction sites in voltage-dependent Ca2+ channels: role in channel function. Trends Neurosci 21, 148-154.

Wallukat, G., Kayser, A., and Wollenberger, A. 1995. The beta 1-adrenoceptor as antigen: functional aspects. Eur Heart J 16 Suppl O, 85-88.

Wallukat, G. and Wollenberger, A. 1987. Effects of the serum gamma globulin fraction of patients with allergic asthma and dilated cardiomyopathy on chronotropic beta adrenoceptor function in cultured neonatal rat heart myocytes. Biomed Biochim Acta 46, S634-S639.

Wallukat, G., Fu, H. M., Matsui, S., Hjalmarson, A., and Fu, M. L. 1999a. Autoantibodies against M2 muscarinic receptors in patients with cardiomyopathy display non-desensitized agonist-like effects. Life Sciences 64, 465-469.

Wallukat, G., Muller, J., Podlowski, S., Nissen, E., Morwinski, R., and Hetzer, R. 1999b. Agonistlike beta-adrenoceptor antibodies in heart failure. The American Journal of Cardiology 83, 75-79.

Wang, M. C., Collins, R. F., Ford, R. C., Berrow, N. S., Dolphin, A. C., and Kitmitto, A. 2004a. The three-dimensional structure of the cardiac L-type voltage-gated calcium channel: comparison with the skeletal muscle form reveals a common architectural motif. J Biol Chem 279, 7159-7168.

Wang, S. Q., Song, L. S., Lakatta, E. G., and Cheng, H. 2001. Ca2+ signalling between single L-type Ca2+ channels and ryanodine receptors in heart cells. Nature 410, 592-596.

Wang, S. Q., Wei, C., Zhao, G., Brochet, D. X., Shen, J., Song, L. S., Wang, W., Yang, D., and Cheng, H. 2004b. Imaging microdomain Ca2+ in muscle cells. Circ Res 94, 1011-1022.

Wang, Y. G., Wagner, M. B., Joyner, R. W., and Kumar, R. 2000. cGMP-dependent protein kinase mediates stimulation of L-type calcium current by cGMP in rabbit atrial cells. Cardiovascular Research 48, 310-322.

Wappl, E., Mitterdorfer, J., Glossmann, H., and Striessnig, J. 2001. Mechanism of dihydropyridine interaction with critical binding residues of L-type Ca2+ channel alpha 1 subunits. J Biol Chem 276, 12730-12735.

Wei, S. K., Colecraft, H. M., DeMaria, C. D., Peterson, B. Z., Zhang, R., Kohout, T. A., Rogers, T. B., and Yue, D. T. 2000. Ca(2+) channel modulation by recombinant auxiliary beta subunits expressed in young adult heart cells. Circ Res 86, 175-184.

Weiergraber, M., Pereverzev, A., Vajna, R., Henry, M., Schramm, M., Nastainczyk, W., Grabsch, H., and Schneider, T. 2000. Immunodetection of alpha1E voltage-gated Ca(2+) channel in chromogranin-positive muscle cells of rat heart, and in distal tubules of human kidney. J Histochem Cytochem 48, 807-819.

Welling, A., Ludwig, A., Zimmer, S., Klugbauer, N., Flockerzi, V., and Hofmann, F. 1997. Alternatively Spliced IS6 Segments of the {alpha}1C Gene Determine the Tissue-Specific Dihydropyridine Sensitivity of Cardiac and Vascular Smooth Muscle L-Type Ca2+ Channels. Circ Res 81, 526-532.

Welsby, P. J., Wang, H., Wolfe, J. T., Colbran, R. J., Johnson, M. L., and Barrett, P. Q. 2003. A mechanism for the direct regulation of T-type calcium channels by Ca2+/calmodulin-dependent kinase II. J Neurosci 23, 10116-10121.

Wetzel, G. T., Chen, F., and Klitzner, T. S. 1993. Ca2+ channel kinetics in acutely isolated fetal, neonatal, and adult rabbit cardiac myocytes. Circ Res 72, 1065-1074.

White, E. and Terrar, D. A. 1992. Inactivation of Ca current during the action potential in guineapig ventricular myocytes. Exp Physiol 77, 153-164.

Wibo, M., Bravo, G., and Godfraind, T. 1991. Postnatal maturation of excitation-contraction coupling in rat ventricle in relation to the subcellular localization and surface density of 1,4-dihydropyridine and ryanodine receptors. Circ Res 68, 662-673.

Wier, W. G. and Balke, C. W. 1999. Ca2+ release mechanisms, Ca2+ sparks, and local control of excitation-contraction coupling in normal heart muscle. Circ Res 85, 770-776.

Wier, W. G., Egan, T. M., Lopez-Lopez, J. R., and Balke, C. W. 1994. Local control of excitationcontraction coupling in rat heart cells. J Physiol (Lond) 474, 463-471.

Wijffels, M. C., Kirchhof, C. J., Dorland, R., and Allessie, M. A. 1995. Atrial fibrillation begets atrial fibrillation. A study in awake chronically instrumented goats. Circulation 92, 1954-1968.

Wilkins, B. J., De Windt, L. J., Bueno, O. F., Braz, J. C., Glascock, B. J., Kimball, T. F., and Molkentin, J. D. 2002. Targeted disruption of NFATc3, but not NFATc4, reveals an intrinsic defect in calcineurin-mediated cardiac hypertrophic growth. Mol Cell Biol 22, 7603-7613.

Williams, M. E., Feldman, D. H., McCue, A. F., Brenner, R., Velicelebi, G., Ellis, S. B., and Harpold, M. M. 1992. Structure and functional expression of alpha 1, alpha 2, and beta subunits of a novel human neuronal calcium channel subtype. Neuron 8, 71-84.

Williams, M. E., Marubio, L. M., Deal, C. R., Hans, M., Brust, P. F., Philipson, L. H., Miller, R. J., Johnson, E. C., Harpold, M. M., and Ellis, S. B. 1994. Structure and functional characterization of neuronal alpha 1E calcium channel subtypes. J Biol Chem 269, 22347-22357.

Wolfe, J. T., Wang, H., Howard, J., Garrison, J. C., and Barrett, P. Q. 2003. T-type calcium channel regulation by specific G-protein betagamma subunits. Nature 424, 209-213.

Wolfe, J. T., Wang, H., Perez-Reyes, E., and Barrett, P. Q. 2002. Stimulation of recombinant Ca(v)3.2, T-type, Ca(2+) channel currents by CaMKIIgamma(C). J Physiol 538, 343-355.

Wong, C., Soeller, C., Burton, L., and Cannell, M. B. 2002. Changes in transverse-tubular system architecture in myocytes from diseased human ventricles. Biophys J 82, a588. Abstract.

Woo, S. H., Cleemann, L., and Morad, M. 2003a. Spatiotemporal characteristics of junctional and nonjunctional focal Ca2+ release in rat atrial myocytes. Circ Res 92, e1-11.

Woo, S. H., Soldatov, N. M., and Morad, M. 2003b. Modulation of Ca2+ signalling in rat atrial myocytes: possible role of the alpha1C carboxyl terminal. J Physiol 552, 437-447.

Wu, J. Y. and Lipsius, S. L. 1990. Effects of extracellular Mg2+ on T- and L-type Ca2+ currents in single atrial myocytes. Am J Physiol 259, H1842-H1850.

Wu, Y., Dzhura, I., Colbran, R. J., and Anderson, M. E. 2001. Calmodulin kinase and a calmodulinbinding 'IQ' domain facilitate L- type Ca2+ current in rabbit ventricular myocytes by a common mechanism. J Physiol 535, 679-687.

Wyatt, C. N., Campbell, V., Brodbeck, J., Brice, N. L., Page, K. M., Berrow, N. S., Brickley, K., Terracciano, C. M., Naqvi, R. U., MacLeod, K. T., and Dolphin, A. C. 1997. Voltage-dependent binding and calcium channel current inhibition by an anti-alpha 1D subunit antibody in rat dorsal root ganglion neurones and guinea-pig myocytes. J Physiol 502 (Pt 2), 307-319.

Xiao, G. Q., Hu, K., and Boutjdir, M. 2001a. Direct inhibition of expressed cardiac l- and t-type calcium channels by igg from mothers whose children have congenital heart block. Circulation 103, 1599-1604.

Xiao, G. Q., Qu, Y., Hu, K., and Boutjdir, M. 2001b. Down-regulation of L-type calcium channel in pups born to 52 kDa SSA/Ro immunized rabbits. FASEB J 15, 1539-1545.

Xiao, R. P. 2001. Beta-adrenergic signaling in the heart: dual coupling of the beta2- adrenergic receptor to G(s) and G(i) proteins. Sci STKE 2001, RE15.

Xiao, R. P., Avdonin, P., Zhou, Y. Y., Cheng, H., Akhter, S. A., Eschenhagen, T., Lefkowitz, R. J., Koch, W. J., and Lakatta, E. G. 1999. Coupling of beta2-adrenoceptor to Gi proteins and its physiological relevance in murine cardiac myocytes. Circ Res 84, 43-52.

Xiao, R. P., Cheng, H., Lederer, W. J., Suzuki, T., and Lakatta, E. G. 1994a. Dual regulation of Ca2+/calmodulin-dependent kinase II activity by membrane voltage and by calcium influx. Proc Natl Acad Sci U S A 91, 9659-9663.

Xiao, R. P., Hohl, C., Altschuld, R., Jones, L., Livingston, B., Ziman, B., Tantini, B., and Lakatta, E. G. 1994b. Beta 2-adrenergic receptor-stimulated increase in cAMP in rat heart cells is not coupled to changes in Ca2+ dynamics, contractility, or phospholamban phosphorylation. J Biol Chem 269, 19151-19156.

Xiao, R. P., Ji, X., and Lakatta, E. G. 1995. Functional coupling of the beta 2-adrenoceptor to a pertussis toxin-sensitive G protein in cardiac myocytes. Mol Pharmacol 47, 322-329.

Xiao, R. P. and Lakatta, E. G. 1993. Beta 1-adrenoceptor stimulation and beta 2-adrenoceptor stimulation differ in their effects on contraction, cytosolic Ca2+, and Ca2+ current in single rat ventricular cells. Circ Res 73, 286-300.

Xu, M., Welling, A., Paparisto, S., Hofmann, F., and Klugbauer, N. 2003. Enhanced Expression of L-type Cav1.3 Calcium Channels in Murine Embryonic Hearts from Cav1.2-deficient Mice. J Biol Chem 278, 40837-40841.

Xu, W. and Lipscombe, D. 2001. Neuronal Ca(V)1.3alpha(1) L-type channels activate at relatively hyperpolarized membrane potentials and are incompletely inhibited by dihydropyridines. J Neurosci 21, 5944-5951.

Xu, X. and Best, P. M. 1992. Postnatal changes in T-type calcium current density in rat atrial myocytes. J Physiol 454, 657-672.

Yamaguchi, H., Okuda, M., Mikala, G., Fukasawa, K., and Varadi, G. 2000a. Cloning of the beta(2a) subunit of the voltage-dependent calcium channel from human heart: cooperative effect of alpha(2)/delta and beta(2a) on the membrane expression of the alpha(1C) subunit. Biochem Biophys Res Commun 267, 156-163.

Yamaguchi, S., Okamura, Y., Nagao, T., and Adachi-Akahane, S. 2000b. Serine Residue in the IIIS5-S6 Linker of the L-type Ca2+ Channel alpha 1C Subunit Is the Critical Determinant of the Action of Dihydropyridine Ca2+ Channel Agonists. J Biol Chem 275, 41504-41511.

Yamaguchi, S., Zhorov, B. S., Yoshioka, K., Nagao, T., Ichijo, H., and Adachi-Akahane, S. 2003. Key Roles of Phe1112 and Ser1115 in the Pore-Forming IIIS5-S6 Linker of L-Type Ca2+ Channel {alpha}1C Subunit (CaV 1.2) in Binding of Dihydropyridines and Action of Ca2+ Channel Agonists. Mol Pharmacol 64, 235-248.

Yang, D., Song, L. S., Zhu, W. Z., Chakir, K., Wang, W., Wu, C., Wang, Y., Xiao, R. P., Chen, S. R., and Cheng, H. 2003. Calmodulin regulation of excitation-contraction coupling in cardiac myocytes. Circ Res 92, 659-667.

Yang, J., Drazba, J. A., Ferguson, D. G., and Bond, M. 1998. A-kinase anchoring protein 100 (AKAP100) is localized in multiple subcellular compartments in the adult rat heart. J Cell Biol 142, 511-522.

Yang, J., Ellinor, P. T., Sather, W. A., Zhang, J. F., and Tsien, R. W. 1993. Molecular determinants of Ca2+ selectivity and ion permeation in L-type Ca2+ channels. Nature 366, 158-161.

Yang, Y., Chen, X., Margulies, K., Jeevanandam, V., Pollack, P., Bailey, B. A., and Houser, S. R. 2000. L-type Ca2+ channel alpha 1c subunit isoform switching in failing human ventricular myocardium. J Mol Cell Cardiol 32, 973-984.

Yang, Z. and Steele, D. S. 2004. Properties of nuclear Ca(2+) regulation in isolated rat ventricular myocytes. J Physiol Abstract *in press*.

105

Yatani, A., Wakamori, M., Niidome, T., Yamamoto, S., Tanaka, I., Mori, Y., Katayama, K., and Green, S. 1995. Stable expression and coupling of cardiac L-type Ca2+ channels with beta 1-adrenoceptors. Circ Res 76, 335-342.

Yoshida, A., Takahashi, M., Nishimura, S., Takeshima, H., and Kokubun, S. 1992. Cyclic AMPdependent phosphorylation and regulation of the cardiac dihydropyridine-sensitive Ca channel. FEBS Lett 309, 343-349.

Yu, H., McKinnon, D., Dixon, J. E., Gao, J., Wymore, R., Cohen, I. S., Danilo, P., Jr., Shvilkin, A., Anyukhovsky, E. P., Sosunov, E. A., Hara, M., and Rosen, M. R. 1999. Transient outward current, Ito1, is altered in cardiac memory. Circulation 99, 1898-1905.

Yuan, W. and Bers, D. M. 1994. Ca-dependent facilitation of cardiac Ca current is due to Cacalmodulin- dependent protein kinase. Am J Physiol 267, H982-H993.

Yuan, W., Ginsburg, K. S., and Bers, D. M. 1996. Comparison of sarcolemmal calcium channel current in rabbit and rat ventricular myocytes. J Physiol 493 (Pt 3), 733-746.

Yue, D. T., Herzig, S., and Marban, E. 1990. Beta-adrenergic stimulation of calcium channels occurs by potentiation of high-activity gating modes. Proc Natl Acad Sci U S A 87, 753-757.

Yue, D. T. and Marban, E. 1990. Permeation in the dihydropyridine-sensitive calcium channel. Multi-ion occupancy but no anomalous mole-fraction effect between Ba2+ and Ca2+. J Gen Physiol 95, 911-939.

Yue, L., Feng, J., Gaspo, R., Li, G. R., Wang, Z., and Nattel, S. 1997. Ionic remodeling underlying action potential changes in a canine model of atrial fibrillation. Circ Res 81, 512-525.

Yue, L., Melnyk, P., Gaspo, R., Wang, Z., and Nattel, S. 1999. Molecular mechanisms underlying ionic remodeling in a dog model of atrial fibrillation. Circ Res 84, 776-784.

Yunker, A. M. 2003. Modulation and pharmacology of low voltage-activated ("T-Type") calcium channels. J Bioenerg Biomembr 35, 577-598.

Zaccolo, M. and Pozzan, T. 2002. Discrete microdomains with high concentration of cAMP in stimulated rat neonatal cardiac myocytes. Science 295, 1711-1715.

Zhang, T., Johnson, E. N., Gu, Y., Morissette, M. R., Sah, V. P., Gigena, M. S., Belke, D. D., Dillmann, W. H., Rogers, T. B., Schulman, H., Ross, J., Jr., and Brown, J. H. 2002a. The cardiac-specific nuclear delta(B) isoform of Ca2+/calmodulin-dependent protein kinase II induces hypertrophy and dilated cardiomyopathy associated with increased protein phosphatase 2A activity. J Biol Chem 277, 1261-1267.

Zhang, Z., Xu, Y., Song, H., Rodriguez, J., Tuteja, D., Namkung, Y., Shin, H. S., and Chiamvimonvat, N. 2002b. Functional Roles of Cav1.3 ({alpha}1D) Calcium Channel in Sinoatrial Nodes: Insight Gained Using Gene-Targeted Null Mutant Mice. Circ Res 90, 981-987.

Zhao, R., Wang, W., Wu, B., Hoebeke, J., Hjalmarson, A., and Fu, M. L. 1996. Effects of antipeptide antibodies against the second extracellular loop of human M2 muscarinic acetylcholine receptors on transmembrane potentials and currents in guinea pig ventricular myocytes. Mol Cell Biochem 163-164, 185-193.

Zhou, J., Olcese, R., Qin, N., Noceti, F., Birnbaumer, L., and Stefani, E. 1997. Feedback inhibition of Ca2+ channels by Ca2+ depends on a short sequence of the C terminus that does not include the

Ca2+ -binding function of a motif with similarity to Ca2+ -binding domains. Proc Natl Acad Sci U S A 94, 2301-2305.

Zhou, Z. and January, C. T. 1998. Both T- and L-type Ca2+ channels can contribute to excitationcontraction coupling in cardiac Purkinje cells. Biophys J 74, 1830-1839.

Zhou, Z. and Lipsius, S. L. 1994. T-type calcium current in latent pacemaker cells isolated from cat right atrium. J Mol Cell Cardiol 26, 1211-1219.

Zong, X., Schreieck, J., Mehrke, G., Welling, A., Schuster, A., Bosse, E., Flockerzi, V., and Hofmann, F. 1995. On the regulation of the expressed L-type calcium channel by cAMP-dependent phosphorylation. Pflugers Arch 430, 340-347.

Zuhlke, R. D., Pitt, G. S., Deisseroth, K., Tsien, R. W., and Reuter, H. 1999. Calmodulin supports both inactivation and facilitation of L-type calcium channels. Nature 399, 159-162.

Zuhlke, R. D., Pitt, G. S., Tsien, R. W., and Reuter, H. 2000. Ca2+-sensitive inactivation and facilitation of L-type Ca2+ channels both depend on specific amino acid residues in a consensus calmodulin-binding motif in the(alpha)1C subunit. J Biol Chem 275, 21121-21129.

Zuhlke, R. D. and Reuter, H. 1998. Ca2+-sensitive inactivation of L-type Ca2+ channels depends on multiple cytoplasmic amino acid sequences of the alpha1C subunit. Proc Natl Acad Sci U S A 95, 3287-3294.

Zygmunt, A. C. and Maylie, J. 1990. Stimulation-dependent facilitation of the high threshold calcium current in guinea-pig ventricular myocytes. J Physiol 428, 653-671.
<b>Table 1:</b> Nomenclature of $Ca^{2+}$ channels. This table shows all $Ca^{2+}$ channels described to date, for
further information about non-cardiac Ca <sup>2+</sup> channels see Catterall, 2000.
L: long lasting: T: tiny and Transient: N: Neither T nor L and Neuron: P: Purkinie: O: After P: R:

L: long lasting; T: tiny and Transient; N: Neither T nor L and Neuron; P: Purkinje; Q: After P; R: Resistant. HVA: High Voltage Activated; LVA: Low Voltage Activated. DHP: dihydropyridine.

Ca <sup>2+</sup> channel	Ca <sup>2+</sup> current	Previous nomenclature	Localization	Electrophysiological and pharmacological phenotype
Ca <sub>v</sub> 1.1	L	$\alpha_{1S}$	Skeletal muscle	
Ca <sub>v</sub> 1.2	L	$\alpha_{1C}$	<b>Heart</b> , Endocrine cells, Neurons, smooth muscle	HVA DHP sensitive
Ca <sub>v</sub> 1.3	L	$\alpha_{1D}$	Endocrine cells, Neurons, Heart	
Ca <sub>V</sub> 1.4	L	$\alpha_{1F}$	Retina	
Ca <sub>v</sub> 2.1	P/Q	$\alpha_{1A}$	Neurons	
Ca <sub>v</sub> 2.2	Ν	$\alpha_{1B}$	Neurons	HVA DHP insensitive
Ca <sub>v</sub> 2.3	R	$\alpha_{1E}$	Neurons, Heart	
Ca <sub>v</sub> 3.1	Т	$\alpha_{1G}$	Heart, Neurons	
Ca <sub>v</sub> 3.2	Т	$lpha_{1\mathrm{H}}$	Heart, Neurons	LVA DHP insensitive
Ca <sub>v</sub> 3.3	Т	$\alpha_{11}$	Neurons	

**Table 2:** Summary of the effects of NO/cGMP/PKG cascade modulation on basal  $I_{Ca}$  and prestimulated  $I_{Ca}$ . All these studies were performed with the whole cell configuration of the patch-clamp technique (otherwise stated). (WT: wild-type; TG: transgenic).

Modultion of cGMP dpt- pathway	Preparations	Tissu	Basal $\Delta I_{Ca}$	$\Delta I_{Ca}$ prestimulated	cAMP-elevating agents	Refs
NOS inhibitors/NO scavengers						
L-NMMA 1 mM L-NNA 1 mM	Guinea pig	ventricle	>			1
PTIO 0.5 mM	Guinea pig	ventricle	1			2
L-NMMA 1mM	Human	atrium	$\leftrightarrow$			3
NO donors						
SIN-1 0.1-10 nM SIN-1 0.1-100 μM	Frog	ventricle	$    \leftrightarrow \\ \leftrightarrow $	≯ `\ ↔	Fsk 0.3 μM / ISO 0.1-1 μM / cAMP 10 μM Fsk 0.3 μM / ISO 0.1-1 μM / cAMP 10 μM 8-Br-cAMP 10 μM	4
nitroprusside 10 µM	Guinea pig	ventricle	$\leftrightarrow$	5	IBMX 0.1 mM	5
SIN-1 10 μM	Guinea pig	ventricle	$\leftrightarrow$	У	IBMX 0.1 mM	6
SIN-1 10-100μM SIN-1 10μM SIN-1 100μM SIN-1 1-100 μM	Guinea pig	ventricle	±↔	ノor S S S	ISO 10 nM ISO 10 nM ISO 1μΜ / IBMX 0.1 mM / cAMP 10 μM or 8-Br-cAMP 10 μM	7
DEANO 100 μM GSNO 1 mM SNAP 1μM-1 mM SNP 5-500 μM SIN-1 0.1-1 mM SPNO 0.1 μM DEANO 10-100 μM DEANO 100 μM	Rat	ventricle	$\begin{array}{c} \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \end{array}$	$\begin{array}{c} \searrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \leftrightarrow \\ \searrow \end{array}$	ISO 1-10 nM ISO 1-10 nM / cAMP 10-100 μM cAMP 10-100-μM ISO 1-10 nM ISO 1-10 nM cAMP 10-100 μM IBMX 10 80 μM	8
SNAP 1 pM-1µM DEANO 1 pM-1 nM SNAP 1 pM-1µM	Rat	ventricle	$\leftrightarrow$	$\begin{array}{c} \searrow \\ \searrow \\ \leftrightarrow \end{array}$	ISO 0.1-0.5 nM ISO 0.01pM-0.5 nM L858051 0.1-0.3 μM	9
DEANO 0.1 µM overexpressed PKG 1	TG vs WT mice	ventricle (cell- attached)	\$	Y	ISO 1µM	10
GSNO 10 µM	Newborn Rabbit	ventricle	/	1	IBMX 50µM	11
GSNO 10 µM	Adult Rabbit	ventricle	$\leftrightarrow$			11
SIN-1 1 pM-10 nM SIN-1-100 μM SIN-1 0.1-1 nM	Human	atrium	✓ moderate ✓	$\leftrightarrow$	milrinone 10 μM	12
SNAP 100 mM	Human	atrium	7			3
GSNO 10-100 μM	Rabbit	atrium	7	$\stackrel{\nearrow}{\leftrightarrow}$	ISO 1 nM / IBMX 10 μM ISO 100 nM	13
cGMP intrapipette						
0.3-20 μM	Frog	ventricle	$\leftrightarrow$	$\stackrel{\checkmark}{\underset{\leftrightarrow}{}}$ moderate $\stackrel{\checkmark}{}{}$	ISO 0.3 μM cAMP intra 5-20 μM IBMX 0.1 mM 8-Br-cAMP 1μM	14
0.1-20μΜ	Frog	ventricle	$\leftrightarrow$	$\begin{array}{c} \searrow \\ \leftrightarrow \\ \pm \leftrightarrow \end{array}$	ISO 0.3-2 μM/ cAMP intra 5-30 μM 8-Br-cAMP 5 μM cAMP 5 μM + IBMX 100 μM	15
5 μΜ	Guinea-pig	ventricle	$\leftrightarrow$	5	cAMP intra 50 μM / 8-Br-cAMP 50 μM IBMX 0.1 mM	16
1-10 μM 1 mM	Guinea pig	ventricle	$\leftrightarrow$	$\rightarrow$ $\rightarrow$ $\rightarrow$ $\rightarrow$ $\rightarrow$	ISO 1-100nM / Fsk 0.5-1μM / cAMP 50-100 μM 8-CPT-cAMP / 8-Br-cAMP (both 50μM) IBMX 40 μM ISO 30 nM	17
30 μM 30-300 μM 1 mM 1 mM 100 μM	Guinea pig	ventricle	2	Z \ \ \ \	ISO 3-nM ISO 3-nM ISO 30 nM IBMX 0.1 mM	18
10 μM	Guinea pig	ventricle	$\leftrightarrow$	7	IBMX 0.1 mM	5
0.1-100μΜ	Rat	ventricle	$\leftrightarrow$	2	(all at 0.1mM)	19
100 μΜ	Young rabbit	ventricle	7	1	cAMP 100 μM	20

109

Modultion of cGMP dpt- pathway	Preparations	Tissu	Basal $\Delta I_{Ca}$	ΔI <sub>Ca</sub> prestimulated	cAMP-elevating agents	Refs
0.5 μM 50 μM	Human	atrium	≯ moderate≯			22
0.5 uM	Human	atrium	7			
PKG activators						
8-Br-cGMP 20 uM	Frog	ventricle		$\leftrightarrow$	cAMP intra 5 uM	14/15
8-Br-cGMP 100 µM	Guinea-pig	ventricle		<b>N</b>	IBMX 0.1 mM	16
8-Br-cGMP 10 µM	Guinea pig	ventricle	$\leftrightarrow$	±γ	cAMP intra 50uM	17
8-Br-cGMP 0 1mM	Guinea pig	ventricle	$\leftrightarrow$	<b>N</b>	ISO 10 nM / IBMX 0.1 mM	6
8-Br-cGMP 1mM	Guinea pig	ventricle		↔ (ATPySpip)	IBMX 0.1 mM	23
8-Br-cGMP 10-30 µM	Guinea pig	ventricle		± >	ISO 30 nM	18
8-Br-cGMP 100 μM	Guinea pig	ventricle		$\leftrightarrow$	сАМР 100µМ IBMX 100 µM	24
8-Br-cGMP 0.1-10 μM	Young Rat	ventricle	<b>`</b>			25
8-Br-cGMP 0.1-10μM	Rat	ventricle		5	cAMP intra / 8-Br-cAMP intra (all at 0.1mM)	19
8-p-CPT-cGMP 100 μM	Rat	ventricle		<b>`</b>	ISO 1 nM	8
8-Br-cGMP 30-100 μM 8-p-CPT-cGMP 30 μM	Newborn Rabbit	ventricle	7 7			11
8-Br-cGMP 100 μM 8-p-CPT-cGMP 400 μM 8-p-CPT-cGMP 400 μM	Young rabbit	ventricle	7	7	IBMX 0.1 mM	20
8-Br-cGMP 30-100 μM	Adult Rabbit	ventricle	$\leftrightarrow$			11
8-Br-cGMP 30 μM	Rabbit	atrium	7			13
8-Br-cGMP 300 μM	Rabbit	ventricle	\$	5	IBMX 0.1 mM	26
8-bromo-cGMP 0.5-5 μM 8-bromo-cGMP 100 μM	Human	atrium	$ \stackrel{\leftrightarrow}{\scriptstyle \nearrow} $			22
8-Br-cGMP 0.1-1 mM	Cultured chick cells (cell-attac	embryonic hed)	У			27
8-Br-cGMP 1 mM	Cultured chick embryonic cells		7	\$	ΡΚΑ 1.5 μΜ	28
8-Br-cGMP 1 mM	Xenopus oocyte	$\begin{array}{l} \alpha_{1c}(\text{double} \\ \text{microelectrode}) \end{array}$	2			29
8-Br-cGMP 1 mM overexpressed PKG 1	TG vs WT mice	ventricle (cell- attached)	>	У	ISO 1µM	10
8-Br-cGMP 1 mM	Mice	Ventricle (cell- attached)	$\leftrightarrow$	У	ISO 1µM / 8-Br-cAMP 1 mM	30
PKG						
Active fragment 0.8 µM	Guinea pig	ventricle		$\pm \leftrightarrow$	ISO 0.1 μM	17
Pre-activated PKG 50 nM	Young Rat	ventricle	± <b>`</b>	N N N	ISO 2 μM Bay K 8644 1 μM IBMX 0.1-0.3 mM subsequent ISO 2 μM	25
Active fragment 10 nM-1µM	Rat	ventricle	$\leftrightarrow$	7	cAMP intra / 8-Br-cAMP intra (all at 0.1mM)	19
Pre-activated PKG 25 nM	Cultured chick embryonic cells		7	$\leftrightarrow$	Subsequent ISO 1-10 µM	28
Overexpressed PKG 1	TG vs WT mice	ventricle (cell- attached)	$\leftrightarrow$			10

**References:** (1) Gallo et al., 1998; (2) Gallo et al., 2001; (3) Vandecasteele et al., 1998; (4) Méry et al., 1993; (5) Levi et al., 1994; (6) Imai et al., 2001; (7) Wahler and Dollinger 1995; (8) Abi-Gerges et al., 2001; (9) Abi-Gerges et al., 2002; (10) Schröder et al., 2003; (11) Kumar et al., 1997; (12) Kirstein et al., 1995; (13) Wang et al., 2000; (14) Hartzell and Fischmeister, 1986; (15) Fischmeister and Hartzell, 1987; (16) Levi et al., 1989; (17) Ono and Trauwein 1991; (18) Shirayama and Pappano 1996; (19) Méry et al., 1991; (20) Han et al., 1998; (21) Rivet-Bastide et al., 1997; (22) Vandecasteele et al., 2001; (23) Shen and Pappano, 2002; (24) Sakai et al., 1999; (25) Sumii and Sperelakis, 1995; (26) Tohse et al., 1995; (27) Tohse and Sperelakis, 1991; (28) Haddad et al., 1995; (29) Jiang et al., 2000; (30) Klein et al., 2000.

# **Figures legends:**

#### Figure 1:

Comparison of cardiac Ca<sup>2+</sup> channels electrophysiological properties. Peak current-voltage relationships for L-type Ca<sub>V</sub>1.2 ( $\alpha_{1C}$ ) and Ca<sub>V</sub>1.3 ( $\alpha_{1D}$ ), and T-type Ca<sub>V</sub>3.1 ( $\alpha_{1G}$ ) channels are normalized. L-type Ca<sub>V</sub>1.3 channels activate at negative membrane potentials similar to T-type Ca<sub>V</sub>3.1 channels. Activation midpoints (V<sub>1/2</sub>) are approximately -5 mV for L-type Ca<sub>V</sub>1.2, -30 mV for L-type Ca<sub>V</sub>1.3 and T-type Ca<sub>V</sub>3.1. Curves were generated by a Boltzmann-GHK function using parameters obtained from recombinant channels expressed in *Xenopus* oocytes recorded under similar conditions (10 mmol/L extracellular Ba<sup>2+</sup>). Reproduced with permission from "L-Type Calcium Channels Highs and New Lows, 2002, Lipscombe D, Circ Res, 90, 933-2002".

#### Figure 2:

Time course of  $I_{CaL}$  in guinea-pig, rabbit, and rat myocytes voltage-clamped with a guineapig action potential waveform. Top panel: command voltage. Bottom panel: recordings of  $I_{CaL}$ normalised to cell capacitance averaged from 12 (guinea-pig), 10 (rabbit) and 10 (rat) cells. Reproduced with permission from "Profile and kinetics of L-type calcium current during the cardiac ventricular action potential compared in guinea-pigs, rats and rabbits, 2000, Linz K.W., Meyer R., Pflügers Archiv, 439, 588-599".

#### Figure 3:

Sparklet-spark coupling in heart cells. Data were obtained in the loose-seal patch-clamp conditions. A. Triggering  $Ca^{2+}$  spark by single L-type  $Ca^{2+}$  channel opening. B. Representative example showing that a  $Ca^{2+}$  sparklet directly triggers a  $Ca^{2+}$  spark. Reproduced with permission from "Imaging Microdomain  $Ca^{2+}$  in Muscle, 2004, Cells Wang, SQ, Wei, C, Zhao, G, Brochet, DXP, Shen, J, Song, LS, Wang, W, Yang, D, Cheng, H. Circ Res, 2004, 94, 1011-1022".

#### Figure 4:

Possible excitation-transcription coupling in cardiac myocytes. CaM, calmodulin; CamKII, Ca<sup>2+</sup>-calmodulin-dependent protein kinase II; NFAT, nuclear factor of activated T cells; DREAM, downstream regulatory element antagonistic modulator; CREB, cAMP-responsive element binding protein. Sources for the interaction between the molecules listed above are cited in the text.

#### Figure 5:

Schematic diagram of L type  $Ca^{2+}$  channel modulation by cGMP pathway in cardiac myocytes. NOS, nitric oxide synthase; NO, nitric oxide; GC, guanylyl cyclase; AC, Adenylyl Cyclase; cAMP, cyclic adenosine monophosphate; cGMP, cyclic guanosine monophosphate; PDE, phosphodiesterase; PKG, cGMP-dependent protein kinase; PKA, cAMP-dependent protein kinase;  $\beta_1R$ , type 1  $\beta$ -adrenoceptor; AKAP, A-kinase anchoring protein; PPases, Protein phosphatases;  $Ca_V 1.2$ , L type  $Ca^{2+}$  channel. Sources for the interaction between the proteins listed above are cited in the text.

### Figure 6:

Schematic representation of cardiac L-type  $Ca^{2+}$  channel  $Ca_V 1.2$  ( $\alpha_{1C}$ ) subunit with accessory subunits ( $\beta$  and  $\alpha_2\delta$ ). AID, Alpha Interacting Domain; PKA, cAMP-dependent protein kinase; PKC, protein kinase C, PKG, cGMP-dependent protein kinase; CaM, calmodulin, AKAP, A-kinase anchoring protein; see text for details.



113













(Isenberg and Klockner, 1982) (Rougier et al., 1969) (Lederer et al., 1990) (Fabiato, 1983) (Xiao, 2001) (Razani et al., 2002) (Harvey and Belevych, 2003) (Abi-Gerges et al., 2002) (Maurice et al., 2003) (Vandecasteele et al., 1998) (Razani et al., 2002) (Mery et al., 1993) (Imai et al., 2001) (Abi-Gerges et al., 2002) (Abi-Gerges et al., 2002) (Schroder et al., 2003)

(Kumar et al., 1997) (Wang et al., 2000) (Sumii and Sperelakis, 1995; Rivet-Bastide et al., 1997; Jiang et al., 2000; Klein et al., 2000; Shen and Pappano, 2002) (Malkiel et al., 1996; Brodde and Michel, 1999; Caforio et al., 2002; Kierszenbaum, 2003)

(Mijares et al., 1996b; Jahns et al., 2004)

(Hirano et al., 1999) (Brette and Orchard, 2003) (Lipsius et al., 2001) (Klockner et al., 1999) (Fabiato, 1983) (Trafford et al., 2002)

(Carl et al., 1995; Franzini-Armstrong et al., 1999; Scriven et al., 2000; Gathercole et al., 2000) (Ferrier and Howlett, 2001) (Dolmetsch, 2003) (Dolphin, 2003b) (Puglisi et al., 1999)) and rat (~ 40%, Takamatsu et al., 2003 (Takamatsu et al., 2003) (Hare and Stamler, 1999) (Harvey and Belevych, 2003) (Hare, 2003) (Eisner et al., 2000) (Levi et al., 1994; Gallo et al., 1998; Gallo et al., 2001) (Hartzell and Fischmeister, 1986; Fischmeister and Hartzell, 1987; Levi et al., 1989; Tohse and Sperelakis, 1991; Wahler and Dollinger, 1995; Tohse et al., 1995) (Hoebeke, 2001) (Magnusson et al., 1994; Catterall, 2000) (Perez-Reyes et al., 1998) (Woo et al., 2003b) (Romanin et al., 2000) (Zhou et al., 1997) (Gao et al., 1999) (Takahashi et al., 2004) (McGee et al., 2004) (Boutjdir, 2000) (Neher, 1998) (Lipscombe, 2002) (Linz and Meyer, 2000) (Balijepalli et al., 2003) (Bers, 2001).

(Tsien, 1973; Hess et al., 1984; Hess and Tsien, 1984; Almers and McCleskey, 1984; Tsien et al., 1986; Lansman et al., 1986; Hess et al., 1986; Tsien et al., 1987; Yue and Marban, 1990)